UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, DC 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE **SECURITIES EXCHANGE ACT OF 1934**

Date of report (Date of earliest event reported): December 21, 2012

CorEnergy Infrastructure Trust, Inc. (Exact Name of Registrant as Specified in Its Charter)

Maryland (State or Other Jurisdiction of Incorporation)

1-33292 (Commission File Number)

20-3431375 (IRS Employer Identification No.)

4200 W. 115th Street, Suite 210, Leawood, KS (Address of Principal Executive Offices)

(913) 981-1020

(Registrant's Telephone Number, Including Area Code)

(Former Name or Former Address, if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

66211

(Zip Code)

Introductory Note

CorEnergy Infrastructure Trust, Inc. (the "Company") previously announced that on December 7, 2012, Pinedale Corridor, LP ("Pinedale LP"), a newly formed subsidiary of the Company, entered into a Purchase and Sale Agreement with Ultra Wyoming, Inc., an indirect wholly-owned subsidiary of Ultra Petroleum Corp. ("Ultra Petroleum"). The Company also previously announced that on December 13, 2012 the parties to the Purchase and Sale Agreement entered into a First Amendment to Purchase and Sale Agreement (the "Purchase and Sale Agreement"). The Purchase and Sale Agreement provides for Pinedale LP's acquisition of a system of pipelines and central gathering facilities (the "LGS") in the Pinedale Anticline in Wyoming (the "Acquisition"). The Purchase and Sale Agreement provides that at the closing of the Acquisition, Pinedale LP will enter into a 15-year triple net lease with Ultra Wyoming LGS, LLC, an indirect wholly-owned subsidiary of Ultra Petroleum ("Ultra Wyoming"), relating to the use of the LGS (the "Lease").

The Company also previously announced that on December 7, 2012, Pinedale LP and Pinedale GP, Inc., a newly formed subsidiary of the Company and the general partner of Pinedale LP ("Pinedale GP"), entered into a Subscription Agreement with Ross Avenue Investments, LLC, an indirect wholly-owned subsidiary of Prudential Financial, Inc. ("Prudential"), pursuant to which Prudential has agreed to fund (the "Co-Investment") a portion of the Acquisition by investing \$30 million in cash in Pinedale LP.

The Company also previously announced that on December 7, 2012, Pinedale LP entered into a \$65 million secured Term Credit Agreement (the "Credit Facility") with KeyBank National Association serving as a lender and the administrative agent on behalf of other participating lenders and that on December 14, 2012, the parties entered into an Amended and Restated Term Credit Agreement to increase the amount available to borrow thereunder to \$70 million.

The Company also previously announced that on December 18, 2012 it completed a public offering of 13,000,000 shares of its common stock resulting in gross proceeds to the Company of \$78 million (the "Offering").

Item 1.01 Entry into a Material Definitive Agreement

Lease

On December 20, 2012, Pinedale LP entered into the Lease with Ultra Wyoming and all ancillary agreements thereto. The Lease provides for Pinedale to receive fixed monthly rental payments of \$1,666,667 (as adjusted annually for changes based on the consumer price index) and variable rent based on the volume of liquid hydrocarbons and water that flowed through the LGS in a prior month. The minimum and maximum annual rental payments under the Lease during the initial fifteen year term are \$20 million (as adjusted annually for changes based on the consumer price index) and \$27.5 million, respectively, subject to Pinedale LP not being in default under the Lease. The rent will be renegotiated for any extended term. The Lease provides that Ultra Wyoming will be responsible for, among other matters, maintaining the LGS in good operating condition, repairing the LGS in the event of any casualty loss (except upon the occurrence of an event damaging more than 50% of the LGS, in which case Ultra Wyoming may propose to repurchase the LGS at a mutually agreeable price rather than repair the LGS), paying property and similar taxes resulting from Pinedale LP's ownership of the LGS, and causing the LGS to comply with all environmental and other regulatory laws, rules and regulations. The Lease grants Ultra Newco substantially all authority to operate, and imposes on them

the responsibility for the operation of, the LGS. The Lease provides Pinedale LP no control over the operation, maintenance, management or legal compliance of the LGS.

The Lease imposes numerous obligations on Pinedale LP, including maintaining its status as a special purpose entity that only engages in the business of owning, financing and leasing the LGS, keeping confidential certain information provided to it by Ultra Wyoming, keeping the LGS free of certain liens, observing certain limitations on the transfer and ownership of beneficial interests in, and control of, Pinedale LP and maintaining the relationship between Pinedale LP and its lender.

The foregoing summary of the terms of the Lease are only a brief description of certain terms therein, do not purport to be a complete description of the rights and obligations of the parties thereunder, and are qualified in their entirety by the full Lease, a copy of which is incorporated by reference as Exhibit 10.1 to this Current Report on Form 8-K and incorporated herein by reference.

Partnership Agreement

On December 20, 2012, Pinedale GP and Prudential entered into the First Amended and Restated Limited Partnership Agreement of Pinedale LP (the "Partnership Agreement"). The Partnership Agreement provides Pinedale GP broad discretion to manage and make decisions relating to Pinedale LP, but also provides Prudential certain approval rights concerning the Lease and any financings undertaken by Pinedale LP.

The foregoing summary of the terms of the Partnership Agreement are only a brief description of certain terms therein, do not purport to be a complete description of the rights and obligations of the parties thereunder, and are qualified in their entirety by the full Partnership Agreement, a copy of which is attached as Exhibit 10.2 to this Current Report on Form 8-K and is incorporated herein by reference.

Item 2.01 Completion of Acquisition or Disposition of Assets

On December 20, 2012, Pinedale LP completed the Acquisition. The information regarding the Acquisition set forth in the first paragraph of the Introductory Note of this Current Report on Form 8-K is incorporated into this Item 2.01 by reference.

Item 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant

The information regarding the Amended Credit Facility set forth in Item 1.01 of this Current Report on Form 8-K is incorporated into this Item 2.03 by reference.

Item 8.01 Other Events

On December 20, 2012, the Company issued a press release announcing the closing of the Acquisition. The press release is attached hereto as Exhibit 99.1 and is incorporated herein by reference.

Item 9.01 Financial Statements and Exhibits.

- (b) The required pro forma financial information is incorporated by reference herein from Exhibit 99.2 to the Company's Current Report on Form 8-K filed on December 10, 2012.
- (d) Exhibits
- 10.1 Lease Agreement dated December 20, 2012 by and between Pinedale Corridor, LP and Ultra Wyoming LGS, LLC
- 10.2 First Amended and Restated Limited Partnership Agreement of Pinedale Corridor, LP by and between Pinedale GP, Inc. and Ross Avenue Investments, LLC
- 99.1 Press Release dated December 20, 2012

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

CORENERGY INFRASTRUCTURE TRUST, INC.

Dated: December 21, 2012

/s/ David J. Schulte David J. Schulte Chief Executive Officer and President

Exhibit No. Description

- 10.1 Lease Agreement dated December 20, 2012 by and between Pinedale Corridor, LP and Ultra Wyoming LGS, LLC (incorporated by reference from Exhibit L to the Purchase and Sale Agreement filed as Exhibit 2.1 to the Company's Current Report on Form 8-K on December 10, 2012)
- 10.2 First Amended and Restated Limited Partnership Agreement of Pinedale Corridor, LP by and between Pinedale GP, Inc. and Ross Avenue Investments, LLC
- 99.1 Press Release dated December 20, 2012

Confidential Treatment Requested. Confidential treatment has been requested for portions of this exhibit. The copy filed herewith omits the information subject to the confidentiality request. Omissions are designated as "[Redacted]." A complete version of this exhibit has been filed separately with the Securities and Exchange Commission.

LEASE

between

ULTRA WYOMING LGS, LLC,

as Lessee

and

PINEDALE CORRIDOR, LP,

as Lessor

TABLE OF CONTENTS

		Page
ARTICLE 1	FUNDAMENTAL LEASE PROVISIONS	1
1.1	Definitions	1
1.2	Certain Interpretive Provisions	33
ARTICLE II	LEASE CHARACTERIZATION	33
AKTICLE II	LEASE CHARACTERIZATION	35
2.1	Lease Characterization	33
		24
ARTICLE III	LEASED PROPERTY	34
3.1	Lease of Leased Property	34
3.2	Additional Encumbrances	35
3.3	Lessee Other Activities	36
ARTICLE IV	REPRESENTATIONS AND WARRANTIES	36
4.1	Representations and Warranties of Lessor	36
4.2	Representations and Warranties of Lessee	37
ARTICLE V	RENT	38
5.1	Base Rent	38
5.2	Account for Payment of Base Rent	38
5.3	Payment to Party Claiming Rent	38
5.4	Past Due Rent	38
ARTICLE VI	TAXES AND IMPOSITIONS	38
6.1	Taxes and Impositions	39
6.2	Payment	39
6.3	Exclusions from Taxes and Impositions	39
6.4	Payment of Taxes and Impositions	40
6.5	Abatements	40
6.6	Right to Contest Taxes	40
6.7	Right to Contest Impositions	41
ARTICLE VII	CONDITION OF LEASED PROPERTY; MAINTENANCE	41
7.1	Lessee Maintenance of Leased Property	41
7.2	No Trespass	42
ARTICLE VIII	UTILITIES	42
AKTICLE VIII	UTILITIES	42
8.1	Payment of Utility Charges	42
0.1	r ayment of Offiny Charges	42
ARTICLE IX	USE	42
		42
9.1	Lessee Use	42
		12

		Page
9.2	Operating Requirement	42
ARTICLE X	PERMITTED CAPITAL IMPROVEMENT AND ADDITIONAL LINES, PARTS AND LESSEE OTHER ACTIVITIES	42
10.1	Permitted Capital Improvements and Additional Lines	44
10.2	Replacement of Parts	45
10.3	Lessee Other Activities	45
ARTICLE XI	REGULATORY ISSUES	45
		16
11.1	Lessee's Rights	46
11.2	Lessor's Rights	47
11.3	No Application to Condemnation	47
ARTICLE XII	MECHANIC'S LIENS	47
12.1	Lessee's Obligations	47
12.2	Lessor's Obligations	47
1010		• •
ARTICLE XIII	LESSEE PROPERTY	47
13.1	No Lien	47
13.2	Sole Risk of Lessee	48
13.3	Lessee's Ability to Grant Liens on Lessee Property	48
ARTICLE XIV	INSURANCE	48
14.1	Insurance Coverage	48
14.2	Insurance Requirements Generally	49
14.3	Self-Insurance	49
14.4	Waiver of Subrogation	50
14.5	No Release from Liability	50
ARTICLE XV	FIRE AND OTHER CASUALTY	50
15 1	Fire and Other Copulate	50
15.1 15.2	Fire and Other Casualty Restoration Conditions	50 52
13.2	Resolution Columnis	52
ARTICLE XVI	CONDEMNATION	52
16.1	Condemnation Damages and Awards	52
16.2	Apportionment of Award, Termination and Purchase Offer	53
16.3	Restoration Conditions	54
ARTICLE XVII	LIMITS ON TRANSFERS, ASSIGNMENTS, LEASES AND LIENS	55
17.1	Assistant and California has Lances	
17.1	Assignment and Subletting by Lessee	55
17.2	No Restrictions on Indebtedness or Liens of Lessee or Lessee Guarantors	57
17.3	Cure Rights Upon Assignee Default	58

ii

	Page
Release of Lessee and Lessee Guarantor	58
	58
LESSEE'S SURRENDER OF LEASED PROPERTY	62
	()
Surrender	62
ASSIGNMENT OF LEASE	62
Assignment of Lease	62
MORTGAGE SUBORDINATION AND NON-DISTURBANCE AND NOTICE TO MORTGAGEE	62
Mortgage Subordination	62
INDEMNIEICATION	63
	05
Indemnification by Lessee	63
	64
	65
Survival	65
Claims Procedure	65
ENVIRONMENTAL LAWS	65
	65
	65
	66
	66
	67
Survival	67
DEFAULTS AND REMEDIES	67
Lessee Events of Default	67
	68
	71
	72
Mitigation of Damages	74
NOTICE	74
Notices	74
	75
	75
RENEWAL AND END OF TERM OPTIONS, RIGHT OF FIRST REFUSAL, AND OTHER PREFERENTIAL PURCHASE RIGHTS	76
	Surrender ASSIGNMENT OF LEASE Assignment of Lease MORTGAGE SUBORDINATION AND NON-DISTURBANCE AND NOTICE TO MORTGAGEE Mortgage Subordination NDEMNIFICATION Indemnification by Lessee Release and Indemnification by Lessor Concurrent Negligence Survival Claims Procedure ENVIRONMENTAL LAWS Environmental Undertakings Environmental Undertakings Environmental Undertakings Claims Procedure Survival DEFAULTS AND REMEDIES Lessee Events of Default Lessor Events of Default Lessor Semeticies for a Lessee Event of Default Lessor Events of Default Lessor E

iii

		Page
25.1	Demoused Option	76
25.1 25.2	Renewal Option Lessee as Operator After Term	76 78
	Right of First Refusal With Respect to the Transfer of All of the Leased Property	78
25.3 25.4	Right of First Refusal With Respect to the Transfer of All of the Leased Property	81
		81
25.5 25.6	Burdensome Buyout	82
25.0	Application of Certain Right of First Refusal Rights to a Foreclosure Sale	84
ARTICLE XXVI	MISCELLANEOUS PROVISIONS	84
26.1	Memorandum/Notice of Lease	84
26.2	Force Majeure	84
26.3	Consequential Damages	84
26.4	Holding Over	84
26.5	Quiet Enjoyment	84
26.6	Cost and Expense	85
26.7	Access; Reporting	85
26.8	Accord and Satisfaction	86
26.9	Prevailing Party	87
26.10	Confidentiality	87
26.11	Consent of Lessor and Lessee	88
26.12	Permitted Lessee Contests	88
26.13	Waiver	88
26.14	Interpretation	89
26.15	No Derivative Liability	89
26.16	Successors and Assigns	89
26.17	No Offer; Entire Agreement	89
26.18	Headings	89
26.19	Counterparts	89
26.20	Governing Law; Venue; Service of Process; Waiver of Jury Trial	90
26.21	Time of the Essence	90
26.22	Estoppel Certificates	90
ARTICLE XXVII	OTHER AGREEMENTS OF LESSOR	91
27.1	Special Purpose Entity; Lessor's Governing Documents	91
27.2	Additional Lessor Covenants	91

iv

EXHIBITS

EXHIBIT A	BLM Easements
EXHIBIT B	Central Gathering Facility Exhibit
EXHIBIT C	Form of Equity Investor Agreement
EXHIBIT D	Estoppel Certificate
EXHIBIT E	Form of SNDA
EXHIBIT F-1	Form of Services Agreement
EXHIBIT F-2	Form of Gathering Agreement
EXHIBIT G	[Intentionally Deleted]
EXHIBIT H	List of Improvements
EXHIBIT I	List of Initial Lessor Loan Document
EXHIBIT J	Lessee Guaranty
EXHIBIT K	Lessor Guaranty
EXHIBIT L	[Intentionally Deleted]
EXHIBIT M	LGS Pipeline Riser Exhibit
EXHIBIT N	Liquids Gathering System Map
EXHIBIT O	Liquids Gathering System Inlet Flange Photograph
EXHIBIT P	Memo of Lease
SCHEDULE 27.1	SPE

 \mathbf{v}

Page

LEASE

This Lease (as amended, restated, replaced, supplemented, or otherwise modified from time-to-time, this 'Lease'') is dated as of December __, 2012 ('Effective Date'') and is made by and between:

Lessor: **PINEDALE CORRIDOR, LP**, a Delaware limited partnership

and

Lessee: ULTRA WYOMING LGS, LLC, a Delaware limited liability company

In consideration of the rents and the other terms, covenants and conditions set forth in this Lease, Lessor leases to Lessee and Lessee leases from Lessor all of Lessor's rights, title, and interests in and to the Leased Property (defined below) upon the following terms and conditions:

ARTICLE I.

FUNDAMENTAL LEASE PROVISIONS

1.1 **Definitions**. For all purposes of this Lease, except as may be expressly set forth herein or unless the context clearly indicates a contrary intent, the following terms have the following definitions:

"Action" means, for any Person, such Person's filing a petition under the Code, or initiating a proceeding under any similar law or statute relating to bankruptcy, insolvency, reorganization, winding up or adjustment of debts.

"Actual Daily Product Volume" means for any calendar month [redacted] transported through the Liquids Gathering System. For the avoidance of doubt, Actual Daily Product Volume shall include [redacted] physically transported through the Liquids Gathering System from Additional Lines, if any, even if such Additional Lines do not become part of the Leased Property as provided in <u>Section 10.1</u>.

"Actual Knowledge" means (a) with respect to any Ultra Entity Person, as of any date, the then current actual knowledge of the President, Senior Vice President of Operations and Chief Financial Officer of Ultra Resources as of such date, and not any implied, imputed or constructive knowledge of such individuals, and without any independent investigation or inquiry having been made or any implied duty to investigate, make any inquiries or review any information, (b) with respect to Pinedale, as of any date, the then current actual knowledge of the Chair of the Board, the President and the Principal Accounting Officer of CorEnergy Infrastructure Trust, Inc. as of such date, and not any implied, imputed or constructive knowledge of such individuals, and without any independent investigation or inquiry having been made or any implied duty to investigate, make any inquiries or review any information, and (c) with respect to any other Person, as of any date, the then current actual knowledge of the Responsible Officers of such individuals, and without any independent investigation or inquiry having been made or any implied duty to investigate, make any inquiries or review any information, and (c) with respect to any other Person, as of any date, the then current actual knowledge of the Responsible Officers of such any implied duty to investigate, and not any implied, imputed or constructive knowledge of such individuals, and without any independent investigation or inquiry having been made or any implied duty to investigate, make any inquiries or review any information. The foregoing qualification of Actual Knowledge shall in no event give rise to any personal liability on the part of any such Person or any other officer or employee of such party or its Affiliates on account of any breach hereunder.

"Additional Lines" means such additional easements and rights of way and such additional gathering lines, distribution or sales lines, and appurtenances that connect to or adjoin the Leased Property but that are, in all cases, either upstream or downstream of the Liquids Gathering System, as Lessee may from time-to-time desire or be required to use, acquire or make.

"Additional Rent" means all amounts, costs, expenses, Losses, liabilities, indemnities and other monetary obligations (including Lessee's obligation to pay any interest at the Default Interest Rate hereunder) which Lessee is required to pay pursuant to the terms of this Lease, other than Base Rent.

"Adjustment Date" means the first day of the month following the month in which the first anniversary of the Effective Date occurs, and every anniversary of such first day of the month thereafter during the Term.

"Adverse Party" means, as of any date, (a) any Person who is, or whose Affiliate is, an adverse party in any then-existing litigation or arbitration involving Lessee or an Affiliate of Lessee, (b) any Person who is, or whose Affiliate is, the potential plaintiff or claimant in any litigation which has been threatened in writing against Lessee or any Affiliate of Lessee, (c) any Person who (i) is a substantial investor in Lessee or any Affiliate of Lessee, and (ii) has a pending proposal to merge with, acquire or takeover Lessee or any Affiliate of Lessee, which merger, acquisition or other takeover shall not have been approved by the board of directors of Lessee or such Affiliate or otherwise be perceived by Lessee or such Affiliate to be hostile to the management of Lessee or such Affiliate, and (d) any Affiliate of each Person in subsection (c). Within ten (10) Business Days after written request therefor from Lessor, Lessee shall advise Lessor if a Person(s) named by Lessor in such notice is an Adverse Party within the scope of this definition.

"Affiliate" has the meaning ascribed to such term in Rule 12b-2 of the General Rules and Regulations under the Exchange Act and, with respect to a Person, includes any Subsidiary of that Person.

"Agreed FMV Rent" has the meaning given in Section 25.1(b).

"Applicable Leased Assets" has the meaning given in Section 25.4(b).

"Applicable Legal Requirements" mean all statutes, ordinances, regulations and codes of any Governmental Authority having jurisdiction, including, without limitation, zoning, health, fire, safety and building codes, applicable to the Leased Property (or portion thereof at issue), applicable to Lessee (with respect only to Lessee's (rather than Lessor's) obligations hereunder regarding compliance with Applicable Legal Requirements), or applicable to Lessor (with respect only to Lessor's (rather than Lessee's) obligations hereunder regarding compliance with Applicable Legal Requirements).

"Auditor's Report" means, with respect to financial statements or information of Lessee Guarantor or Lessee required to be delivered pursuant tsection 26.7(b)(ii) (a) the written report of the auditor for Lessee Guarantor or Lessee, as applicable, with respect to such financial statements or information (excluding any auditor's report on internal controls), manually executed by such auditor, and (b) a manually executed consent of such auditor to the inclusion of such auditor's report in filings to be made by Lessor or Lessor Guarantor with the Securities and Exchange Commission.

"Bankruptcy Action" means, with respect to any Person, if such Person:

(a) makes an assignment for the benefit of creditors;

(b) files a voluntary petition in bankruptcy;

(c) is adjudged a bankrupt or insolvent, or has entered against it an order for relief, in any bankruptcy or insolvency proceedings;

(d) consents to, orchestrates or participates in the orchestration of, or files a petition or answer seeking for itself, any reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any statute, law or regulation;

(e) files an answer or other pleading admitting or failing to contest the material allegations of a petition filed against it in any bankruptcy or insolvency proceeding;

(f) seeks, consents to, orchestrates or participates in the orchestration of, or acquiesces in the appointment of a trustee, receiver, liquidator, sequestrator, custodian or any similar official of or for such Person or of all or any substantial part of its properties;

(g) fails to cause the dismissal of any proceeding against such Person seeking reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any statute, law or regulation within one hundred twenty (120) days after the commencement of such proceeding;

(h) fails to cause the vacation or stay of any appointment of a trustee, receiver or liquidator of such Person or of all or any substantial part of its properties made without such Person's consent or acquiescence, within ninety (90) days after such appointment and, if stayed within such time period, the vacation of such appointment within ninety (90) days after the expiration of such stay; or

(i) takes any action in furtherance of any of the foregoing.

"Base Daily Product Volume" means the [redacted] transported through the Liquids Gathering System for the fiscal quarter ending December 31, 2011, being [redacted] per day.

"Base Rent" means (a) during the Initial Term, (i) prior to the first Adjustment Date, \$1,666,667 per month, and (ii) from and after the first Adjustment Date, [redacted] and (b) during any Renewal Term, the amount determined pursuant to Section 25.1 hereof.

"Beneficial Owner" a Person shall be deemed the "Beneficial Owner" of, and shall be deemed to "beneficially own" and have "beneficial ownership of" any equity interests or other securities:

(a) which such Person or any of such Person's Affiliates, directly or indirectly, has the right to acquire (whether such right is exercisable immediately or only after the passage of time) pursuant to any agreement, arrangement or understanding (whether or not in writing) or upon the exercise of conversion rights, exchange rights, rights, warrants or options, or otherwise;

(b) which such Person or any of such Person's Affiliates, directly or indirectly, has the right to vote or dispose of or has "beneficial ownership" of (as determined pursuant to Rule 13d-3 of the General Rules and Regulations under the Exchange Act), including pursuant to any agreement, arrangement or understanding, whether or not in writing; provided, however, that a Person shall not be deemed the "Beneficial Owner" of, or to "beneficially own," any equity interest or other security under thisubsection (b) as a result of an agreement, arrangement or understanding to vote such equity interest or other security if such agreement, arrangement or understanding: (i) arises solely from a revocable proxy given in response to a public proxy or consent solicitation made pursuant to, and in accordance with, the applicable provisions of the General Rules and Regulations under the Exchange Act, and (ii) is not also then reportable by such Person on a Schedule 13D under the Exchange Act (or any comparable or successor report); or

(c) which are "beneficially owned," directly or indirectly, by any other Person (or any Affiliate thereof) with which such Person (or any of such Person's Affiliates) has any agreement, arrangement or understanding (whether or not in writing), for the purpose of acquiring, holding, voting (except pursuant to a revocable proxy as described in the proviso to subsection (b) above) or disposing of any voting equity interests or other securities of Lessor,

provided, however, that (A) a Person engaged in business as an underwriter of securities shall not be deemed the "Beneficial Owner" of or to "beneficially own" any securities acquired through such Person's participation in good faith in a firm commitment underwriting, and (B) the existence of rights of first bid, pre-emptive rights, drag along rights, or tag along rights set forth in the Lessor's Governing Documents shall not themselves cause one Person to be deemed the Beneficial Owner of Lessor Equity Interests held by one or more other Equity Investors.

"BLM" means the United States Department of the Interior Bureau of Land Management, Wyoming State Office.

"BLM Easements" means the easements, rights of way and agreements listed in Exhibit A, the rights of the holder thereunder having been assigned to Lessee by Ultra Resources, and as subsequently assigned to Lessor by Lessee by assignment documents dated as of the Effective Date.

"Burdensome Buyout Lease Termination Date" has the meaning given in Section 25.5(c).

"Burdensome Buyout Notice" has the meaning given in Section 25.5(a).

"Burdensome Buyout Response Notice" has the meaning given in Section 25.5(b).

"Burdensome Event" means any one or more of the following events or circumstances: (a) Lessee has been or will be obligated to make indemnity payments or expenditures or take Corrective Action in respect of Lessee's Environmental Liabilities under <u>Article XXII</u> of this Lease in excess of fifty percent (50%) of the Insured Value and any such amounts payable in the future would be avoided by termination of this Lease, (b) Lessee has been or will be obligated to make payments or expenditures as a result of compliance with changes in Applicable Legal Requirements, practices, or standards under this Lease in excess of fifty percent (50%) of the Insured Value and any such amounts payable in the future would be avoided by termination of this Lease, (c) there is a change in Applicable Legal Requirements, or industry practices or standards which (i) requires modification of any part of the Leased Property, the cost of which Lessee reasonably anticipates to be in excess of fifty percent (50%) of the Insured Value, or (ii) makes Lessee's use of the Leased Property under the terms of the Lease uneconomic, or (d) Lessee's use of the Leased Property for its intended purposes becomes, in Lessee's sole but reasonable judgment, uneconomic.

"Business Day" means any day that is not a Saturday, Sunday or other day on which commercial banks in New York City or Houston, Texas, are authorized or required by law to remain closed. If the last day of any time period under this Lease, or the last day for performance of any obligation, or for giving any notice, or for taking any other action under this Lease falls on a day that is not a Business Day, then the last day of such time period shall be extended to the first day thereafter that is a Business Day.

"Capital Improvements" means such alterations, additions and replacements to any of the Improvements or Personal Property as Lessee may from time-to-time desire or be required to make.

"Casualty Date" means the date fire or other casualty damages any portion of the Improvements or Personal Property.

"Casualty Proceeds" means the insurance proceeds actually received by or payable to Lessee with respect to a fire or other casualty of the Leased Property, but excluding from such insurance proceeds (a) Lessee's reasonable third party costs and expenses (including reasonable attorneys' fees and expenses) incurred in collecting same, (b) as of any date, proceeds previously paid to Lessee and used to repair, restore or replace the Leased Property as a result of such fire or other casualty, and (c) the proceeds of any business interruption or similar insurance with respect to such fire or other casualty.

"Casualty Response Notice" has the meaning assigned to such term in Section 15.1(c).

"Casualty Termination Date" has the meaning assigned to such term in Section 15.1(d).

"Central Gathering Facility" or "CGF" means each of the four locations within the Liquids Gathering System at which condensate, produced water, and associated natural gas is collected and separated for sale and transport off the subject location or alternatively, in the case of associated natural gas, used for

fuel or flared on the subject location, as of the Effective Date and as such locations may be modified after the Effective Date as permitted by this Lease. A legal description of each of CGF#1, CGF#2, CGF#3 and CGF#4 as of the Effective Date is set out on the attached Exhibit B.

CGF#1: means the Central Gathering Facility identified in the "Identification" header on Exhibit B attached hereto as "Central Gathering Facility #1".

CGF#2: means the Central Gathering Facility identified in the "Identification" header on Exhibit B attached hereto as "Central Gathering Facility #2".

CGF#3: means the Central Gathering Facility identified in the "Identification" header on Exhibit B attached hereto as "Central Gathering Facility #3".

CGF#4: means the Central Gathering Facility identified in the "Identification" header on Exhibit B attached hereto as "Central Gathering Facility #4".

"Code" means Title 11 of the United States Code, 11 U.S.C. Sec. 101 et seq., as amended.

"Competitor" means, as of any date, (a) any Person engaged in the business of exploration or production of oil or gas, and (b) for so long as an Ultra Entity Person is Lessee, any Person for whom the performance of the types of services commonly known as the midstream services, which services generally include gathering, fractionating, storing, transporting and marketing of oil, natural gas and natural gas liquids, is the primary or a significant source of revenue, and (c) any Affiliate of any Persons described in subsections (a) – (b), above.

"Condensate" means liquid hydrocarbons.

"Condensate Loading Point" means the condensate loading coupling off each condensate storage tank located on a Central Gathering Facility as of the Effective Date, and as the Condensate Loading Points may be modified as permitted by this Lease.

"Condensate Terminus Point" means the (a) inlet flange for each LACT Unit Facility, and (b) each Condensate Loading Point as of the Effective Date, and as they may be modified as permitted by this Lease.

"Confidential Information" means (a) all Proprietary Information, and (b) all other information furnished to Lessor by or on behalf of Lessee, Lessee Guarantor or any of their respective Affiliates prior to, on or after the Effective Date and designated as confidential. Notwithstanding the foregoing, Confidential Information shall not include information that is publicly available other than as a result of actions in violation of <u>Section 26.10</u> hereof.

"Control" or any derivation thereof has the meaning set forth in Rule 12b-2 under the Exchange Act. "Controlling" and "Controlled" have meanings correlative thereto.

"Controlling Lease Rights" means the voting or other rights a Person possesses or exercises, directly or indirectly, which allows or would allow such Person to Control a vote on, consent to, approval of or blocking of decisions made by Lessor with respect to actions, decisions, approvals, waivers, consents, declarations of default, exercise of remedies or other matters under this Lease, whether through ownership of securities, the ability to exercise voting power, by contract, arrangement, understanding, course of conduct or otherwise; provided that for purposes of Section 17.5(a)(v), Sections 17.5(b) and (c), and Section 27.1(b), each Disqualified Person shall be deemed to Control the Lessor Equity Interests Beneficially Owned by each other Disqualified Person.

"Corrective Action" means environmental investigation and/or remediation, including Phase II testing, sampling, engineering, consulting, reporting, active remediation, passive remediation, monitoring and risk assessment or any combination of these activities.

"County and State Permits" means Lessor's right, title and interest, as assignee, under that certain Partial Assignment of County Permits and State License from Ultra Wyoming, Inc., to Lessor, dated as of the Effective Date and recorded in the Official Records of Sublette County, Wyoming, with respect to the following:

(a) Application for Utility Crossing Sublette County, Wyoming, with Ultra Resources, Inc., dated 3/26/2010, to locate utility distribution or transmission lines within a County Road ROW, County Rd No 23-106, Legal: SW 1/4 SE ¹/₄, Section 14, Township 31N, Range 109W.

(b) Application for Utility Crossing Sublette County, Wyoming, with Ultra Resources, Inc., dated 3/26/10, to locate utility distribution or transmission lines within a County Road ROW, County Rd No 23-106, Legal: NE 1/4 NE ¼, Section 13, Township 31N, Range 109W.

(c) Application for Utility Crossing Sublette County, Wyoming, with Ultra Resources, Inc., dated 3/26/10, to locate utility distribution or transmission lines within a County Road ROW, County Rd No 23-106, Legal: SE 1/4 NW ¼, Section 13, Township 31N, Range 109W.

(d) Application for Utility Crossing Sublette County, Wyoming, with Ultra Resources, Inc., dated 3/26/10, to locate utility distribution or transmission lines within a County Road ROW, County Rd No 23-106, Legal: SE 1/4 SE ¼, Section 7, Township 31N, Range 108W.

(e) License dated April 21, 2010, between the Transportation Commission of Wyoming and its Wyoming Department of Transportation and Ultra Resources, Inc., regarding four fiberspar pipelines with respect to WYDOT Highway 351, and located in Section 4, Township 30N, Range 108W, Milepost 18.715.

"Current Lease Term End" means the end of the then current Term assuming that there will not be a Renewal Term at the end thereof.

"Debtor Relief Law" means the Bankruptcy Code of the United States of America, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief laws of the United States of America or other applicable jurisdictions from time-to-time in effect and affecting the rights of creditors generally.

"Default Interest Rate" means, for any period, (a) with respect to any obligation of (i) any Ultra Entity Person, and (ii) any other Lessee or Lessee Guarantor hereunder if as of the date of such Lessee's acquisition of the leasehold estate under this Lease either such Lessee or Lessee Guarantor is an Investment Grade Person, and their respective Affiliates, the lesser of (A) the annual rate of two percent (2%) over the average base, non-default rate of interest for such period (and without imputation of any gross-up components or default interest), under such Person's Indebtedness which is Material Debt and which is held by a third party not an Affiliate of such Person, or, if such Person does not have any such Material Debt, then the Prime Rate for such period, and (B) the highest rate permitted by Applicable Legal Requirements, and (b) with respect to any obligation of Pinedale or any of its Affiliates, the lesser of (i) the annual rate of two percent (2%) over the average base, non-default rate of interest for such period (and without imputation of any gross-up components or default interest), under the Permitted Indebtedness of such Person which is held by a third party not an Affiliate of such Person or, if such Person of any gross-up components or default interest), under the Permitted Indebtedness of such Person which is held by a third party not an Affiliate of such Person or, if such Person does not have any such Permitted Indebtedness, then the Prime Rate for such period, and (ii) the highest rate permitted by Applicable Legal Requirements, and (c) with respect to any other Lessee, Lessee Guarantor, Lessor or Lessor Guarantor with respect to their respective obligations under or related to this Lease, the lesser of (i) the annual rate of six

percent (6%) over the Prime Rate for such period, and (ii) the highest rate permitted by Applicable Legal Requirements.

"Designation Notice" has the meaning given in Section 25.1(c).

"Discount Rate" means the greater of (a) the interest rate (on the date of termination of this Lease by Lessor pursuant to its remedies under<u>Section 23.2(a)(i)</u>) on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers, as published by the Federal Reserve Bank of New York, <u>plus</u> one half of one percent (0.5%), and (b) the Prime Rate on the date of termination of this Lease by Lessor pursuant to its remedies under <u>Section 23.2(a)(i)</u>.

"Disqualified Person" means, as of any date, (a) so long as an Ultra Entity Person is the Lessee, each Ultra Designated Party, (b) each Competitor, and (c) each Adverse Party. Notwithstanding the foregoing, for so long as (i) both (A) Pinedale Corridor, LP, is the Lessor under this Lease, and (B) Pinedale Corridor, LP, is owned, in whole or in part, by direct or indirect subsidiaries of CorEnergy Infrastructure Trust, Inc., then no direct or indirect wholly owned subsidiary of CorEnergy Infrastructure Trust, Inc., shall be a Disqualified Person for purposes of this Lease, and (ii) both (A) Pinedale Corridor, LP, is the Lessor under this Lease, and (B) Pinedale Corridor, LP, is owned, in whole or in part, by direct or indirect subsidiaries of Prudential Financial, Inc., then no direct or indirect wholly owned subsidiary of Prudential Financial, Inc., shall be a Disqualified Person for purposes of this Lease.

"Due Date for Other Additional Rent' has the meaning given in Section 5.4.

"Easement Rights" means, collectively (a) the right, title and interest in the BLM Easements conveyed to Lessor pursuant to the Purchase Agreement, (b) the right, title and interest in the Jensen Easements conveyed to Lessor pursuant to the Purchase Agreement, and (c) the right, title and interest conveyed to Lessor pursuant to the Easement Agreement and Transfer of Improvements (Nerd Farm) pursuant to the Purchase Agreement (but exclusive of the Improvements transferred pursuant to such Easement Agreement and Transfer of Improvements).

"Easements" means the BLM Easements, the Jensen Easements and the Nerd Farm Easement.

"Effective Date" has the meaning specified in the preamble.

"Effective Date Recorded Documents" means all restrictions and charges created or imposed pursuant to documents and instruments recorded in the Official Records against the Leased Property as of the time of conveyance of the Leased Property by Ultra Wyoming, Inc., to Lessor pursuant to the Purchase Agreement.

"Environmental Laws" means all federal, state, and local laws, statutes, rules, regulations, orders, judgments, ordinances, codes, injunctions, decrees, and other legally enforceable requirements relating to (a) pollution or protection of the environment or natural resources, (b) any actual or threatened depositing, spilling, leaking, pumping, pouring, placing, emitting, discarding, abandoning, emptying, discharging, migrating, injecting, escaping, leaching, dumping or disposing into the environment of, or any exposure to, any Hazardous Materials, or (c) the generation, manufacture, processing, distribution, use, treatment, storage, transport, disposal or handling of any Hazardous Materials; including the federal Comprehensive Environmental Response, Compensation and Liability Act, the Superfund Amendments and Reauthorization Act, the Resource Conservation and Recovery Act, the Clean Air Act, the Clean Water Act, the Safe Drinking Water Act, the Toxic Substances Control Act, the Oil Pollution Act of 1990, the Federal Hazardous Materials Transportation Law, the Marine Mammal Protection Act, the Endangered Species Act, and the National Environmental Policy Act.

"Equity Investor" means any Person that beneficially owns any Equity Investor Interest.

"Equity Investor Agreement" means (a) that certain Equity Investors Agreement entered into as of the Effective Date among Lessor, Lessee, CorEnergy Infrastructure Trust, Inc., Pinedale GP, Inc., and Ross Avenue Investments, LLC, and as the same may be subsequently amended, supplemented or restated, and (b) each subsequent Equity Investors Agreement to be entered into after the Effective Date between (i) Lessor, (ii) Lessee, (iii) each owner of a Lessor Equity Interest, and (iv) the respective applicable Affiliates of Lessor and each Lessor Equity Interest Owner to the extent such Affiliates is (or Affiliates are) required to guaranty the obligations of such Person(s) under the Equity Investor Agreement as a result of the application of the definition of the term "Permitted Controller of Lease Rights" to such Person(s), in substantially the form of Exhibit C hereto (but with (A) the last sentence of the defined term "Transfer" therein deleted, and (B) the last sentence of Section 2.7(b) therein and Exhibit D thereto shall be deleted), and in each case as the same may be subsequently amended, supplemented or restated.

"Equity Investor Interest" means each beneficial ownership interest in equity of the Lessor.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"Estimates" has the meaning given in Section 25.1(c).

"Excess Product Volume Factor" means for any calendar month during the Term from and after the first Adjustment Date, the greater of [redacted].

"Exchange Act" means the Securities Exchange Act of 1934.

"Excluded Matters" has the meaning given in Section 6.3.

"Exclusions from Lessee's Indemnification Obligations" has the meaning given in Section 21.1(a).

"Fair Market Rent" means, with respect to the Leased Property during the proposed Renewal Term (or, for purposes of Lessor's Lease termination remedy during the continuance of a Level 1 Lessee Default, during the portion of the Term which would have occurred after termination of this Lease as a result of such Level 1 Lessee Default), that rental rate that is a fair market value rental rate that would apply assuming that the Leased Property is exposed on the open market as a rental property for the proposed Renewal Term (or, with respect to termination of this Lease by Lessor as a result of a Level 1 Lessee Default, during the portion of the Term which would have occurred after such termination of this Lease), taking into account all relevant factors. The Fair Market Rent will be assumed to be a monthly rent payable in advance on the first day of each calendar month. The Fair Market Rent may, but is not required to, follow the model of the current rent structure, i.e., having Base Rent.

"Fair Market Rent Determination Notice" has the meaning given in Section 25.1(c).

"Fair Market Value" or "fair market value" means the price that would be agreed on between a willing buyer and a willing seller, with neither being required to act, and both having reasonable knowledge of the relevant facts.

"Flash Gas Terminus Point" means (a) with respect to CGF#1, CGF #3, and CGF #4, the dehydrator located on each such Central Gathering Facility; and (b) with respect to CGF #2, the gas outlet flange on the Effective Date on (i) each three-phase heated separator located on CGF #2, (ii) each condensate

storage tank located on CGF #2, and (iii) each water storage tank located on CGF #2, in each case, as of the Effective Date, and as they may be modified as permitted by this Lease.

"Flow and Volume Information" means the flow and volume information of the type to be furnished by Lessee pursuant to Section 26.7(b)(i).

"Force Majeure" has the meaning given in Section 26.2.

"Four Quarters Lessor EBITDA" means, on any date, Lessor EBITDA for the four fiscal quarters ending on such date, or if such date is not the last day of Lessor's fiscal quarter, ending on the most recent March 31, June 30, September 30 or December 31, in all events assuming payment of Base Rent by Lessee as required by this Lease; provided that if Lessor elects to pay for a Non-Removable Addition or a Removable Addition under Article X, Four Quarters Lessor EBITDA for any four fiscal quarter period shall be calculated on a pro forma basis to give effect to any such Non-Removable Addition or Removable Addition completed during such period as if such Non-Removable Addition or Removable Addition had been completed on the first day of such period.

"GAAP" means generally accepted accounting principles in effect from time-to-time in the United States of America.

"Gathering Agreement' means the Gathering Agreement in the form attached hereto as Exhibit F-2.

"Good Condition and Repair" means good condition and repair consistent with (a) Lessee's past practices and industry standards, and (b) the condition and repair that a reasonably prudent operator would maintain for a gathering system of similar size, nature, use, age and location.

"Governing Documents" means (a) with respect to Pinedale Corridor, LP, the First Amended and Restated Limited Partnership Agreement of Lessor dated as of the Effective Date, and the Certificate of Limited Partnership of Lessor executed on November 30, 2012, and filed with the Delaware Secretary of State on December 3, 2012, (b) with respect to a Person that is a limited partnership, its limited partnership agreement and certificate of limited partnership, (c) with respect to a Person who is a general partnership, its partnership agreement, (d) with respect to a Person who is a limited liability company, its limited liability company agreement, together with its certificate of formation and any operating agreement, regulations and similar agreements or documents of such Person, (e) with respect to a Person who is a corporation, its articles of organizational agreements, together with any shareholders agreement and similar agreements and documents of such Person, (f) in the case of any other form of entity, its organizational agreements, certificates and documents, and (g) in each case, with respect to subsections (a) –(f), as such agreements, certificates and documents may be amended and/ or restated from time-to-time, subject however to the limitations on amendments set forth herein.

"Governmental Authority" means any governmental authority, agency, department, commission, bureau, board, instrumentality, court or quasi-governmental authority having jurisdiction or supervisory or regulatory authority over any of the Leased Property, applicable to Lessee (with respect only to Lessee's (rather than Lesser's) obligations hereunder regarding compliance or interaction with Governmental Authority), or applicable to Lessor (with respect only to Lessor's (rather than Lessee's) obligations hereunder regarding compliance or interaction with Governmental Authority).

"Hazardous Materials" means (a) any substance that is designated as a hazardous waste, solid waste, hazardous material, pollutant, contaminant or toxic or hazardous substance, as those terms are used in the Comprehensive Environmental Response, Compensation, and Liability Act, (b) any petroleum, petroleum hydrocarbons, petroleum products, petrochemical products and any components, fractions or derivatives

thereof, any oil or gas exploration or production waste, and any natural gas, synthetic gas, and any mixtures thereof, and (c) radioactive materials or polychlorinated biphenyls.

"Impositions" mean, collectively, all charges, fees and expenses imposed on the Leased Property under any Easement or any other Record Agreement during and accruing with respect to the Lease Term, but excluding in any event all Excluded Matters.

"Improvements" means all of the improvements and fixtures used directly in connection with the Liquids Gathering System including, without limitation, any and all surface and/or subsurface pipelines, surface and/or subsurface machinery and equipment, line pipe, pipe connections, fittings, flanges, welds, or other interconnections, valves, control and monitoring equipment, cathodic or electrical protection units, by-passes, regulators, drips, treating equipment, dehydration equipment, separation equipment, processing equipment, condensate and water storage tanks and other storage facilities, generators, gas compressors, vapor recovery units, combustors, flares, storage sheds, towers, gas and electric fixtures, radiators and heaters, in each case that are downstream of the inlet flange to each LGS Pipeline Riser and that are upstream of each LGS Terminus Point, including any of the foregoing described on the attached <u>Exhibit H</u>, but specifically excluding (a) any of the facilities, improvements and fixtures located downstream of each LGS Terminus Point, except in each case to the extent explicitly described on <u>Exhibit H</u>. The term "Improvements" includes all of the improvements and fixtures which are a part of the Liquids Gathering System as described herein, regardless of whether they are included or properly described in <u>Exhibit H</u>.

"Indebtedness" means with respect to a Person, such Person's (a) liabilities for borrowed money, (b) liabilities for the deferred purchase price of property acquired by it (excluding accounts payable arising in the ordinary course of businesses), (c) obligations that are required to be accounted for as capital leases on a balance sheet under GAAP (and the amount of such obligations shall be the capitalized amount thereof determined in accordance with GAAP), and (d) guaranty obligations with respect to liabilities of another Person of the type described in the preceding subsections (a)-(c).

"Indemnified Party" means a Lessor Indemnified Party or a Lessee Indemnified Party, as applicable.

"Indemnifying Party" has the meaning given in Section 21.3.

"Independent Appraiser" means an independent valuation firm with at least five (5) years' experience in undertaking valuations of gas or other commodity pipeline properties.

"Independent Director" means in the case of a corporation, a natural person provided by a nationally-recognized company that provides professional independent directors who has not been, and during the continuation of his or her service as Independent Director is not, directly or indirectly:

(a) an employee, manager, stockholder, director, member, partner, officer, attorney or counsel of the corporation or any of its Affiliates (other than his or her service as an Independent Director of the corporation);

(b) a creditor, customer of, or supplier or other Person who derives any of its purchases or revenues from its activities with the corporation or any of its shareholders or Affiliates (other than his or her service as an Independent Director if such Person has been provided by a nationally-recognized company that provides professional independent managers);

(c) a Person controlling or under common control with any such employee, manager, stockholder, director, member, partner, officer, attorney, counsel, customer, supplier or other Person; or

(d) any member of the immediate family (including a grandchild or sibling) of a person described in subsections (a), (b) or (c) immediately above.

A natural person who otherwise satisfies the foregoing definition shall not be disqualified from serving as an Independent Director of the corporation because such person is an independent director of a "Special Purpose Bankruptcy Remote Entity" affiliated with the corporation that does not own a direct or indirect equity interest in the corporation if such individual is an independent director provided by a nationally-recognized company that provides professional independent directors.

"Independent Manager" means in the case of a limited liability company, a natural person provided by a nationally-recognized company that provides professional independent managers who has not been, and during the continuation of his or her service as Independent Manager is not, directly or indirectly:

(a) an employee, manager, stockholder, director, member, partner, officer, attorney or counsel of the limited liability company or any of its Affiliates (other than his or her service as an Independent Manager of the limited liability company);

(b) a creditor, customer of, or supplier or other Person who derives any of its purchases or revenues from its activities with the limited liability company or any of its members or Affiliates (other than his or her service as an Independent Manager if such Person has been provided by a nationally-recognized company that provides professional independent managers);

(c) a Person controlling or under common control with any such employee, manager, stockholder, director, member, partner, officer, attorney, counsel, customer, supplier or other Person; or

(d) any member of the immediate family (including grandchildren or siblings) of a person described in subsections (a), (b) or (c) immediately above.

A natural person who otherwise satisfies the foregoing definition shall not be disqualified from serving as an Independent Manager of the limited liability company because such person is an independent manager of a "Special Purpose Bankruptcy Remote Entity" affiliated with the limited liability company that does not own a direct or indirect equity interest in the limited liability company if such individual is an independent manager provided by a nationally-recognized company that provides professional independent managers.

"*Index*" means the Consumer Price Index which is designated for the applicable month of determination as the United States City Average for All Urban Consumers, All Items, Not Seasonally Adjusted, with a base period equaling 100 in 1982-1984, as published by the United States Department of Labor's Bureau of Labor Statistics or any successor agency.

"Index Adjustment" means the quotient (expressed as a percentage) of (a) the change, if any, between (i) the average of the Index for the twelve months ended two months prior to the applicable Adjustment Date, and (ii) the average of the Index for the twelve months ended fourteen months prior to such Adjustment Date (the "Base Index"), and (b) the Base Index. If the statistics are not available or if publication of the Index is modified or discontinued in its entirety, the Index Adjustment shall be determined on the basis of an index chosen by Lessor as a comparable and recognized index of the purchasing power of the United States consumer dollar published by the United States Department of Labor or other governmental agency. If the Index contemplated herein is not reported for the months required for the calculation set forth above, the parties agree to use the Index reported for the month(s) nearest preceding the month(s) required for such calculation.

"Initial Lessor Loan" means the Indebtedness evidenced and secured by the Initial Lessor Loan Documents.

"Initial Lessor Loan Documents" means the documents and instruments listed on Exhibit I.

"Initial Term" means that period beginning on the Effective Date and ending on the last day of the 180^h full calendar month thereafter.

"Insured Value" means the full replacement cost of the Improvements and Personal Property as determined (a) if Lessee carries fire and casualty property insurance with respect to the Leased Property, by Lessee's property insurer from time-to-time for purposes of issuing the property insurance required to be carried by Lessee under this Lease, and (b) if Lessee self-insures for fire and casualty loss or damage to the Leased Property and has no third party property insurer for such loss or damage to the Leased Property, by Lessee, determined in good faith.

"Investment Grade Person" means (a) a Person that has a long term issuer corporate credit rating or long-term unsecured debt rating of "BBB-" or higher from Standard & Poor's or Fitch, or "Baa3" or higher from Moody's Investor Service Inc., 2 or better from the National Association of Insurance Commissioners or an equivalent rating from another nationally recognized rating agency, or (b) an Affiliate of such Person whose obligations under this Lease are guaranteed by such Person.

"Involuntary" and "Involuntarily" includes, with respect to a Transfer or Lien, any transaction, proceeding or action by or in which Lessor or an Equity Investor is involuntarily deprived or involuntarily divested of any right, title or interest in and to its applicable property, rights or interests (including any seizure under levy of attachment or execution, transfer in connection with bankruptcy or other court proceeding to a trustee or receiver or other officer or agency or any transfer to a state or to a public officer or agency pursuant to any statute pertaining to escheat or abandoned property).

"Jensen Easements" means the following easements: (a) that certain Grant of Pipeline Easements dated June 24, 2010, recorded in Book 95, Page 423 of the Official Public Records, from Mary Kay Jensen, as grantor, to Ultra Resources, as grantee, and (b) that certain Grant of Pipeline Easements dated June 24, 2010, in Book 95, Page 418 of the Official Public Records, from Mary Kay Jensen, as personal representative of the Estate of John Wayne Jensen, individually, as grantor, to Ultra Resources, as grantee, as both such easements were assigned by Ultra Resources to Lessee, and as such easements were further partially assigned to Lessor pursuant to that certain Partial Assignment of Easements and Transfer of Improvements (LGS) [Jensen Easements] from Lessee, as assignor, to Lessor, as assignee, dated as of the Effective Date and recorded in the Official Public Records.

"LACT Unit Facilities" means the condensate pump, meter and sales point as well as the associated equipment downstream of the inlet risers to any Lease Automated Custody Transfer Unit located on a Central Gathering Facility as of the Effective Date, and as they may be modified as permitted by this Lease

"Land" means the land underlying, subject to and covered by the Easement Rights.

"Lease" has the meaning given in the preamble.

"Lease Assignment" means a Transfer by Lessee of its interest under this Lease or its interest in the Leased Property hereunder but excluding in any event any change or other circumstance which is a Non-Consent Matter.

"Leased Property" means the Liquids Gathering System, the Personal Property and the County and State Permits.

"Lessee" has the meaning given in the preamble.

"Lessee Casualty Termination Notice" has the meaning given in Section 15.1(b).

"Lessee Condemnation Proceeds" has the meaning given in Section 16.1(a).

"Lessee Taking Termination Notice" has the meaning given in Section 16.2(b).

"Lessee Environmental Liabilities" means Losses arising or resulting from (a) the actual presence, use, storage, generation or Release of any Hazardous Materials on, under, from or at the Leased Property or any portion thereof during the Lease Term for which Lessee or Lessor has any legal obligation, (b) allegations made in writing by a third party not Lessor, another Lessor Party or a Lessor Lender (but only so long as neither Lessor, any Lessor Party nor any Lessor Lender has affirmatively caused such third party to raise such allegation unless such allegation is required to be made by such third party under applicable law) regarding the presence, use, storage, generation or Release of any Hazardous Materials on, under, from or at the Leased Property or any portion thereof during the Lease Term for which Lessee or Lessor has any legal obligation, or (c) any violation of Environmental Laws occurring during the Lease Term at or adversely affecting any portion of the Leased Property for which Lessee or Lessor has any legal obligation, but excluding in any case with respect to <u>subsections (a), (b)</u> and (c), all Lessor Environmental Liabilities, all Excluded Matters and the matters or claims arising or occurring prior to the Effective Date, all of which such pre-Effective Date matters and claims shall be governed exclusively by the terms and conditions of such Purchase Agreement.

"Lessee Event of Default' has the meaning given in Section 23.1.

"Lessee Guarantor" means Ultra Petroleum and any successor Lessee Guarantor executing a Lessee Guaranty until any such Person is released from further liability upon a Lease Assignment as provided in Section 17.4. For avoidance of doubt, the term "Lessee Guarantor" does not include Ultra Resources.

"Lessee Guaranty" means that certain Lessee Guaranty executed by Ultra Petroleum and dated as of the Effective Date and any subsequent Lessee Guaranty executed after the Effective Date and in substantially the form of Lessee Guaranty attached as <u>Exhibit J</u> unless and until the Lessee Guarantor under such Lessee Guaranty is released to the extent provided in <u>Section 17.4</u>. For avoidance of doubt, the term "Lessee Guaranty" does not include the Resources Guaranty.

"Lessee Indemnified Parties" means Lessee, Lessee Guarantor, Ultra Resources and their respective Related Parties (solely in their capacity as such), and permitted successors and assigns of any of the foregoing, including any permitted successors by merger, consolidation or acquisition of all or substantially all of the Lessee's rights under this Lease.

"Lessee Other Activities" shall have the meaning assigned to such term in Section 3.3.

"Lessee Party" means Lessee, any direct or indirect owner of Lessee, and their respective officers, employees, agents, servants or assignees.

"Lessee Property" means (a) the Permitted Capital Improvements and Additional Lines which Lessor does not purchase and which, accordingly, do not become a part of the Leased Property during the Lease Term as provided in <u>Section 10.1</u>, and (b) the other real and personal property now or hereafter owned or used by Lessee or its Affiliates in connection with operation of the Leased Property upstream or downstream of the Liquids Gathering System by Lessee or its Affiliates or in connection with Lessee Other Activities.

"Lessee Termination and Purchase Offer Rights" means Lessee's Lease termination and purchase offer rights under Article XV (Casualty), Article XVI (Condemnation) and Section 25.5 (Burdensome Buyout).

"Lessee's Possession" means, with respect to information, records, property, documents and materials, only such information, records, property, documents and materials as may be in the actual possession or control of Lessee or its Affiliates, and without including any information or materials in the possession or control of a third Person or any agent of Lessee or its Affiliates.

"Lessor" has the meaning given in the preamble.

"Lessor EBITDA" means the sum of Lessor's net income plus, to the extent deducted in the calculation of net income, (a) income taxes, (b) interest expense, (c) amortization and depreciation, (d) Level 1A Lessor Default Payment Amount, and (e) amounts offset against Base Rent under Section 23.4(a)(vii). Lessor's net income, interest expense and amortization and depreciation shall be determined in accordance with GAAP.

"Lessor Environmental Liabilities" means Losses arising or resulting from any of the following (a) the actual presence, use, storage, generation or Release of any Hazardous Materials on, under, from or at the Leased Property or any portion thereof during the Lease Term if and to the extent caused by or resulting from the negligence or willful misconduct of Lessor, any other Lessor Party or any Lessor Lender, (b) allegations made in writing by a third party not Lessee or another Lessee Party (but only so long as neither Lessee nor any Lessee Party has affirmatively caused such third party to raise such allegation unless such allegation is required to be made by such third party under applicable law) regarding the presence, use, storage, generation or Release of any Hazardous Materials on, under, from or at the Leased Property or any portion thereof during the Lease Term for which Lessee or Lessor has any legal obligation, if and to the extent caused by the negligence of willful misconduct of Lessor, any other Lessor Party or any Lessor has any legal obligation if and to the extent caused by the negligence or willful misconduct of Lessor Party for which Lessee or Lessor has any legal obligation if and to the extent caused by the negligence of weight of the Leased Property for which Lessee or Lessor has any legal obligation if and to the extent caused by the negligence of Lessor, any other Lessor Lender, but excluding in any case with respect to <u>subsections (a), (b)</u> and (c), all Lessee Environmental Liabilities.

"Lessor Equity Interests" means the equity interests issued by Lessor including, as applicable, any general or limited partnership interests, limited liability company membership interests, common or preferred stock, or other interest which would be classified as equity on Lessor's balance sheet under GAAP and any right to acquire such equity interests in Lessor, including upon conversion or exercise of securities. References herein to a percentage of Lessor Equity Interests means Lessor Equity Interests which confer on the holder thereof the right to vote the applicable percentage of votes for the election of directors or similar governing body, or to receive the applicable percentage of dividends, distributions, including liquidating distributions or other payments with respect to the Lessor Equity Interests.

"Lessor Equity Interest Owner" means each owner of a Lessor Equity Interest.

"Lessor Event of Default' has the meaning given in Section 23.3.

"Lessor Governing Document Required Terms" has the meaning given in Section 27.1(e).

"Lessor Guarantor" means CorEnergy Infrastructure Trust, Inc., a Maryland corporation (formerly known as Tortoise Capital Resources Corporation), and any successor Lessor Guarantor executing a Lessor Guaranty until any such Person is released from further liability upon a Permitted Sale as provided in <u>Section 17.5(b)</u>.

"Lessor Guaranty" means that certain Lessor Guaranty executed by CorEnergy Infrastructure Trust, Inc., a Maryland corporation (formerly known as Tortoise Capital Resources Corporation) and dated as of the Effective Date and any subsequent Lessor Guaranty executed after the Effective Date and in substantially the

form of Lessor Guaranty attached as Exhibit K unless and until the Lessor Guarantor under such Lease Guaranty is released to the extent provided in Section 17.5(b) upon a Permitted Sale.

"Lessor Indebtedness" means Indebtedness of the Lessor.

"Lessor Indemnified Parties" means Lessor, Lessor Guarantor and their respective Related Parties (solely in their capacity as such), and permitted successors and assigns.

"Lessor Interests" means (a) all or any portion of Lessor's right, title and interest in and to the Leased Property, and (b) all or any portion of Lessor's right, title and interest under this Lease.

"Lessor Lease Transaction" means a transaction pursuant to which Lessor enters into a lease or other agreement or arrangement with respect to the use and/ or occupancy by a third party of all or any portion of the Leased Property.

"Lessor Lender" means (a) any holder of (i) Lessor Indebtedness, or (ii) any Lien on any of the Lessor Interests or Lessor Equity Interests, and (b) any agent or trustee for such holders and counterparties.

"Lessor Loan" means the Indebtedness evidenced by the Lessor Loan Documents.

"Lessor Loan Document Criteria" means, with respect to Lessor Loan Documents, (a) such Lessor Loan Documents include each and every Required Loan Document Provision, and (b) such Lessor Loan Documents do not include any Prohibited Loan Document Provision. Any formal or informal waiver by Lessee of any failure of any Lessor Loan Document to comply with the Lessor Loan Document Criteria in any respect shall not in any event constitute a waiver of any other or further failure of any Lessor Loan Document to comply with the Lessor Loan Document Criteria thereafter.

"Lessor Loan Document Amendment" means an amendment, restatement, modification, supplement, waiver, refinancing, termination, supplement or replacement of a Lessor Loan Document, and "Amended" shall have a correlative meaning.

"Lessor Loan Documents" means the agreements and instruments evidencing and/or governing Indebtedness of the Lessor and the Liens securing such Indebtedness. Such term includes the Initial Lessor Loan Documents, all Lessor Loan Document Amendments thereto, all future Lessor Loan Documents, and all future Lessor Loan Document Amendments thereto.

"Lessor Parent" means (a) for so long as Pipeline is the Lessee, CorEnergy Infrastructure Trust, Inc., a Maryland corporation formerly known as Tortoise Capital Resources Corporation, and (b) for so long as any other Person is the Lessor, such Person's ultimate parent company.

"Lessor Party" means Lessor, any Equity Investor, and their respective officers, employees, agents, servants, or assignees.

"Level 1 Lessee Default' means (a) any Lessee Event of Default described in Section 23.1(a) (Base Rent Defaults), (b) any Lessee Event of Default described in Section 23.1(b) (Additional Rent Defaults), (c) any Lessee Event of Default described in Section 23.1(d) (Lease Assignment or Sublease), (d) any Lessee Event of Default described in Section 23.1(g) (Debtor Relief Laws), and (e) any Lessee Event of Default described in Section 23.1(h) (Cross Acceleration with Material Debt).

"Level 1 Lessor Default" means, for so long as an Ultra Entity Person is Lessee, (a) any Lessor Event of Default under<u>Section 23.3(b)</u> (Governing Documents, Loan & Indebtedness Representations), (b) any Lessor Event of Default under <u>Section 23.3(d)</u> (includes <u>Section 3.2(a)</u> (Prohibited Record Agreements),

Section 17.5(a)(i)-(iv) (Prohibited Lessor Transfers, Indebtedness, Liens and Leases), Section 27.1(a) and (d)-(g) (SPE, Lessor Owners & Governing Document Requirements), & Sections 27.2(a), (b), (c), or (f) (Loan Related Requirements & Maintenance of Existence), (c) any Lessor Event of Default arising out of a breach of Article XI (Regulatory Issues) or Section 26.10 (Confidentiality), (d) any Lessor Event of Default under Section 23.3(f) (Equity Investor Defaults), (e) any Lessor Event of Default under Section 23.3(g) (Failure to deliver sales documents), (f) any Lessor Event of Default under Section 23.3(h) (ROFR Breach), or (g) any Lessor Event of Default under Section 23.3(j) (Debtor Relief Laws). During any period that an Ultra Entity Person is not the Lessee, no Lessor Event of Default shall constitute a Level 1 Lessor Default.

"Level 1A Lessor Default" means, for so long as an Ultra Entity Person is Lessee, (a) any Lessor Event of Default under<u>Section 23.3(b)</u> (Governing Documents, Loan & Indebtedness Representations), (b) any Lessor Event of Default under <u>Section 23.3(d)</u> (picks up <u>Section 3.2(a)</u> (Prohibited Record Agreements), <u>Section 17.5(a)(i)-(iv)</u> (Prohibited Transfers, Indebtedness, Liens and Leases), <u>Section 27.1(a) and (d)-(g)</u> (SPE, Lessor Owners & Governing Document Requirements), <u>& Sections 27.2(a)</u>, (b), (c), or (f) (Loan Related Requirements & Maintenance of Existence), (c) any Lessor Event of Default under <u>Section 23.3(f)</u> (Requity Investor Defaults), (d) any Lessor Event of Default under <u>Section 23.3(g)</u> (Failure to deliver sales documents), (e) any Lessor Event of Default under <u>Section 23.3(h)</u> (ROFR Breach), or (f) any Lessor Event of Default under <u>Section 23.3(j)</u> (Debtor Relief Laws). During any period that an Ultra Entity Person is not the Lessee, no Lessor Event of Default shall constitute a Level 1A Lessor Default.

"Level 1A Lessor Default Payment Amount" has the meaning assigned to such term in Section 23.4(a)(vi).

"Level 1A Lessor Default Payments" has the meaning assigned to such term in Section 23.4(a)(vi).

"LGS Pipeline Riser" means the pipeline riser downstream of the liquids legs of the production separators located on each of the producing well pads described on the attached <u>Exhibit M</u>, as such pipeline risers may be modified after the Effective Date as permitted by this Lease.

"LGS Terminus Points" means each Condensate Terminus Point, each Flash Gas Terminus Point and each Produced Water Terminus Point.

"Lien" means any mortgage lien, deed of trust lien, deed to secure debt, vendor's lien, security interest, pledge, collateral assignment, mechanic's or materialman's lien, or other similar instruments and all restatements, modifications, amendments, consolidations, extensions, renewals or substitutions thereto or thereof.

"Liquids Gathering System" means generally, the system of pipelines and central gathering facilities highlighted on the map attached as<u>Exhibit N</u> as it exists as of the Effective Date and as it may be subsequently modified as permitted by this Lease, together with the related equipment that is capable of gathering, separating, collecting, and delivering for sale or transport, condensate and water, together with associated natural gas, produced from natural gas and oil wells that are, as of the Effective Date, operated by Ultra Resources and are located in the Pinedale Anticline Field in Sublette County, Wyoming, including, specifically, the Easement Rights and the Improvements, which Liquids Gathering System begins at the inlet flange to each LGS Pipeline Riser (an example of such an inlet flange is shown, for the Parties' convenience, on the photograph on the attached <u>Exhibit O</u>, extends through an interconnected system of pipelines to the production and separation equipment and storage tanks located at the Central Gathering Facilities, and terminates at each LGS Terminus Point.

"Losss" or "Losses" means mean any actual losses, costs, expenses (including court costs, reasonable fees and expenses of attorneys, technical experts and expert witnesses and the cost of investigation), liabilities, damages, demands, suits, claims, and sanctions of every kind and character (including civil fines)

arising from, related directly or indirectly or reasonably incident to, matters indemnified against, excluding however all of the following:

(a) <u>Consequential and Similar Damages</u> Any special, consequential, punitive or exemplary damages or loss of profits incurred by an indemnified party hereto; and

(b) <u>Diminution in Value</u>. Any diminution in value incurred or suffered (i) by a Lessor Indemnified Party except to the extent, and proportionately to the extent, such diminution in value results directly from the actions of, or the failure to act by, Lessee or another Lessee Party, and (ii) by a Lessor Indemnified Party except to the extent, and proportionately to the extent, such diminution in value results directly from the actions of, or the failure to act by, Lessor or another Lessor Party; and

(c) <u>Third Party Indemnifies</u>. Any such losses, costs, expenses, liabilities, damages, demands, suits, claims and sanctions incurred by an Indemnified Party and arising as a result of such Indemnified Party indemnifying a third party (i) with respect to a Lessor Indemnified Party, if (A) such third party is either (I) an Affiliate to such Lessor Indemnified Party, (II) another Lessor Indemnified Party or (III) a Lessor Party, or (B) such third party was caused by any Person described in <u>subsections (I)</u> – (III), above, to raise such claim (except to the extent such Person is required by Applicable Legal Requirements to cause such third party to raise such claim), (ii) with respect to a Lessee Indemnified Party, if (A) such third party is either (I) an Affiliate to such Lessee Indemnified Party, (II) another Lessee Party, or (B) such third party was caused by any Person described in <u>subsections (I)</u> – (III), above to raise such claim (except to the extent such Person is required by Applicable Legal Requirements to cause such third party or (III) a Lessee Party, or (B) such third party was caused by any Person described in <u>subsections (I)</u> – (III), above to raise such claim (except to the extent such Person is required by Applicable Legal Requirements to cause such third party to raise such claim), or (iii) to the extent such third party indemnified claim duplicates or exceeds the loss, cost or expense to the Indemnified Party; and

(d) Excluded Matters. Any such losses, costs, expenses, liabilities, damages, demands, suits, claims and sanctions which constitute an Excluded Matter; and

(e) <u>Section 2.1 Waivers</u>. Any such losses, costs, expenses, liabilities, damages, demands, suits, claims and sanctions with respect to claims and other matters waived in <u>Section 2.1</u>.

"Material Action" means, for purposes of <u>Part B</u> of <u>Schedule 27.1</u> to this Lease, (a) any Bankruptcy Action, (b) to take any action to merge, consolidate or combine the Lessor or Lessor's general partner or managing member (as the case may be) with any other Person, to dissolve or wind-up the Lessor or Lessor's general partner or managing member (as the case may be), or to approve any plan or agreement to engage in any of the foregoing actions, (c) to amend, alter or change any of the Lessor Governing Document Required Terms, (d) to take any action in contravention of the Lessor Governing Document Required Terms, or (e) to take action in furtherance of any of the foregoing.

"Material Adverse Event" means, for purposes of the definition of "Permitted Liens" herein, any event or circumstance resulting in a material adverse effect on (a) the ability of Lessee to perform the monetary and other material obligations under this Lease, or (b) the ownership, financial condition or operation of the Leased Property taken as a whole.

"Material Amendment" means (a) with respect to Section 25.3 hereof, (i) any change in the purchase price by more than three percent (3%); (ii) any change in the identity of the direct or indirect owners of the proposed purchaser; (iii) any change in the Lessor Interests included in the Transfer that is the subject of such ROFR rights of Lessee, or (iv) any other change in the terms of the Transfer that, taken together with any other changes after the relevant ROFR Asset Sale Notice, results in economic or business terms of the relevant Transfer that are materially more favorable to the proposed purchaser than the terms first disclosed to Lessee, and (b) with respect to Section 25.4 hereof, (i) any change in the identity of the direct or indirect owners of the

proposed lessee, licensee or sublessee; (iii) any change in the Applicable Leased Assets included in such Lessor Lease Transaction, or (iv) any other change in the terms of the Lessor Lease Transaction that, taken together with any other changes after the relevant ROFR Lease Notice, results in economic or business terms of the relevant Lessor Lease Transaction that are materially more favorable to the proposed lessee, licensee or sublessee than the terms first disclosed to Lessee.

"*Material Debt*" means, (a) for so long as (i) an Ultra Entity Person is the Lessee, or (ii) a successor Lessee who is an Investment Grade Person or whose Affiliate is a Lessee Guarantor that is an Investment Grade Person, in each case at the time such Person became the successor Lessee, Indebtedness the principal balance of which is in excess of \$100,000,000, and (b) with respect to any other Lessee, Indebtedness the principal balance of which is in excess of \$20,000,000.

"Material Lease Amendment" means any of the following:

(a) A waiver of any Level 1 Lessee Default or a waiver of any other Lessee Event of Default that remains uncured more than ninety (90) days after the occurrence of such Lessee Event of Default;

(b) A reduction in Base Rent, Minimum Rent, Total Rent or Variable Rent;

(c) A material increase in Lessor's obligations under this Lease;

(d) A direction to Lessee to pay Base Rent to any location other than the following: Account No. 359681346326, Account Name: Pinedale Corridor, LP, Reference "Transaction Description", Attn: Jennifer Duke, held at KeyBank National Association, 4910 Tiedeman Road 3rd Floor Brooklyn, OH 44144, ABA No. 041001039;

(e) A material reduction to Lessee's obligations to (i) maintain insurance as required by <u>Article XIV</u>, provided, however, for the avoidance of doubt, no Lessor Lender consent shall be required to maintain the self-insurance rights of Lessee as set forth in this Lease, (ii) pay Taxes and Impositions as required by <u>Article VI</u>, or (iii) maintain the Leased Property as required by <u>Section 7.1</u>;

(f) A material modification of Lessee's obligation to pay Casualty Proceeds or Taking Proceeds to Lessor or Lessor Lender as and to the extent required by <u>Articles</u> <u>XV</u> and <u>XVI</u>;

(g) A change to the definition of the terms Permitted Lease Assignment or Permitted Lessee Transferee which would expand Lessee's right to make Lease Assignments without Lessor's consent;

(h) A change to the definition of any of the following terms which results in a reduction to such amount or calculation: (i) Maximum Amount, (ii) Four Quarters Lessor EBITDA, or (iii) Monthly Debt Service;

(i) A reduction of the Lease Term except in connection with Lessee's election or deemed election not to extend for a Renewal Term or in connection with a termination of this Lease pursuant to and as provided in this Lease;

(j) Any material limitation of the terms Permitted Sale, Permitted Indebtedness, Permitted Lessor Liens or Permitted Equity Transfer;

(k) A material reduction of or limitation to the Lessee Environmental Liabilities or Lessee's indemnification obligations under Section 21.1;

(1) A termination of this Lease except pursuant to its terms, which permitted termination provisions include Lessee's election or deemed election not to extend for a Renewal Term or in connection

with a termination of the Lease pursuant to and as provided in the Lease (including Lessee Termination and Purchase Offer Rights and Lessee First Refusal Rights);

(m) A change to the definition of any of the terms Disqualified Person, Beneficial Owners, Ultra Designated Person, Competitor or Adverse Party which would have the effect of expanding the universe of "Persons" that could be included within any such term;

(n) A modification of Section 17.4 that expands the conditions under which the Lessee can be released from its obligations under the Lesse, under which the Lessee Guaranty, or under which Ultra Resources can be released from its obligations under the Resources Guaranty; or

(o) A modification of the Lessor Events of Default or the rights and remedies arising as a result of a Lessor Event of Default in a manner that is materially more beneficial to the Lessee.

"Maximum Amount" means, on any date, the greater of (a) Seventy Million Dollars (\$70,000,000), and (b) four times Four Quarters Lessor EBITDA on such date.

"Memo of Lease" means the Memorandum of Lease in the form attached to this Agreement as Exhibit P.

"Minimum Rent' means \$1,666,667 per month; provided that on each Adjustment Date, the Minimum Rent shall be adjusted by the applicable Index Adjustment, except that in no event will the Minimum Rent ever increase by more than two percent (2%) per year, and provided further, that such adjusted Minimum Rent shall constitute the Minimum Rent hereunder until the next Adjustment Date (it being understood and agreed that, notwithstanding anything herein to the contrary, for the avoidance of doubt, no decrease in the Index shall ever result in Minimum Rent of less than \$1,666,667 per month).

"Monthly Debt Service" means regularly scheduled monthly installments of principal and interest required to be paid by Lessor on a term loan or other term debt constituting Permitted Indebtedness secured only by Permitted Lessor Liens, it being understood and agreed that such monthly installments shall not include (a) default interest, penalties or late fees or charges, (b) gross-up or withholding payments, (c) payment of interest prior to accrual of such interest, (d) principal payments in excess of the greater of the amounts necessary to amortize the original principal amount of such Permitted Indebtedness either (i) in equal monthly payments of principal over a period of not less than ten (10) years, or (ii) on a straight line amortization method over a period not less than ten (10) years, (e) any "balloon" payment of principal, payment of principal at maturity, principal reductions required due to a failure to satisfy financial or other covenants or operating results, or other similar disproportionate payments of principal, (f) amounts due with respect to funding, hedging, swap or similar arrangements, or (g) other fees, charges or expenses.

"Monthly Operating Fee" means the fair market value for applicable services under the Operating Agreements as determined in accordance with Section 25.2.

"Natural Gas" means any mixture of natural gaseous hydrocarbons or of hydrocarbons and other gases, in a gaseous state, consisting primarily of methane.

"Nerd Farm Easement" means that certain Easement Agreement and Transfer of Improvements between Lessee, as grantor, and Lessor, as grantee, dated as of the Effective Date, and recorded in the Official Public Records.

"Non-Cash Consideration" has the meaning given in Section 25.3(g).

"Non-Cash Consideration Value" has the meaning given in Section 25.3(g).

"Non-Consent Matters" means a change of direct or indirect ownership of Lessee as a result of a merger, consolidation, reorganization, sale, distribution, contribution or other transfer of assets or equity interests, joint venture, or public offering of common stock or other equity interests, <u>provided that</u> the primary purpose of the transaction which results in a change in ownership of Lessee is not to circumvent or avoid Lessor's approval rights with respect to a Lease Assignment or Sublease.

"Non-Removable Addition" has the meaning given in Section 10.1(b).

"Notice Action" has the meaning given in Section 17.5(c)(ii).

"Offer" has the meaning given in Section 25.1.

"Official Public Records" means the applicable Official Public Records of Sublette County, Wyoming.

"Operating Agreements" means the Services Agreement and the Gathering Agreement, with such changes as mutually agreed by Lessor and Lessee in accordance with Section 25.2.

"Other Recorded Document" means any easement agreement, restrictive covenant, declaration, right-of-way or any other similar agreement or document affecting, benefiting or burdening all or any portion of the Liquids Gathering System, but excluding Liens.

"Parts" has the meaning given in Section 10.2.

"Permitted Capital Improvements and Additional Lines" means Capital Improvements and Additional Lines which (a) will not materially and adversely affect the structural integrity of the then existing Improvements, (b) will not materially and adversely affect the use or functionality of the Leased Property, and (c) will not adversely affect the value of the Leased Property in any material respect.

"Permitted Controller of Lease Rights" means any Person (a) that is not an individual, and (b) whose net worth, or the net worth of an Affiliate providing an appropriate guarantee, both immediately before and after giving effect to the applicable transfer, is at least \$50,000,000 determined in accordance with GAAP, provided that such Affiliate guarantees the obligations, if any, of such Person under the Governing Documents of Lessor, the Lessor Loan Documents, the SNDA and the Equity Investor Agreement to which such Person is subject, and provided further, that each Permitted Controller of Lease Rights that is a direct beneficial owner of Lessor Equity Interests shall be a party to the Equity Investor Agreement as an "Equity Investor" (as such term is defined in the Equity Investor Agreement).

"Permitted Equity Transfers" has the meaning given in Section 17.5(c).

"Permitted Indebtedness" means Lessor Indebtedness (a) consisting of a term loan or other term debt in an aggregate principal amount which does not at any time exceed the Maximum Amount, (b) with respect to which no Lessor Lender is a Disqualified Person, and (c) which is evidenced, governed and secured by Lessor Loan Documents which comply with the Lessor Loan Document Criteria and otherwise comply with Section 27.2.

"Permitted Lease Assignment" has the meaning given in Section 17.1(a).

"Permitted Lessee Contest" means any contest made in accordance with Section 26.12.

"Permitted Lessee Transferee" means, with respect to any Lessee, (a) Ultra Resources, (b) any other Ultra Entity Person, (c) any successor to an Ultra Entity Person as a result of a Non-Consent Matter, (d) any Affiliate of a Lessee, (e) so long as an Ultra Entity Person is the Lessee, any other Person (or an Affiliate of such Person whose obligations under this Lease are guaranteed by such acquiring Person) who purchases or otherwise accepts a transfer or assignment of all or a substantial portion of the interest of such Lessee and/ or its Affiliates in the oil and gas leases in connection with which the Leased Property is used, (f) so long as an Ultra Entity Person is the Lessee, any other Investment Grade Person approved by Lessor, which approval shall not be unreasonably withheld and which approval shall be based solely on the following: (i) a review of such Investment Grade Person's experience in the oil and gas exploration and production business, and (ii) the proposed use by such Investment Grade Person would not (A) constitute a breach of <u>Article IX</u>, or (B) in Lessor's reasonable judgment, impair the structural integrity, functionality or value of the Leased Property in any material respect, and (g) any other Person's business, (iii) such other Person's ability to recover costs or pass them on to its customers and other Persons, and (iv) the proposed use by such other Person's experience in the oil and gas exploration and production philosophy and reputation, (ii) such other Person's experience in the oil and gas exploration and production business, them on to its customers and other Persons, and (iv) the proposed use by such other Person's ability to recover costs or pass them on to its customers and other Persons, and (iv) the proposed use by such other Person's ability to recover costs or pass them on to its customers and other Persons, and (iv) the proposed use by such other Person's any proved use by such other Person's reasonable judgment, impair the structural integrity, functionality or value o

"Permitted Lessor Contest" has the meaning given in Section 27.2(g).

"Permitted Lessor Liens" means (a) Liens on the Leased Property and/ or the Lessor's right, title and interest under this Lease but only if (i) such Liens secure only Permitted Indebtedness and, if applicable, Permitted Secured Interest Rate Hedging Agreements, (ii) the Lessor Loan Documents evidencing and securing such Indebtedness comply with the Lessor Loan Document Criteria and otherwise comply with <u>Section 27.2</u>, (iii) such Liens do not result in a breach, default, violation or failure of performance under the Governing Documents of Lessor, and (iv) Lessor and each Lessor Lender with respect to such Liens have executed and delivered to Lessee the SNDA, and (b) Liens on any Lessor Equity Interest which is a general partnership interest but only if (i) such Liens secure only Permitted Indebtedness and, if applicable, Permitted Secured Interest Rate Hedging Agreements, secured by the Liens permitted in <u>subsection (a)</u>, above, (ii) the Lessor Loan Documents evidencing and securing such Indebtedness comply with <u>Section 27.2</u>, (iii) such Liens do not result in a breach, default, violation or failure of performance under the Governing Documents, secured by the Liens permitted in <u>subsection (a)</u>, above, (ii) the Lessor Loan Documents evidencing and securing such Indebtedness comply with <u>Section 27.2</u>, (iii) such Liens do not result in a breach, default, violation or failure of performance under the Governing Documents of Lessor, and (iv) Lessor and each Lessor Lender with respect to such Liens have executed and delivered to Lessee the SNDA. Notwithstanding the foregoing, in the event that after the Effective Date Pinedale GP, Inc., converts a portion of its general partnership interest in Pinedale Corridor, LP, the Lien on such general partnership interest which qualified as a Permitted Lessor Lien prior to such conversion may continue in effect with respect to such limited partnership interest for so long as the owner of such converted limited partnership interest is a wholly owned direct or indirect subsidi

"Permitted Liens" means, for purposes of Sections 3.1 and 3.2(a)(v) and the definition of the term "Material Adverse Event", (a) liens imposed by law for taxes, assessments, or similar charges, incurred in the ordinary course of business that are not yet due and payable, (b) liens of mechanics, materialmen, warehousemen, carriers, landlords or other like Liens, securing obligations incurred in the ordinary course of business that are not yet due and payable, (c) covenants, zoning restrictions, rights, easements, Liens, governmental environmental permitting and operation restrictions, the exercise by governmental bodies or third parties of eminent domain or condemnation rights, or any other restrictions on the use of real property, none of which individually or in the aggregate would result in a Material Adverse Event, and (d) the following, if the validity or amount thereof is being contested in good faith by appropriate and lawful proceedings and with respect to which adequate reserves are set aside on its books, and so long as they would

not, individually or in the aggregate, result in a Material Adverse Event: (1) claims or liens for taxes, assessments, or charges due and payable and subject to interest or penalty, (2) claims, liens upon, and defects of title to, real or personal property, including any attachment of personal or real property or other legal process prior to adjudication of a dispute on the merits, (3) claims or liens of mechanics, materialmen, warehousemen, carriers, or other like liens, and (4) adverse judgments on appeal, and (e) inchoate liens in respect of royalty owners; provided that the term "Permitted Liens" shall not include any Lien securing Indebtedness.

"Permitted Sale" has the meaning given in Section 17.5.

"Permitted Secured Interest Rate Hedging Agreements" means interest rate hedging agreements entered into by Lessor, with counterparties who are not Disqualified Persons, to mitigate interest rate risk in connection with Permitted Indebtedness.

"Permitted Sublease" has the meaning in Section 17.1(b).

"Permitted Ultra Petroleum Successor" means any successor to Ultra Petroleum so long as no change in control has occurred with respect to Ultra Petroleum. As used in this defined term, "change in control" means an event or series of events by which any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act, but excluding any employee benefit plan, and any Person acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan) becomes the Beneficial Owner, directly or indirectly, of 50% or more of the equity interests of Ultra Petroleum entitled to vote, on a fully diluted basis, for members of the board of directors or equivalent governing body of Ultra Petroleum.

"Person" means any individual, firm, corporation, partnership, limited liability company, incorporated or unincorporated association, joint venture, governmental authority or any other entity of any kind.

"Personal Property" means (a) the monitoring equipment located in or on the Liquids Gathering System, (b) the computer hardware located in the centralized monitoring building for each CGF constituting a portion of the Liquids Gathering System to which such monitoring equipment connects, (c) the wires and other connectors for such monitoring equipment between such computer hardware and such monitoring equipment, (d) all office furniture in each centralized monitoring building for each Central Gathering Facility, and (e) the Records, all as of the Effective Date and as such personal property may be modified or replaced as permitted or required by this Lease.

"Pinedale" means Pinedale Corridor, LP, a Delaware limited partnership, the initial Lessor under this Lease; however, the term Pinedale shall not include any successor or assignee of the interest of Pinedale hereunder.

"Prime Rate" means the prime rate of interest as reported in The Wall Street Journal or, if The Wall Street Journal is no longer published or no longer reports such prime rate, the prime rate of interest as reported in other authoritative publication or news retrieval service.

"Produced Water Terminus Point" means each outlet flange to the water pump facility or facilities, as the case may be, on each Central Gathering Facility that is directly upstream of each water meter and through which the produced water flows after it is separated from the condensate and the associated natural gas, as of the Effective Date and as such outlet flanges may be modified as permitted by this Lease.

"Prohibited Loan Document Provisions" means each of the following:

a. Any provision which (i) provides that the exercise by Lessee of any right or remedy of Lessee under this Lease would constitute, or (ii) otherwise results in Lessee's exercise of any right or remedy under this Lease constituting a default under a Lessor Loan Document;.

b. Any provision which provides that a Permitted Lease Assignment or a Permitted Sublease would constitute or otherwise result in a default under a Lessor Loan Document;

c. Any provision which imposes any condition or restriction on the ability of the Lessor to (i) give an approval or consent under this Lease with respect to a Permitted Lessee Transferee except with respect to a Person proposed under <u>subsection (g)</u> of the term "Permitted Lessee Transferee", or with respect to a Permitted Lease Assignment except with respect to a Lease Assignment proposed under <u>Section 17.1(a)(ii)</u> of the term "Permitted Lessee Assignment", (ii) amend, modify, restate or supplement this Lease except for a Material Lease Amendment, (iii) waive a breach, default or failure of performance under this Lease except with respect to a Level 1 Lessee Default or with respect to any other Lessee Event of Default that remains uncured ninety (90) days after the occurrence of such Lessee Event of Default, (iv) refrain from declaring a Lessee Event of Default that remains uncured ninety (90) days after the occurrence of Such Lessee Event of Default, (v) refrain from exercising rights or remedies in connection with a Lessee Event of Default except with respect to a Level 1 Lessee Default or with respect to any other Lessee Event of Default that remains uncured ninety (90) days after the occurrence of such Lessee Event of Default, (v) refrain from exercising rights or remedies in connection with a Lessee Event of Default except with respect to a Level 1 Lessee Default or with respect to any other Lessee Event of Default that remains uncured ninety (90) days after the occurrence of such Lessee Event of Default, or (v) refrain from exercising rights or remedies in connection with a Lessee Event of Default, or (vi) accept any Lessee Termination and Purchase Offer Rights offer, so long as the purchase price therefor is sufficient to repay such Permitted Indebtedness in full;

d. Any provision which imposes conditions or restrictions that would prevent Lessor from meeting its obligations to grant approvals and consents under this Lease; provided, however, that the foregoing shall not be deemed to prohibit provisions which (i) give a majority in interest of the Lessor Lenders approval rights with respect to (A) a Material Lease Amendment, or (B) a waiver of a Level 1 Lessee Default or with respect to any other Lessee Event of Default that remains uncured ninety (90) days after the occurrence of such Lessee Event of Default, or (ii) give Lessor Lenders approval rights with respect to (X) a Person proposed as a Permitted Lessee Transferee under subsection (g) of the definition of the term "Permitted Lessee Transferee" in this Lease, (Y) a Permitted Lease Assignment proposed under Section 17.1(a)(ii), or (Z) whether or not Lessor requires a Lessee Guaranty under subsection (ii) of the proviso at the end of Section 17.4, in any the case of subsections (X), (Y) or (Z), upon the vote of seventy five percent (75%) in interest of the Lessor Lenders;

e. Any provision which results in Lessee's use, maintenance or operation of the Leased Property as permitted or required by this Lease constituting or resulting in a default under a Lessor Loan Document;

f. Any provision imposing covenants, requirements or defaults with respect to Lessee (other than those covered by <u>subsection (e)</u>, above, with respect to Lessee's use, maintenance or operation of the Leased Property, for which <u>subsection (e)</u> shall govern and control) that (i) either (A) are inconsistent or in direct conflict with corresponding covenants or requirements imposed on Lessee under this Lease, or (B) impose additional covenants, requirements or defaults with respect to Lessee other than those set forth in this Lease, and (ii) the breach of, or failure to comply with, results in a default under any Lessor Loan Document;

g. Any representations, warranties, covenants or other provisions (i) which are expressly inconsistent by their terms with the SNDA, and (ii) the breach of, or failure to comply with, results in a default under any Lessor Loan Document;

h. Any provision which provides that the occurrence of an event or circumstance which, with the giving of notice or the passage of time, or both, would constitute a Lessee Event of Default or Lessor

Event of Default also constitutes a default under a Lessor Loan Document, in any case prior to the expiration of the applicable cure period with respect to such matter under this Lease;

i. Any provision which provides for a cross-default or cross-collateralization in each case with Indebtedness or other obligations of any Person other than Lessor or (subject to the other Prohibited Loan Document Provisions) Lessee, Lessee Guarantor or Ultra Resources;

j. Any default under the Lessor Loan Documents which is triggered by or based on (i) the creditworthiness, financial condition, business, operations or other status or condition of any Person other than Lessor, Lessee Guarantor, Ultra Resources or (subject to the other Prohibited Loan Document Provisions) Lessee, or (ii) on any representation, warranty or covenant made by, or with respect to, any Person other than Lessor, Lessee Guarantor, Ultra Resources or (subject to the other Prohibited Loan Document Provisions) Lessee.

As used in this definition, "default," means a default, event of default, requirement for prepayment or requirement that indebtedness be converted to a demand obligation.

Lessee agrees that notwithstanding the foregoing, the provisions of the Lessor Loan Documents as in effect on the date hereof pertaining to a Cash Dominion Event upon a CORR Default or a Lessee Debt Default (as such terms are therein defined), including the provisions pertaining to mandatory prepayment as a consequence thereof, and any provisions that are substantially the same that are contained in any amendment, restatement or refinancing of such Lessor Loan Documents or in any Lessor Loan Documents entered into in the future, shall not be deemed to be Prohibited Loan Document Provisions.

"Proprietary Information" means (a) the business concept, operating techniques, marketing methods, financial information, plans, site and system renderings, schedules, itemized costs, development plans and all related trade secrets or confidential or proprietary information treated as such by Lessee, whether by course of conduct, by letter or report or by use of any appropriate proprietary stamp of legend designating such information item to be confidential or proprietary, by communication to such effect made prior to or at the time any such Proprietary Information is disclosed to Lessor, or otherwise, (b) all financial statements and financial information that relates to the Liquids Gathering System delivered to Lessor by an Ultra Entity Person prior to the Effective Date, and (c) all Flow and Volume Information and similar information that relates to the Liquids Gathering System delivered to Lessor by an Ultra Entity Person prior to the Effective Date. Notwithstanding the foregoing, Proprietary Information shall not include information that is publicly available other than as a result of actions in violation of Section 26.10 hereof.

"Purchase Agreement" means the Purchase and Sale Agreement dated as of December 7, 2012, between Ultra Wyoming, Inc., as Seller, and Lessor, as Buyer, with respect to the transfer of the Leased Property to Lessor, as amended on or prior to the Effective Date.

"Real Property" means the Liquids Gathering System.

"*Record Agreements*" shall mean the agreements constituting (a) the Effective Date Recorded Documents (and including, without limitation, those evidencing the Easements), (b) any Other Recorded Document executed by Lessee after the Effective Date other than in violation of this Lease, and (c) any Other Recorded Document executed by Lessor other than in violation of this Lease. In no event shall any Lien constitute a "Record Agreement".

"Records" means the following information, to the extent acquired by Lessor (a) from Lessee under the Purchase Agreement or (b) in connection with construction of Permitted Capital Improvements and

Additional Lines paid for by Lessor and constructed by Lessor hereunder: all engineering drawings or plans and specifications of or covering the Liquids Gathering System or any component thereof, site assessments and environmental reports regarding or covering the Liquids Gathering System or any component thereof, manuals relating to the operation of the Leased Property, and "as-built" surveys of the pipelines and drawings of the Liquids Gathering System.

"Related Parties" of a Person means such Person's directors, officers, employees, agents, trustees, administrators, managers, advisors, accountants, attorneys and representatives.

"*Release*" means any leaking, spilling, pouring, pumping, emitting, injecting, escaping, leeching, dumping, discharging, depositing or disposing of any Hazardous Materials into the environment (including the air, soil, groundwater or surface water) in sufficient quantity or concentration such that notification to a Governmental Authority is required under Environmental Laws.

"Renewal Notice Date" has the meaning given in Section 25.1(b).

"Renewal Option" shall have the meaning given in Section 25.1(a).

"Renewal Term" has the meaning given in Section 25.1(a).

"Required Loan Document Provisions" means all of the following:

a. The maximum principal amount of the Lessor Loan, as of the date the Lessor Loan is made and as of the date of any renewal, extension, modification, or increase of the Lessor Loan, shall not exceed the Maximum Amount;

b. A provision granting Lessor the right to request a non-consenting Lessor Lender to assign its interests in the Lessor Loan to one or more replacement Lessor Lenders or increasing Lessor Lenders in the event that, in connection with any proposed amendment, waiver or consent with respect to this Lease requiring the consent of a majority or more than a majority of Lessor Lenders, Lessor Lenders holding more than 50% of the Lessor Loan have consented to such amendment, waiver or consent;

c. A provision prohibiting any participant in the Lessor Loan from voting with respect to approvals, consents, waivers, declarations of default, exercise of remedies or other matters or decisions to be made by Lessor Lenders under the Lessor Loan Documents;

d. A provision prohibiting any participant in the Lessor Loan who is a Disqualified Person from receiving any Proprietary Information or having any inspection rights with respect to the Leased Property;

e. A provision requiring Lessor's consent to any assignment by any Lessor Lender to Persons other than Lessor Lenders or their affiliates of any of its respective rights or interests in or obligations under the Lessor Loan Occuments so long as no event of default exists under the Lessor Loan Documents;

f. A provision prohibiting any Lessor Lender from assigning all or any portion of its rights or interests in, or obligations under, the Lessor Loan or the Lessor Loan Documents to a Disqualified Person;

g. A provision requiring that at least fifteen (15) Business Days prior to any assignment or grant of a participation by any Lessor Lender to Persons other than Lessor Lenders or their affiliates of any of its respective rights or interests in, or obligations under, the Lessor Loan or the Lessor Loan Documents, the assigning or granting Lender must deliver to Lessee or the agent for such Lessor Lenders, as applicable,

written notice of the proposed assignment or participation, together with sufficiently detailed information concerning the identity of the proposed assignee or participant to enable Lessee to determine whether the proposed assignee or participant is, or meets the criteria required in order for Lessee to designate the proposed assignee as, a Disqualified Person;

h. A provision permitting Lessor to prepay the Lessor Loan; and

i. A provision authorizing and requiring Lessor Lenders to execute the SNDA, and confirming that (i) each Lessor Lender shall be bound by the SNDA and its terms, and (ii) the rights and remedies of the Lessor Lenders under the Lessor Loan Documents shall be subject to the SNDA in all respects.

"Resources Guaranty" means the Limited Guaranty of Collection dated of even date herewith executed by Ultra Resources.

"Responsible Officer" means (a) with respect to any Ultra Entity Person, the President, the Senior Vice President of Operations, the Chief Financial Officer of Ultra Petroleum or an equivalent officer of Ultra Petroleum, (b) with respect to any other Lessee, any of the chief executive officer, chief operating officer, chief financial officer, principal accounting officer, treasurer or controller of Lessee Guarantor, if applicable, and (c) with respect to Lessor, any of the chief executive officer, chief operating officer, chief financial officer, treasurer or controller of Lessor or its Lessor Guarantor, if applicable.

"ROFR Asset Election Period" has the meaning given in Section 25.3(e).

"ROFR Asset Offer" has the meaning given in Section 25.3(d).

"ROFR Asset Sale Notice" has the meaning given in Section 25.3(c).

"ROFR Lease Election Period" has the meaning given in Section 25.4(e).

"ROFR Lease Notice" has the meaning given in Section 25.4(c).

"ROFR Lease Offer" has the meaning given in Section 25.4(d).

"ROFR Lease Period" means the period beginning at the end of the Lease Term and ending two (2) years later.

"ROFR Transfer Period" means (a) the period during the Lease Term, and (b) the period beginning at the end of the Lease Term and ending two (2) years thereafter.

"Separateness Covenants" means, collectively, each of the following covenants of a Person:

(a) to correct any known misunderstanding regarding the separate identity of such Person;

- (b) to maintain its accounts, books and records separate from any other Person and will file its own tax returns;
- (c) to maintain its books, records, resolutions and agreements as official records;
- (d) to not commingle its funds or assets with those of any other Person;
- (e) to hold its assets in its own name;



- (f) to conduct its business in its name only, and not use any trade name
- (g) to maintain its financial statements, accounting records and other entity documents separate from any other Person;
- (h) to pay its own liabilities, including the salaries of its own employees, out of its own funds and assets;
- (i) to observe all partnership, corporate or limited liability company formalities (as applicable);
- (j) to maintain an arm's-length relationship with its Affiliates;

(k) to not assume or guarantee or otherwise become obligated for the debts of any other Person or hold out its credit as being available to satisfy the obligations of any other Person;

(l) to not acquire obligations or securities issued by its partners, members or shareholders (as applicable), provided that for the avoidance of doubt, this clause does not restrict an exchange by Ross Avenue Investments, LLC, of its Company Equity Interests with the Lessor or CorEnergy Infrastructure Trust, Inc.;

- (m) to allocate fairly and reasonably shared expenses, including shared office space;
- (n) to use separate stationery, invoices and checks;
- (o) except in connection with a Lessor Loan which complies with the Lease, to not pledge its assets for the benefit of any other Person;
- (p) to hold itself out and identify itself as a separate and distinct entity under its own name and <u>not</u> as a division or part of any other Person;
- (q) to maintain its assets in such a manner that it will not be costly or difficult to segregate, ascertain or identify its individual assets from those of any other Person, except as may result from Permitted Capital Improvements and Additional Lines under this Lease;
 - (r) to not make loans to any Person;
 - (s) to not identify its partners, members or shareholders (as applicable), or any Affiliate of any of them, as a division or part of it;

(t) to not enter into or be a party to, any transaction with its partners, members or shareholders (as applicable) or any Affiliates except in the ordinary course of its business and on terms which are intrinsically fair and are no less favorable to it than would be obtained in a comparable arm's-length transaction with an unrelated third party;

(u) to not indemnify (or have any obligation to indemnify) its partners, officers, directors or members (as the case may be) (other than an Independent Director for such Person serving through an entity providing such services and which indemnity is required for such Independent Director to serve in such capacity), unless such an obligation is fully subordinated to the Lessor Loan and will not constitute a claim against it if its cash flow is insufficient to make payments due under the Lessor Loan; and

(v) to consider the interests of its creditors in connection with all corporate, partnership or limited liability company actions (as the case may be). *"Services Agreement*" means the Services Agreement in the form attached hereto as Exhibit F-1.

"Single Member Bankruptcy Remote LLC" means a limited liability company organized under the laws of the State of Delaware which at all times since its formation and at all times thereafter:

(a) was and will be organized solely for the purpose of acting as the general partner of the limited partnership or the managing member of the limited liability company (as the case may be) that owns the Leased Property;

(b) has not engaged and will not engage in any business unrelated to acting as the general partner of the limited partnership or the managing member of the limited liability company (as the case may be) that owns the Leased Property;

(c) has not had and will not have any assets other than those related to its member interest in the limited partnership or limited liability company (as the case may be) that owns the Leased Property;

(d) has not engaged, sought or consented to and will not engage in, seek or consent to any dissolution, winding up, liquidation, termination, consolidation, merger or sale of all or substantially all of its assets;

(e) has not, and without the consent of all of its directors and members (including the Independent Manager) will not, with respect to itself of any other entity in which it has a direct or indirect legal or beneficial ownership interest, take any Bankruptcy Action;

- (f) has observed, and will continue to observe, the Separateness Covenants;
- (g) has maintained and will maintain its accounts, books and records separate from any other person;
- (h) has and will have an operating agreement which provides that the business and affairs of Lessor shall be managed by or under the direction of:

(i) a board of one or more directors designated by the sole member of the Single Member Bankruptcy Remote LLC (the "Sole Member"), and at all times there shall be at least one (1) duly appointed Independent Director on the board of directors, and the board of directors will not take any action requiring the unanimous affirmative vote of one hundred percent (100%) of the members of its board of directors unless, at the time of such action there is at least one member of the board of directors who is an Independent Director, and all of the directors and the Independent Director shall have participated in such vote; or

(ii) the Sole Member, *provided that* at all times there shall be at least one Independent Manager designated by the Sole Member and the operating agreement provides that the Sole Member shall not take any Bankruptcy Action without the unanimous affirmative vote of all of its members (including the Independent Manager);

(i) has and will have an operating agreement which provides that, as long as the Lease remains in effect:

(i) upon the occurrence of any event that causes the Sole Member to cease to be a member of Lessor (other than (x) upon an assignment by the Sole Member of all of its

limited liability company interest in Lessor and the admission of the transferee, if permitted pursuant to the organizational documents of Lessor, the Lessor Loan Documents and this Lease, or (y) the resignation of Sole Member and the admission of an additional member of dissolution; Lessor, if permitted pursuant to the organizational documents of Lessor, the Lessor Loan Documents and this Lease), the Independent Manager shall, without any action of any Person and simultaneously with Sole Member ceasing to be a member of Lessor, automatically be admitted as the sole member of Lessor (the "*Special Member*") and shall preserve and continue the existence of Lessor without

(ii) the Special Member may not resign or transfer its rights as Special Member unless (x) a successor Special Member has been admitted to Lessor as a Special Member, and (y) such successor Special Member has also accepted its appointment as an Independent Manager; and

(iii) except as expressly permitted pursuant to the terms of this Agreement, Sole Member may not resign and no additional member shall be admitted to Lessor;

(j) has and will have an operating agreement which provides that, at all times during the Lease Term:

(i) Lessor shall be dissolved, and its affairs shall be would up only upon the first to occur of the following: (x) the termination of the legal existence of the last remaining member of Lessor or the occurrence of any other event which terminates the continued membership of the last remaining member of Lessor in Lessor unless the business of Lessor is continued in a manner permitted by its operating agreement or the Delaware Limited Liability Company Act (the "*Act*") or (y) the entry of a decree of judicial dissolution under Section 18-802 of the Act;

(ii) upon the occurrence of any event that causes the last remaining member of Lessor to cease to be a member of Lessor or that causes Sole Member to cease to be a member of Lessor (other than (x) upon an assignment by Sole Member of all of its limited liability company interest in Lessor and the admission of the transferee, if permitted pursuant to the organizational documents of Lessor, the Lessor Loan Documents and this Lease, or (y) the resignation of Sole Member and the admission of an additional member of Lessor, if permitted pursuant to the organizational documents of such member shall be authorized to, and shall, within ninety (90) days after the occurrence of the event that terminated the continued membership of such member in Lessor, agree in writing to continue the existence of Lessor and to the event that terminated the continued membership of such member in Lessor, as a substitute member of Lessor, effective as of the occurrence of the event that terminated the continued member in Lessor;

(iii) the bankruptcy of Sole Member or a Special Member shall not cause such Sole Member or Special Member, respectively, to cease to be a member of Lessor and upon the occurrence of such an event, the business of Lessor shall continue without dissolution;

(iv) in the event of dissolution of Lessor, Lessor shall conduct only such activities as are necessary to wind up its affairs (including the sale of the assets of Lessor in an orderly manner), and the assets of Lessor shall be applied in the manner, and in the order of priority, set forth in Section 18-804 of the Act; and

(v) to the fullest extent permitted by law, each of Sole Member and Special Member shall irrevocably waive any right or power that each might have to cause Lessor or any of its assets to be partitioned, to cause the appointment of a receiver for all or any portion of the assets of Lessor, to compel any sale of all or any portion of the assets of Lessor pursuant to any applicable law or to file a complaint or to institute any proceeding at law or in equity to cause the dissolution, liquidation, winding up or termination of Lessor.

"SNDA" means (a) that certain Subordination, Non-Disturbance and Attornment Agreement entered into as of the Effective Date among KeyBank National Association, Lessor and Lessee, and consented to by the Lessor Guarantor, the Lessee Guarantor, and Ultra Resources, and as the same may be subsequently amended, supplemented or restated, and (b) each subsequent Subordination, Non-Disturbance and Attornment Agreement to be entered into after the Effective Date between each Lessor Lender, Lessor and Lessee, and consented to by the Lessor Guarantor, the Lessee Guarantor, and, if the Resources Guaranty is still in effect, Ultra Resources, in substantially the form of <u>Exhibit E</u> hereto (but with the first three sentences of <u>Section 3.4</u> thereof updated to include correct factual information regarding any such Equity Interest Lien held by such Lessor Lender which is not prohibited by this Lease and with appropriate changes to <u>Sections 8.1</u> and <u>8.2</u> of the SNDA to assure that no such Lessor Lenders are Disqualified Persons and that such Lessor Loan Documents comply with the Lessor Loan Document Criteria), and in each case as the same may be subsequently amended, supplemented or restated.

"Special Purpose Bankruptcy Remote Entity" has the meaning given in Part A of Schedule 27.1.

"Sublease" means a Transfer by Lessee of a portion or less than all of Lessee's rights under this Lease or in the Leased Property hereunder.

"Subsidiaries" means, with respect to any Person (the "parent") at any date, any corporation, limited liability company, partnership, association or other entity the accounts of which would be consolidated with those of the parent in the parent's consolidated financial statements if such financial statements were prepared in accordance with GAAP as of such date, as well as any other corporation, limited liability company, partnership, association or other entity (a) of which securities or other ownership interests representing more than fifty percent (50%) of the equity or more than fifty percent (50%) of the ordinary voting power or, in the case of a partnership, more than fifty percent (50%) of the general partnership interests are, as of such date, owned, controlled or held, or (b) that is, as of such date, otherwise Controlled, by the parent or one or more subsidiaries of the parent or by the parent and one or more subsidiaries of the parent.

"Taking" means acquisition of all or any portion of the Leased Property for any public or quasi-public use through taking by condemnation, eminent domain or any like proceeding, or purchase in lieu thereof.

"Taking Proceeds" means damages and proceeds accruing and paid by or on behalf of the applicable Governmental Authority on account of any Taking of the Leased Property, save and except (a) the reasonable third party costs and expenses (including reasonable attorneys' fees and expenses) incurred by Lessee in connection with the collection of same, (b) the amount incurred by Lessee prior to the date of payment of such Taking Proceeds to repair or restore the Leased Property, and (c) any Lessee Condemnation Proceeds.

"Taking Response Notice" has the meaning assigned to such term in Section 16.2(c).

"Taking Termination Date" has the meaning assigned to such term in Section 16.2(d).

"Tax Challenge" has the meaning given in Section 6.6.

"Taxes" means (a) all property taxes and assessments actually imposed by a Governmental Authority on the ownership or operation of the Leased Property during and accruing with respect to the Lease Term, (b) all general and special assessments, levies, permits, inspection and licenses fees levied by a Governmental Authority on or with respect to the ownership or operation of the Leased Property during and accruing with respect to the Lease Term, (c) all other charges and/ or taxes levied by a Governmental Authority with respect to Lessee or the ownership or operation of the Leased Property, whether of a like or different nature from those described in <u>subsections (a)</u> and (b), preceding, even if unforeseen or extraordinary, during and accruing with respect to the Lease Term, and (d) the costs of any contest or appeal pursued by Lessee with respect to any of the taxes, levies and assessments described in <u>subsections (a)</u>, (b) or (c), preceding, but excluding in all cases with respect to <u>subsections (a)</u>, (b) and (c), all Excluded Matters.

"Term" means the Initial Term, as it may be extended by Lessee's exercise of the extension and renewal rights set forth in Section 25.1, or as it may be sooner terminated as provided in this Lease.

"Third Party Asset Offer" means a bona fide outstanding offer or binding contract (that is subject to Section 25.3 of this Lease) that Lessor receives with respect to Lessor's proposed Transfer of all of Lessor's interest in the Leased Property and the Lease from a Person that is not an Affiliate of Lessor which Lessor intends to accept and which complies with the following requirements:

(a) the offer (or contract) must be in writing and must be executed by the proposed purchaser (or, if a contract, executed by both the proposed purchaser and Lessor);

(b) the offer (or contract) must be by (or entered into by) the proposed buyer identified in the offer (or contract), and not by an agent acting on behalf of an undisclosed principal;

(c) the offer (or contract) must identify the direct and indirect owners of the proposed buyer and must specifically provide that the offer is not assignable by the proposed buyer except to an Affiliate thereof; and

(d) the offer (or contract) must include an agreement by the proposed buyer to provide evidence, as reasonably requested, as to the business, character, reputation and financial capacity of the proposed buyer and the direct and indirect owners associated therewith (and each Person that will have Controlling Lease Rights) to carry out the terms of the offer (or contract) as well as other information and documentation as may be required to evidence that the proposed Transfer will be a Permitted Sale.

"Third Party Lease Offer" means a bona fide outstanding offer or binding contract for a Lessor Lease Transaction that is subject to Section 25.4 of this Lease, which Lessor receives from a Person that is not an Affiliate of Lessor, which Lessor intends to accept and which complies with the following requirements:

(a) the offer (or contract) must be in writing and must be executed by the proposed lessee (or, if a contract, executed by both the proposed lessee and Lessor);

(b) the offer (or contract) must be by (or entered into by) the proposed lessee identified in the offer (or contract), and not by an agent acting on behalf of the proposed lessee

(d) the offer (or contract) must identify the direct and indirect owners of the proposed lessee and must specifically provide that the offer is not assignable by the proposed lessee except to an Affiliate thereof; and

(e) the offer (or contract) must include an agreement by the proposed lessee to provide evidence, as reasonably requested, as to the business, character, reputation and financial capacity of the proposed lessee and the direct and indirect owners of such proposed lessee to carry out the terms of the offer (or contract).

"Total Rent" means the lesser of (a) the sum [redacted] and (b) [redacted] per month.

"Transfer" means with respect to any property or rights, the sale, conveyance, transfer, exchange, gift, contribution or assignment, in whole or in part, of any right, title or interest in and to such property or rights, whether voluntary or Involuntary. As to an Equity Investor Interest or a Lessor Equity Interest, Transfer shall also include (i) a change in the Person or Persons who have beneficial ownership of a Lessor Equity Interest and (ii) a transfer to a Person or Persons of Controlling Lease Rights. The granting, attachment or creation of any Lien on all or any portion of such property or rights shall not constitute a voluntary or Involuntary Transfer of such property or rights. Notwithstanding the foregoing, the term "Transfer" shall not include (a) any sale, assignment or other transfer of the Lessor Equity Interest held initially by Ross Avenue Investments, LLC to any Affiliate of Prudential Financial, Inc., or (b) any exchange by Ross Avenue Investments, LLC of its Lessor Equity Interest with Lessor Equity Interest Owner (and any Affiliate guarantor(s) required by application of the "Permitted Controller of Lease Rights" definition to such Persons) have executed and delivered to Lesse an Addendum or Supplement to the Equity Investor Agreement (in form and substance reasonably acceptable to Lessor and Lesse) which has the effect of making such Lessor Equity Interest Owner (s) parties to, and subject in all respects to, the Equity Investor Agreement.

"Ultra Designated Party" means, so long as an Ultra Entity Person is the Lessee, (a) each Person from time-to-time identified in writing to Lessor by Lessee as a Person with a management or financial philosophy, management style or reputation which is unacceptable to Lessee, and (b) any Affiliates of each Person described in subsection (a).

"Ultra Entity Person" means (a) Ultra LGS and any Affiliate of or successor to Ultra LGS so long as Ultra LGS and/ or such Affiliate or successor is Controlled, directly or indirectly, by a Permitted Ultra Petroleum Successor, (b) Ultra Wyoming and any Affiliate of or successor to Ultra Wyoming so long as Ultra Wyoming and/ or such Affiliate or successor is Controlled, directly or indirectly, by a Permitted Ultra Petroleum Successor, (c) Ultra Resources and any Affiliate of or successor to Ultra Resources so long as Ultra Resources so long as Ultra Resources and/ or such Affiliate or successor is Controlled, directly or indirectly, by a Permitted Ultra Petroleum Successor, (d) Ultra Petroleum or any successor to Ultra Petroleum so long as such successor is a Permitted Ultra Petroleum Successor, and (d) any Affiliate of Ultra Petroleum so long as such Affiliate is Controlled, directly or indirectly, by a Permitted Ultra Petroleum Successor.

"Ultra LGS" means Ultra Wyoming LGS, LLC, a Delaware limited liability company, the initial Lessee under this Lease; however, the term Ultra LGS shall not include any successor or assignee of the interest of Ultra LGS hereunder.

"Ultra Petroleum" means Ultra Petroleum Corp. a Yukon Territory of Canada corporation, the initial Lessee Guarantor; however, the term "Ultra Petroleum" shall not include any successor Lessee Guarantor under this Lease.

"Ultra Resources" means Ultra Resources, Inc., a Wyoming corporation.

"Ultra Wyoming" means Ultra Wyoming, Inc., a Wyoming corporation.

"Underlying Lien Claims" means, with respect to the Nerd Farm Easement, (a) Mortgage recorded April 30, 1997, recorded in Book 68, Page 75 of the Mortgage Records of Sublette County, Wyoming, from Nerd Enterprises, Inc., to George P. Westman, and (b) Judgment dated December 8, 2010, recorded January 6, 2011, in Book 96, Page 421 of the Miscellaneous Records of Sublette County, Wyoming, against Questar Exploration and Production Company et al and in favor of Doyle Hartman et al.

"Variable Rent" means, for any calendar month during the Term from and after the first Adjustment Date, the [redacted] of (a[redacted] (b) the [redacted]. Variable Rent shall be payable on the first day of the third month following the month for which Excess Product Volume Factor is calculated. By way of example, if the first Adjustment Date occurs on May 1, the Variable Rent due on that date will be based on Excess Product Volume Factor for the preceding month of February.

1.2 Certain Interpretive Provisions.

As used in this Lease: (a) the word "or" is not exclusive and the word "including" is not limiting, (b) references to a law include any rule or regulation issued under the law and any amendment to the law, rule or regulation, (c) whenever the words "include," "includes," or "including" appear, they shall be deemed to be followed by the words "without limitation," (d) personal pronouns shall be deemed to include the other genders and the singular shall include the plural and vice versa, and (e) the words "herein," "hereof" and "hereunder" and other words of similar import refer to his Lease as a whole and not to any particular Article, Section or other subdivision. Wherever a period of time is stated in this Lease as commencing or ending on specified dates, such period of time shall be deemed (i) inclusive of such stated commencement and ending dates, and (ii) to commence at 12:00 A.M. Central Time on such stated commencement date and to end at 11:59 P.M. Central Time on such stated ending date. Unless the context otherwise requires, (A) any definition or reference to any agreement, instrument or other document shall be construed as referring to such agreement, instrument or other document shall be construed as referring to such agreement, instrument or other document as from time-to-time amended, supplemented or otherwise modified (subject to any restrictions on such amendments, supplements or modifications set forth herein), (B) subject to the restrictions on Transfers, Liens and assignments in this Lease, any reference herein to any Person shall be construed to include such Person's successor and assignments in this Lease, any reference herein to any person shall be construed to include such Person's successor and assignments on regulation provisions consolidating, amending, replacing or interpreting such law and reference to any law or regulation shall, unless otherwise specified, refer to such law or regulation as amended, modified or supplemented from time-to-time. Section headings herein are included for

ARTICLE II. LEASE CHARACTERIZATION

2.1 Lease Characterization.

(a) *True Lease*. Lessor and Lessee intend that this Lease is a "true lease" and not a financing lease, capital lease, mortgage, equitable mortgage, deed of trust, trust agreement, security agreement or other financing or trust arrangement, and the economic realities of this Lease are those of a true lease. Lessor and Lessee intend that the business relationship created by this Lease and any related documents is solely that of a long-term commercial lease between Lessor and Lessee.

(b) *Waivers and Stipulations.* Each of the parties (i) waives any claim or defense based upon the characterization of this Lease as anything other than a "true lease" or an "operating Lease" or that asserts that this Lease is anything other than a "true lease" or an "operating lease," (ii) waives any claim or defense based upon the characterization of all or any portion of the Leased Property as real or personal property, (iii) stipulates and agrees not to challenge the validity, enforceability or characterization of the lease of the Leased Property as a "true lease" or "operating lease," (iv) stipulates and agrees that nothing contained in

this Lease creates or is intended to create a joint venture, partnership (either de jure or de facto), equitable mortgage, trust, financing device or arrangement, security interest or the like, and (v) shall support, at such party's sole cost and expense, the intent of the parties that the lease of the Leased Property pursuant to this Lease is a true lease and does not create a joint venture, partnership (either de jure or de facto), equitable mortgage, trust, financing device or arrangement, security interest or the like, if, and to the extent that, any challenge occurs.

(c) *Material Inducement*. The expressions of intent, the waivers, the representations and warranties, the covenants, the agreements and the stipulations set forth in this Section are a material inducement to each of Lessor and Lessee in entering into this Lease.

(d) *Lease Continuation.* Except as limited in and otherwise provided in this Lease (and including, without limitation, Lessee's rights under <u>Articles XV, XVI, XXII, XXIII, and XXV</u>), this Lease shall not terminate and Lessee shall not have any right to terminate this Lease during the Term. Except as limited in and otherwise provided in this Lease (and including, without limitation, Lessee's rights under <u>Articles XV, XVI, XXII, XXIII, and XXV</u>): (1) this is a net lease and Lessee shall not be entitled to any setoff, counterclaim, recoupment, abatement, suspension, deferment, diminution, deduction, reduction or defense of or to Base Rent or Additional Rent, and (2) the obligations of Lessee under this Lease shall not be affected by any circumstance or event, or for any reason, including but not limited to the following: (i) any damage to or destruction of any of the Leased Property by any cause whatsoever, (ii) any Taking, (iii) the prohibition, limitation or restriction of, or interference with, Lessee's use of any of the Leased Property, (iv) any eviction by paramount title or otherwise, (v) Lessee's acquisition of ownership of any of the Leased Property other than pursuant to an express provision of this Lease, (vi) any default on the part of Lessor under this Lease or under any other agreement, (vii) any latent or other defect in, or any theft or loss of, any of the Leased Property, (viii) the breach of any warranty of any seller or manufacturer of any of the Improvements or Personal Property or Lessee Property, or (ix) any other cause, whether similar or dissimilar to the foregoing, any present or future Applicable Legal Requirement to the contrary notwithstanding. It is the intention of the parties hereto that except as limited in and otherwise equal thereto), and (C) the obligations of Lessee under this Lease shall continue unaffected, unless this Lease shall have been terminated pursuant to an express provision of this Lease.

ARTICLE III.

LEASED PROPERTY

3.1 Lease of Leased Property. Lessor leases to Lessee and Lessee leases from Lessor the Leased Property, subject to the Record Agreements and the Permitted Liens, for the Term, upon the terms and conditions of this Lease. Lessee shall have exclusive use and possession of the Leased Property subject to the Underlying Lien Claims, the Record Agreements and the Permitted Liens. The Leased Property is leased to Lessee subject to the Underlying Lien Claims, the Record Agreements, the Permitted Liens and all Applicable Legal Requirements now or hereafter in effect. LESSOR LEASES AND WILL LEASE AND LESSEE TAKES AND WILL TAKE THE LEASED PROPERTY "AS IS", AND LESSEE ACKNOWLEDGES THAT LESSOR (WHETHER ACTING AS LESSOR HEREUNDER OR IN ANY OTHER CAPACITY) HAS NOT MADE AND WILL NOT MAKE, NOR SHALL LESSOR BE DEEMED TO HAVE MADE, ANY WARRANTY OR REPRESENTATION, EXPRESS OR IMPLIED, WITH RESPECT TO ANY OF THE LEASED PROPERTY, INCLUDING ANY WARRANTY OR REPRESENTATION AS TO ITS FITNESS FOR USE OR DESIGN OR CONDITION FOR ANY PARTICULAR USE OR PURPOSE, AS TO THE QUALITY OF THE MATERIAL OR WORKMANSHIP THEREIN, LATENT OR PATENT, AS TO LESSOR'S TITLE THERETO, OR AS TO VALUE, COMPLIANCE WITH SPECIFICATIONS, LOCATION, USE, CONDITION, MERCHANTABILITY,

QUALITY, DESCRIPTION, DURABILITY OR OPERATION, IT BEING AGREED THAT ALL RISKS INCIDENT THERETO ARE TO BE BORNE BY LESSEE. In the event of any defect or deficiency in any of the Leased Property of any nature, whether patent or latent, except as otherwise expressly provided herein, Lessor shall not have any responsibility or liability with respect thereto or for any incidental or consequential damages (including strict liability in tort). The provisions of this <u>Section 3.1</u> have been negotiated, and the foregoing provisions are intended to be a complete exclusion and negation of any warranties by Lessor, express or implied, with respect to the physical condition of any of the Leased Property, and arising pursuant to the Uniform Commercial Code or any other Applicable Legal Requirements now or hereafter in effect or otherwise.

3.2 Additional Encumbrances.

(a) Execution of Additional Record Agreements.

(i) <u>Execution by Lessor</u>. Lessor shall not during the Lease Term enter into, create, place of record, amend, assign, restate, modify, or terminate any Record Agreement (including any document or instrument creating or governing any of the Easements) or any Other Recorded Document except (A) "Closing Deliveries" pursuant to the Purchase Agreement, (B) as requested by Lessee in writing, or (C) as consented to by Lessee in writing prior to the date of Lessor's execution thereof.

(ii) <u>Execution by Lessee</u>. Lessee shall not during the Lease Term enter into, create, place of record, amend, assign, restate, modify, or terminate any Record Agreement (including any document or instrument creating or governing any of the Easements) or any Other Recorded Document except (i) "Closing Deliveries" pursuant to the Purchase Agreement, or (ii) as Lessee from time-to-time in good faith deems necessary or appropriate for (A) use and operation of the Liquids Gathering System and Other Lessee Activities by Lessee and its Affiliates, (B) compliance with Applicable Legal Requirements and contractual arrangements with third parties regarding operation of the Liquids Gathering System, (C) compliance by Lessee with its obligations under this Lease, or (D) compliance with the requirements of this Lease regarding Tax Challenges and Permitted Lessee Contests.

(iii) Notice Prior to Execution; Special Provisions Regarding BLM Easement Extensions Lessor and Lessee shall each give the other Party not less than ten (10) Business Days' notice prior to executing any document creating, amending, assigning, restating, modifying or terminating any Record Agreement or Other Recorded Document, and in the event Lessor and Lessee disagree on the need for, or benefit or adverse effects of, any such document, Lessor and Lessee shall work together in good faith to resolve their differences, but in any event in a manner consistent with the rights and obligations of Lessor and Lessee under this Lease. Notwithstanding anything to the contrary in this Lease, Lessor and Lessee shall cooperate in good faith to execute such documents as may be required to extend the term of the BLM Easements through the end of the Lease Term, although the foregoing shall in no event be deemed to require Lessor or Lessee to pay any fee or charge BLM may impose for any such extended BLM Easement term which occurs during the Lease Term and which portion of the extended BLM Easement term occurs after the end of the Lease Term and which portion of the extended BLM Easement term occurs after the end of the Lease Term and which portion of the extended BLM Easement term occurs after the end of the Lease Term and which portion of the extended BLM Easement term occurs after the end of the Lease Term and which portion of the extended BLM Easement term occurs after the end of the Lease Term and which portion of the extended BLM Easement term occurs after the end of the Lease Term and which portion of the extended BLM Easement term occurs after the end of the Lease Term and which portion of the extended BLM Easement term occurs after the end of the Lease Term and which portion of the extended BLM Easement term occurs after the end of the Lease Term and which portion of the extended BLM Easement term occurs after the end of the Lease Term and which portion of the extended BLM Easement term occurs after the end of the Le

(iv) <u>Lessor Delegation of Enforcement Authority</u>. Lessor authorizes Lessee to enforce all Record Agreements (including any document or instrument creating or governing any of the Easements) and any Other Recorded Documents on Lessor's or Lessee's behalf, and Lessor shall

cooperate and furnish any pertinent information needed toward Lessee's enforcement of same, at no cost or expense to Lessor other than any de minimus cost or expense.

(v) <u>Lessee Compliance</u>. During the Lease Term, Lessee shall be obligated to comply in all material respects with the Record Agreements and the Permitted Liens, provided, however, it being understood and agreed that Lessee shall not be responsible for (A) any obligations, charges or liabilities arising out of the Underlying Lien Claims, the Permitted Lessor Liens, or any other matters of record that were voluntarily created or imposed by, through or under Lessor or an Affiliate of Lessor after the Effective Date in violation of this Lease, or (B) any fees or charges to extend the term of any BLM Easement, which is governed by the last sentence of <u>subsection (iii)</u>, above.

(b) Creation of Liens by Lessee on Lessor's Interest in the Leased Property. Subject to the provisions of Sections 6.6 and 12.1, without the prior written consent of Lessor (which consent shall not be unreasonably withheld), Lessee shall not, directly or indirectly, create or permit to be created or to remain, and shall promptly discharge, any Lien on Lessor's interest in (i) the Leased Property, (ii) this Lease, or (iii) the Base Rent or any Additional Rent, in the case of <u>subsections (i)</u>, (ii) or (iii), preceding, created by or resulting from any act or omission of Lessee or those claiming by, through or under Lessee (except Lessor and its Affiliates). Notice is hereby given that Lessor shall not be liable for any labor, services or materials furnished or to be furnished to Lessee, or to anyone holding any of the Leased Property through or under Lessee, and that no mechanic's or other liens for any such labor, services or materials shall attach to or affect the interest of Lessor or any of the Lessor Property. Notwithstanding the foregoing, Lessee has no liability for, or any obligation to cause to be paid or discharged, any Lien created by, through or under Lessor Party.

3.3 Lessee Other Activities. Lessor and Lessee acknowledge and confirm that (a) Lessee owns oil and gas wells and other rights and properties in and around Leased Property separate and apart from the Liquids Gathering System, the Personal Property and the Lessee Property; (b) Ultra Resources operates such oil and gas wells and other rights and properties and may itself own rights and property in and around the Leased Property separate and apart from the Liquids Gathering System and the Personal Property; (c) after the Effective Date, Lessee intends to continue to own such oil and gas wells and other rights and properties for the foreseeable future; and (d) after the Effective Date, Ultra Resources intends to continue to operate such oil and gas wells and other rights and properties and to own its other rights and property in and around the Leased Property for the foreseeable future (subsections (a)-(d) being collectively referred to as the "Lessee Other Activities").

ARTICLE IV. REPRESENTATIONS AND WARRANTIES

4.1 **Representations and Warranties of Lessor**. Lessor represents and warrants to Lessee as of the Effective Date as follows, it being understood and agreed that the following representations and warranties shall be deemed to be remade by a successor Lessor as of the effective date of any Permitted Sale (and with applicable revisions to <u>subsection (a)</u> with respect to such successor Lessor's organization and owners):

(a) *Existence and Ownership.* Lessor is duly organized, validly existing and in good standing under the laws of the state of Delaware, is qualified to do business and in good standing in the State of Wyoming (to the extent Lessor is required to be so by applicable Legal Requirements) and has full power, authority and legal right to execute and deliver and to perform and observe the provisions of this Lease to be observed and/or performed by Lessor. The sole partners of Lessor are (i) Pinedale GP, Inc., a Delaware corporation, the sole general partner of Lessor and a wholly owned subsidiary of CorEnergy Infrastructure

Trust, Inc., a Maryland corporation (formerly known as Tortoise Capital Resources Corporation), and (ii) Ross Avenue Investments, LLC, a Delaware limited liability company, the sole limited partner of Lessor and an indirect wholly owned subsidiary of Prudential Financial, Inc. Pinedale GP, Inc., holds an 82% general partnership interest in Lessor, and Ross Avenue Investments, LLC, owns an 18% limited partnership interest.

(b) *Binding Obligation.* This Lease has been duly authorized, executed and delivered by Lessor, and constitutes the valid and binding obligations of Lessor enforceable against Lessor in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and other laws affecting creditors' rights generally and subject to general principles of equity, whether considered in a proceeding in equity or at law.

(c) **Solvency**. Lessor is solvent, has timely and accurately filed all tax returns and extensions required to be filed by Lessor, and is not in default in the payment of any material taxes levied or assessed against Lessor or any of its material assets, and is not subject to any judgment, order, decree, rule or regulation of any Governmental Authority having jurisdiction over the Leased Property or Lessor which would, in the aggregate, otherwise materially and adversely affect Lessor's condition, financial or otherwise, or Lessor's prospects or the Leased Property.

(d) SPE. Lessor is and has at all times since its formation been a Special Purpose Bankruptcy Remote Entity.

(e) *No Violations.* The performance by Lessor of its obligations hereunder will not violate any provision of the Governing Documents of Lessor and, to Lessor's Actual Knowledge, will not (i) result in a material breach of any agreement to which Lessor is a party, (ii) violate in any material respect any provision of any judgment or order binding on Lessor, or (iii) constitute a material violation by Lessor of any law or governmental regulation applicable to Lessor.

(f) **Delivery of Governing Documents and Information.** Lessor has delivered, or caused Lessor Parent to deliver, to Lessee true, correct and complete copies of (i) its Governing Documents, and (ii) the information required by Section 2.1 of the Equity Investor Agreement. Lessor's Governing Documents comply in all respects with the requirements of Part B of Schedule 27.1.

(g) Lessor Loan Documents, Permitted Indebtedness and Permitted Lessor Liens A true, correct and complete list of the Lessor Loan Documents in effect as of the Effective Date is set forth in Exhibit I to this Lease, and Lessor has delivered to Lessee a true, correct and complete copy of each such Lessor Loan Document.

4.2 **Representations and Warranties of Lessee**. Lessee represents and warrants to Lessor as of the Effective Date as follows, it being understood and agreed that the following representations and warranties shall be deemed to be remade by a successor Lessee as of the effective date of any Permitted Lease Assignment:

(a) *Existence.* Lessee is duly organized, validly existing and in good standing under the laws of the State of Wyoming and has the requisite power and authority to execute and deliver this Lease and perform its obligations hereunder.

(b) *Binding Obligation.* This Lease has been duly authorized, executed and delivered by Lessee, and constitutes the valid and binding obligations of Lessee enforceable against Lessee in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and other laws affecting creditors' rights generally and subject to general principles of equity, whether considered in a proceeding in equity or at law.

(c) *Financial Statements.* If and to the extent any such financial information is provided prior to the date of this representation for a particular Lessee, the unaudited consolidated balance sheet and related statements of operations, stockholders' equity and cash flows for Lessee Guarantor or, if no there is no Lessee Guarantor for Lessee, copies of which were provided to Lessor by Lessee or Lessee's Guarantor prior to such Lessee becoming the Lessee hereunder, present fairly in all material respects the financial conditions and results of operations for such Person and its consolidated subsidiaries on a consolidated basis in accordance with GAAP consistently applied for such periods, and with respect to quarterly statements, subject to normal year-end audit adjustments and absence of footnotes.

(d) *No Violations.* The performance by Lessee of its obligations hereunder will not violate any provision of the Governing Documents of Lessee and, to Lessee's Actual Knowledge, will not (i) result in a material breach of any agreement to which Lessee is a party, (ii) violate in any material respect any provision of any judgment or order binding on Lessee, or (iii) constitute a material violation by Lessee of any law or governmental regulation applicable to Lessee.

ARTICLE V. RENT

5.1 **Base Rent**. Lessee agrees to pay to Lessor the Base Rent, without setoff, deduction, notice or demand except as expressly set forth in this Lease. Base Rent shall be payable on the first day of each calendar month, in advance, during the Term, with the first payment payable on the Effective Date. If Lessee is obligated to pay Base Rent for less than a full calendar month, Base Rent shall be prorated on a daily basis based upon the actual number of days in the prorated month.

5 . 2 Account for Payment of Base Rent. All payments of Base Rent shall be made to Account No. 359681346326, Account Name: Pinedale Corridor, LP, Reference "Transaction Description", Attn: Jennifer Duke, held at KeyBank National Association, 4910 Tiedeman Road 3 rd Floor Brooklyn, OH 44144, ABA No. 041001039, or such other account as Lessor shall advise Lessee in writing from time-to-time, but at least ten (10) Business Days before the date any payment of Base Rent is due.

5.3 **Payment to Party Claiming Rent**. If any party other than Lessor demands payment of Base Rent or Additional Rent due hereunder and alleges its right to receive the Base Rent or Additional Rent as a result of a Transfer of Lessor's interest in this Lease or otherwise, Lessee shall not be obligated to honor the demand unless Lessee receives written instructions to do so from the party to whom Lessee is then paying Base Rent or Additional Rent or shall have otherwise received evidence acceptable to Lessee of the right of the party making the demand. The withholding by Lessee of Base Rent or Additional Rent pending the determination of the rights of the party making the demand shall not be a default by Lessee and no interest at the Default Interest Rate or otherwise shall be due with respect to the delay pending such determination.

5.4 **Past Due Rent**. If any installment of Base Rent is not paid when due, Lessee shall pay to Lessor, on demand, as Additional Rent, interest on such installment from the date such installment was due to the date such installment is paid at the Default Interest Rate. If any other Additional Rent (except for such payments as Lessee is contesting in good faith or pursuant to a Tax Challenge or a Permitted Lessee Contest) is not paid within ten (10) Business Days following written demand from Lessor to Lessee (the "*Due Date for Other Additional Rent*"), Lessee shall pay to Lessor, on demand, as Additional Rent, interest on such installment from the applicable Due Date for Other Additional Rent to the date such other Additional Rent is paid at the Default Interest Rate.

ARTICLE VI. TAXES AND IMPOSITIONS

6.1 Taxes and Impositions

(a) Subject to the other terms and provisions of this <u>Article VI</u>, Lessee shall, before interest or penalties are due thereon, pay directly to the applicable third party and discharge all Taxes and Impositions accruing with respect to the Leased Property during the Term.

6.2 **Payment**. In the event that any Taxes or Impositions against any of the Leased Property may be paid in installments, Lessee shall have the option to pay such Taxes or Impositions in installments; and in such event, Lessee shall be liable only for those installments which become due and payable during the Term and which accrue or arise with respect to the Leased Property during the Term. Lessee shall prepare and file all tax reports required by Governmental Authorities which relate to the Taxes which accrue or arise with respect to the Leased Property during the Term. Lessee shall be deemed to have satisfied its obligation to pay Taxes as required hereby by paying such Taxes before the earlier of the date (i) any penalty (other than non-delinquent interest) accrues thereon, or (ii) any Lien is imposed against any Leased Property pursuant to Applicable Legal Requirements as a result of such failure (excluding Liens arising and attaching as a matter of law prior to the date such Taxes are due).

6.3 **Exclusions from Taxes and Impositions.** Notwithstanding the foregoing and for the avoidance of doubt, in no event will Lessee be required to pay, and the terms "Taxes" and "Impositions" shall exclude any and all of the following (collectively, the "*Excluded Matters*"):

(a) net income taxes (however denominated), gross receipts, net receipts or gross income taxes (however denominated, that are imposed in lieu of net or gross receipts or income taxes) or franchise taxes of Lessor, any Equity Investor or any Affiliate of Lessor; and

(b) any tax imposed with respect to the sale, exchange or other disposition by Lessor, any Equity Investor or any Affiliate of Lessor, in whole or in part, of the Leased Property, Lessor's interest in this Lease or any Equity Investor Interest; and

(c) taxes on, or with respect to or measured by the capital or net worth of Lessor, any Equity Investor or any Affiliate of Lessor or in the nature of a franchise, use, or margin tax or a tax for the privilege of doing business; and

(d) taxes or impositions imposed with respect to any period after the Term; and

(e) excess profits taxes, accumulated earnings taxes, capital gains taxes, succession or estate taxes or personal holding company taxes; and

(f) taxes imposed on or against or with respect to Lessor, any Equity Investor or any Affiliate of Lessor to the extent such taxes would not have been imposed if Lessor, any Equity Investor or any Affiliate of Lessor had not engaged in activities, or had a presence in, the taxing jurisdiction which activities or presence is unrelated to this Lease; and

(g) taxes imposed as a result of any voluntary or Involuntary sale, Transfer or other conveyance by Lessor, any Equity Investor or any Affiliate of Lessor; and

(h) taxes that are interest, penalties or additions to tax (or similar fees or charges) that are imposed as a result of the failure of Lessor, any Equity Investor or any Affiliate of Lessor to file any return or other filing properly and timely unless such failure is caused by Lessee failing to fulfill its obligations under this Lease; and

(i) any tax resulting from or that would not have been imposed but for the existence of Liens attributable to Lessor, any Equity Investor or any Affiliate of Lessor; and

(j) taxes to the extent that the amount of such tax is greater than the amount of taxes that would have been imposed if Lessor were not a successor or assignee of Pinedale Op .Co., LP; and

- (k) taxes imposed under the Code or ERISA with respect to any Person other than Lessee and its Affiliates; and
- (1) sales, transfer, recording, mortgage and similar taxes which Lessor is required to pay pursuant to the Purchase Agreement; and
- (m) foreign Taxes; and
- (n) taxes on rental or any other amounts paid by Lessee under this Lease; and

(o) any increase in taxes resulting from a Transfer or other conveyance by Lessor or any Equity Investor of all or any portion of the Leased Property or Lessor or any interest therein; and

- (p) withholding taxes of any kind; and
- (q) taxes, fees or amounts due and owing under or in connection with any Lien against the Leased Property, this Lease or any Equity Investor Interest; and
- (r) assessments, charges or other matters due and owning under any Other Recorded Documents executed by, through or under Lessor in violation of this Lease.

6.4 **Payment of Taxes and Impositions.** If bills or invoices in respect of a Tax or an Imposition are received by Lessor, Lessor shall, within ten (10) Business Days of Lessor's receipt thereof, but in any event at least ten (10) Business Days prior to the due date for the related Tax or Imposition (provided that Lessor has received such bill or invoice prior to ten (10) Business Days preceding such due date), deliver to Lessee any bill or invoice with respect to any Tax or Imposition. All taxing authorities shall be instructed to send all Tax invoices to Lessee. Within thirty (30) days after Lessor or Lessee has received confirmation of payment on account of Taxes, such Party shall provide to the other Party a copy of such confirmation that such Tax was paid.

6.5 **Abatements.** For the purpose of determining payments due from Lessee under this <u>Article VI</u>. Taxes and Impositions shall be the Taxes and Impositions are reduced by abatement, refund or rebate. If any abatement, refund or rebate is granted, the Taxes or Impositions, as applicable, shall be the Taxes or Impositions as so reduced. Any rebate, refund or abatement received subsequent to payment of such Taxes or Impositions by Lessee shall be refunded to Lessee by Lessor within ten (10) Business Days of receipt by Lessor, even if the abatement is received after the expiration or earlier termination of this Lease, and such obligations shall survive termination of this Lease. If Lessor or any taxing authority applies any rebate, refund or abatement as a credit against Taxes or Impositions due for a period following the termination of this Lease, then the rebate, refund or abatement that will be deemed received by Lessor upon granting of the rebate, refund or abatement. If Lessor or any taxing authority fails to refund any rebate, refund or abatement to Lessee within thirty (30) days after receipt (other than any rebate, refund or abatement that will be applied as a credit against Taxes and Impositions due for a batement realized by Lessor prior to payment of the Taxes and Impositions by Lessee shall result in the immediate reduction of Taxes and Impositions then due from Lessee to Lessor.

6.6 **Right to Contest Taxes.** Lessee shall have the right to file and prosecute to completion an application contesting the amount, validity, or application of any Taxes, contesting the assessed value of all

or any portion of any Leased Property, or seeking an abatement of any Taxes (any such application, a "*Tax Challenge*") either in its own name or in the name of Lessor, at no cost or expense to Lessor other than any *de minimus* cost or expense and throughout all administrative and judicial proceedings (trial court and all appeals) until there is a final, nonappealable determination. Lessee may discontinue a Tax Challenge at any time. If Lessee files or prosecutes a Tax Challenge, Lessor and its Affiliates shall cooperate and furnish any pertinent information needed for the Tax Challenge, at no cost or expense to Lessor other than any *de minimus* cost or expense see shall be entitled to be reimbursed out of the award for the reasonable costs and expenses occurred in connection with a Tax Challenge. Lessor shall not have the right to file a Tax Challenge with respect to any Taxes accruing during the Term unless the taxing authority requires the Tax Challenge to be filed by Lessor and Lessee requests Lessor to do so. Lessor shall immediately provide Lessee with a copy of any notices or demands from any Governmental Authority or other Persons which concerns Taxes. Upon request of Lessor at any time or from time-to-time, Lessee shall provide a written report to Lessor regarding the status of any such Tax Challenge.

In no event shall the manner in which Lessee pursues any Tax Challenge exacerbate in any material respect the risk to Lessor of civil or criminal liability, penalty or sanction, in addition to such risks as may exist for the matters that are the subject of the Tax Challenge prior to such Tax Challenge, and except for liabilities, penalties or sanctions for which Lessee may, and in fact does, post a bond. Further, the manner in which Lessee pursues a Tax Challenge shall not exacerbate in any material respect the risk to Lessor of defeasance of its interest in the Leased Property in addition to the risk of such defeasance as may exist for the matters that are the subject of the Tax Challenge prior to such Tax Challenge prior to such risk which Lessee may, and in fact does, bond around. Lessee agrees that Lessee shall use commercially reasonable efforts to diligently prosecute any such Tax Challenge at any time. Lessee shall promptly after the final determination of such Tax Challenge, fully pay and discharge the amounts which shall be levied, assessed, charged or imposed or be determined to be payable therein or in connection therewith, together with all penalties, fines, interest, costs and expenses thereof or in connection therewith, and perform all acts the performance of which shall be ordered or decreed as a result thereof.

6.7 **Right to Contest Impositions**. Lessee shall have the right to contest Impositions pursuant to Permitted Lessee Contests.

ARTICLE VII. CONDITION OF LEASED PROPERTY; MAINTENANCE

7.1 **Lessee Maintenance of Leased Property.** During the Term, Lessee shall (a) maintain the Leased Property in Good Condition and Repair, subject to reasonable and ordinary wear and tear, and subject to the provisions of <u>Articles X, XV</u> and <u>XVI</u>, and (b) pay all maintenance and operating costs of the Leased Property in the ordinary course of business. Lessee waives any right to (i) require Lessor to maintain, repair, replace or rebuild all or any part of the Leased Property or (ii) make repairs at the expense of Lessor pursuant to any Applicable Legal Requirements at any time in effect. Subject to Permitted Lessee Contests and the other provisions of this Lease, Lessee, at its own expense, will make (or cause to be made) such alterations and modifications in and additions to the Improvements and Personal Property as may be required from time-to-time to satisfy Applicable Legal Requirements in all material respects in a manner consistent with <u>Article X</u>. Nothing in this <u>Section 7.1</u> or otherwise in this Lease imposes any obligations upon Lessee, or waives any of Lessee's rights, with respect to Lessor's Environmental Liabilities.

7.2 **No Trespass.** Lessor authorizes Lessee to enforce any no trespass actions regarding any Real Property and to initiate any proceedings to remove any third parties from the Land (to the extent allowable under the applicable Easements) which Lessee's reasonable business judgment, deems

necessary or appropriate for Lessee's continued quiet enjoyment of the Leased Property. Lessee shall endeavor to give Lessor notice when exercising rights under this Section 7.2.

ARTICLE VIII. UTILITIES

8.1 **Payment of Utility Charges.** Lessee shall contract for, in its own name, and pay when due, all charges for the connection and use of water, gas, electricity, and other utility services, if any, supplied to the Leased Property during the Term. Under no circumstances shall Lessor be responsible for any interruption of any such utility service unless caused by the negligence, gross negligence or willful acts or omissions of any Lessor Party. If utilities serving the Leased Property are so disrupted on account of the negligence, gross negligence, or willful act or omission of any such Lessor Party, Lessor shall promptly restore the utilities at Lessor's sole cost and expense.

ARTICLE IX. USE

9.1 Lessee Use. The Leased Property may be used for any use permitted by Applicable Legal Requirements and the terms and conditions of the Easements and other Record Agreements. During such time as this Lease is in effect, Lessee shall, at Lessee's sole cost and expense and subject to Permitted Lessee Contests, comply in all material respects with all Applicable Legal Requirements now or hereafter in force relating to or affecting the Leased Property and Lessee's use, occupancy, operation, maintenance, alteration and/or improvement thereof. Notwithstanding the preceding or any other provision of this Lease to the contrary, (a) Lessor shall be fully responsible for any violation of Applicable Legal Requirements caused at any time by (i) any Lessor Party or (ii) any Lessor Lender; or (iii) any of their respective officers, agents, contractors, servants or employees, except to the extent such violation is caused by the failure of Lessee to perform its obligations hereunder, and (b) Lessee shall have no responsibility for any such matter described in <u>subsection (a)</u>. Lessee shall have the right to delay making the changes, alterations and/or additions Lessee is responsible for making pursuant to the immediately preceding provisions of this <u>Section 9.1</u> while Lessee is contesting in good faith the action or actions being taken by a Governmental Authority pursuant to a Permitted Lessee Contest

9.2 **Operating Requirement**. In no event shall Lessee be obligated to keep all or any part of any Leased Property operating. If Lessee discontinues operations, Lessee shall comply in all material respects with all Applicable Legal Requirements and otherwise comply in all material respects with the terms and conditions of this Lease except as to the continuing operation of the Leased Property. Subject to the other terms and conditions of this Lease regarding Lessee's right in certain circumstances to terminate this Lease and / or to offset Base Rent and Additional Rent against amounts due from Lessor hereunder, Lessee shall pay the Base Rent as and when due under this Lease during any period in which Lessee discontinues operations in whole or in part.

ARTICLE X. PERMITTED CAPITAL IMPROVEMENT AND ADDITIONAL LINES, PARTS AND LESSEE OTHER ACTIVITIES

10.1 **Permitted Capital Improvements and Additional Lines.**

(a) *Permitted Capital Improvements and Additional Lines.* Notwithstanding anything set forth in this Lease, Lessee, at its own expense and without Lessor's approval, shall have the right, at any time and from time-to-time as Lessee deems necessary or desirable, to make Permitted Capital Improvements and Additional Lines



Non-Removable Additions. If Lessee proposes to construct Permitted Capital Improvements and Additional Lines that are not readily removable or cannot (b) be detached from the Liquids Gathering System without adversely affecting the structural integrity or functionality of the Liquids Gathering System (each, a "Non-Removable Addition"), Lessee may provide Lessor written notice (a "Notice of Non-Removable Addition"), which Notice of Non-Removable Addition shall (i) describe the Non-Removable Addition in reasonable detail, (ii) provide a good faith estimate of the total cost of constructing the Non-Removable Addition, (iii) offer Lessor the option to both (A) pay for such Non-Removable Addition, and (B) engage the services of Lessee as construction agent for Lessor with respect to the construction of such Non-Removable Addition, and (iv) specify an increase in the Base Rate if Lessor elects to pay for and engage Lessee as Lessor's construction agent for such Non-Removable Addition. Lessor shall not have the option to pay for such Non-Removable Addition unless Lessor also engages the services of Lessee as construction agent for such Non-Removable Addition. Lessor shall, by written notice delivered to Lessee within ten (10) Business Days after receipt of the Notice of Non-Removable Addition, elect in writing either to pay for and engage Lessee as Lessor's construction agent for such Non-Removable Addition or to refrain from paying for such Non-Removable Addition; failure to give such timely notice shall constitute Lessor's election not to pay for such Non-Removable Addition; provided, however, that Lessor shall consider such offer as set forth in such Notice of Non-Removable Addition in good faith and use commercially reasonable efforts to obtain financing for such Non-Removable Addition. If Lessor accepts such offer, Lessor shall pay for such Non-Removable Addition and engage Lessee as Lessor's construction agent for such Non-Removable Addition. In the event Lessor elects to pay for such Non-Removable Addition and engage Lessee as Lessor's construction agent, such Non-Removable Addition shall become a part of the Leased Property under this Lease, with an increase in Base Rent as indicated in the Notice of Non-Removable Addition, unless Lessor and Lessee instead agree that such Non-Removable Addition shall be the subject of a separate lease, upon substantially the same terms as this Lease, but with Base Rent as set forth in the Notice of Non-Removable Addition. In the event Lessor timely elects to pay for such Non-Removable Addition and engage Lessee as Lessor's construction agent for such Non-Removable Addition on the terms in the Notice of Non-Removable Addition, (I) Lessor and Lessee shall execute the necessary documents to evidence such amendment to this Lease or a new lease for such Non-Removable Addition, as applicable, (II) Lessor and Lessee shall execute documents, reasonably satisfactory to each of them, governing advances by Lessor to pay costs of such Non-Removable Addition, (III) upon completion of such Non-Removable Addition, Base Rent under this Lease shall be increased or shall be payable under the new lease at the rate provided in the Notice of Non-Removable Addition, and (IV) Lessor shall be the owner of, and Lessee shall be the lessee of, such Non-Removable Addition. If Lessor elects not to pay for such Non-Removable Addition and Lessee elects to proceed with construction of the Non-Removable Addition, then subject to subsection (e), below, such Non-Removable Addition shall remain the sole property of Lessee, and this Lease shall continue in full force and effect

(c) **Removable Additions.** If Lessee proposes to construct Permitted Capital Improvements and Additional Lines that are <u>not</u> Non-Removable Additions (each, a "**Removable Addition**"), Lessee may provide Lessor written notice (a "*Notice of Removable Addition*"), which Notice of Removable Addition shall (i) describe the Removable Addition in reasonable detail, (ii) provide a good faith estimate of the total cost of constructing the Removable Addition,(iii) offer Lessor the option to both (A) pay for such Removable Addition, and (B) engage the services of Lessee as construction agent for Lessor with respect to the construction of such Removable Addition, and (iv) specify an increase in the Base Rate if Lessor elects to pay for and engage Lessee as Lessor's construction agent for such Removable Addition. Lessor shall not have the option to pay for such Removable Addition unless Lessor also engages the services of Lessee as construction agent for such Removable Addition. Lessor shall, by written notice delivered to Lessee within ten (10) Business Days after receipt of the Notice of Removable Addition; failure to give such timely notice shall constitute Lessor's election not to pay for such Removable Addition; provided, however, that Lessor shall consider such offer as set forth in such Notice of Removable Addition in good faith and use commercially reasonable efforts to obtain financing

for such Removable Addition. If Lessor accepts such offer, Lessor shall pay for such Removable Addition and engage Lessee as Lessor's construction agent for construction of such Removable Addition. In the event Lessor elects to pay for such Removable Addition and engage Lessee as Lessor's construction agent, such Removable Addition shall become a part of the Leased Property under this Lease, with an increase in Base Rent as indicated in the Notice of Removable Addition, unless Lessor and Lessee instead agree that such Removable Addition shall be the subject of a separate lease, upon substantially the same terms as this Lease, but with Base Rent as set forth in the Notice of Removable Addition. In the event Lessor timely elects to pay for such Removable Addition and engage Lessee as Lessor's construction agent for such Removable Addition on the terms in the Notice of Removable Addition, (I) Lessor and Lessee shall execute the necessary documents to evidence such amendment to this Lease or a new lease for such Removable Addition, as applicable, (II) Lessor and Lessee shall execute documents, reasonably satisfactory to each of them, governing advances by Lessor to pay costs of such Removable Addition, Base Rent under this Lease shall be increased or shall be payable under the new lease at the rate provided in the Notice of Removable Addition, and (IV) Lessor shall be the owner of, and Lessee shall be the lessee of, such Removable Addition. If Lessor the pay for such Removable Addition, then subject to <u>subsection (e)</u>, below, such Removable Addition shall remain the sole property of Lessee, and this Lease shall continue in full force and effect.

(d) Third Party Financing for Permitted Capital Improvements and Additional Lines Lessee shall at all times have the right to finance any Permitted Capital Improvements and Additional Lines with a third party without Lessor's consent; provided, however, that Lessee will notify Lessor in advance of entering into any such financing secured by a Lien on such Permitted Capital Improvements and Additional Lines (as applicable) and give Lessor an opportunity to make a proposal (non-binding on Lessee) to pay for, and engage Lessee as Lessor's construction agent for, such Permitted Capital Improvements and Additional Lines (as applicable). If requested by Lessee, Lessor agrees to cooperate in good faith with Lessee in the financing of any such Permitted Capital Improvements and Additional Lines (as applicable) that Lessee deems to be necessary or desirable for the operation of the Leased Property or the Lessee Other Activities.

(e) *End of Term Ownership.* At the end of the Term, (i) all Non-Removable Additions which Lessor elected not to pay for and make a part of the Leased Property pursuant to <u>subsection (b)</u>, above, will become the property of Lessor without compensation or reimbursement to Lessee, and (ii) all Removable Additions which Lessor elected not to pay for and make a part of the Leased Property pursuant to <u>subsection (c)</u>, above shall be removed by Lessee or, in the case of Additional Lines, detached by Lessee from the Liquids Gathering System unless otherwise agreed by Lessor in writing, and Lessee shall repair any damage to the Leased Property caused by such removal. The transfer of all Non-Removable Additions which become the property of Lessor at the end of the Lease Term pursuant to <u>subsection (i)</u>. (A) shall be made on an **AS IS WHERE IS** basis, without any express or implied warranties from Lessee other than special warranties of title (except as otherwise expressly agreed between the parties at the time of such transfer), provided, however, that such conveyance shall in any event be made by Lessee to Lessor free and clear of (I) any Liens placed on the Lease's exceeded Documents executed by, through or under Lessee in violation of this Lease, and (B) shall be evidenced by Lessee's execution of such deeds, bills of sale, or other appropriate documents conveying title to such Non-Removable Additions as Lessor and Lessee may reasonably agree, consistent with <u>subsection (A)</u> of this sentence.

10.2 **Replacement of Parts**. Lessee may, at its own cost and expense, remove in the ordinary course of maintenance, service, repair, overhaul or testing, or as otherwise required or permitted by this Lease, any Parts, whether or not worn out, lost, stolen, destroyed, seized, confiscated, damaged beyond repair or permanently rendered unfit for use. Lessee will, at its own cost and expense, replace any such Parts as is necessary to maintain the Leased Property in accordance with the standards for Lessee's maintenance obligations as set forth in Section 7.1 of this Lease. All replacement Parts shall be in as good operating

condition as, and shall have a value (including the residual value), utility and remaining useful life at least equal to the Parts replaced, assuming such replaced Parts were in condition and repair required to be maintained by, and otherwise in compliance with, the terms of this Lease (nothing in this Lease, however, shall be construed to require Lessee to replace any Parts having a useful life adequate to operate the Leased Property through the end of the Term). All Parts at any time removed from any item or portion of the Leased Property shall remain the property of Lessor until such time as such Parts shall be replaced by Parts which have been incorporated or installed in or attached to such item or portion of the Leased Property and which meet the requirements for replacement Parts specified above. Immediately upon any replacement Part becoming incorporated into or installed in or attached to any item or portion of the Leased Property, without further act (except in the case of replacement property temporarily installed on an emergency basis), (a) tile to such replacement Part shall thereupon vest in Lessor, (b) such replacement Part shall become subject to this Lease and be deemed part of the Leased Property for all purposes to the same extent as the Parts originally incorporated or installed in or attached to the Leased Property, and (c) tile to the removed Part shall thereupon vest in Lesser AS **IS WHERE IS**, free and clear of all rights of Lessor, and shall no longer be deemed a part of the Leased Property. Notwithstanding the foregoing, "Parts" means all appliances, parts, instruments, appurtenances, accessories, furnishings, spare parts, and other equipment and property of whatever nature which constitute a portion of the Improvements or Personal Property and any replacements or substitutions therefor made under this <u>Section 10.2</u>. "Parts" shall not in any event include (i) any Lessee Property, (ii) removed Parts as provided in <u>subsection (c)</u>, above, or (iii) any Permitted Capital Improvements and

10.3 Lessee Other Activities. Notwithstanding anything set forth in this Lease, Lessee, at its own expense and without Lessor's approval, shall have the right at any time and from time-to-time to conduct Lessee Other Activities.

ARTICLE XI. REGULATORY ISSUES

11.1 **Lessee's Rights.** Except (i) as provided in <u>Section 11.2</u>, below, and (ii) to the extent Lessor is required to engage in such communications by Applicable Legal Requirements which are not the subject of a Permitted Lessee Contest and after consultation with Lessee regarding Lessor's concern that Lessor's communication with Governmental Authorities is required by Applicable Legal Requirements, but notwithstanding any other provision of this Lease to the contrary:

(a) Ultra Lessee; Leased Property. For so long as an Ultra Entity Person is the Lessee under this Lease, Lessee shall have the <u>exclusive</u> right and obligation to (I) deal with all Governmental Authorities, including the Bureau of Land Management and the Wyoming Oil and Gas Conservation Commission, and (II) conduct all regulatory proceedings, governing or affecting the Leased Property or the ownership, use or operation thereof.

(b) *All Lessees; Lessee Property and Lessee Other Activities* Lessee shall have the <u>exclusive</u> right and obligation to (I) deal with all Governmental Authorities, including the Bureau of Land Management and the Wyoming Oil and Gas Conservation Commission, and (II) conduct all regulatory proceedings, governing or affecting the ownership, use or operation of the Lessee Property or the Lessee Other Activities.

(c) *Other Lessees; Leased Property.* If an Ultra Entity Person is not the Lessee under this Lease, Lessee shall have the right and obligation, but not the exclusive right and obligation, to (I) deal with all Governmental Authorities, including the Bureau of Land Management and the Wyoming Oil and Gas Conservation Commission, and (II) conduct all regulatory proceedings, governing or affecting the use or operation of the Leased Property.

Consistent with the foregoing subsections (a) through (c) of this Section 11.1, Lessor hereby designates Lessee as Lessor's agent with respect to, and solely with respect to, the exercise of Lessor's Easement Rights in and to the BLM Easements, but only during the term of and to the limited extent set forth in this Lease. This designation creates a special agency relationship, and Lessee and Lessor expressly acknowledge that the scope of Lessee's authority to act as agent for Lessor is expressly limited to the use of the BLM Easements as set forth in this Lease and those certain Bureau of Land Management regulatory matters which are enumerated in subsections (a) through (c) of this Section 11.1. Except for such use and enumerated matters, Lessee has and shall have no other power, authority or right to act as agent for Lessor, and Lessee shall have no implied powers or authority to act as agent for Lessor.

At Lessee's request, Lessor shall execute all such documents and take all such action as may be necessary to ensure and/or evidence Lessee's rights under this <u>Section 11.1</u> unless, in Lessor's reasonable and good faith judgment, such documents or actions will directly result in payments required to be made by Lessor which Lessee is unwilling to pay on behalf of Lessor and, upon such determination by Lessor and prior to taking any such action, Lessor shall promptly notify Lessee in writing of such determination by Lessor, and Lessor and Lessor and Lessee shall work diligently and in good faith to resolve their differences with respect to the requested action or document(s). Notwithstanding the foregoing, Lessor shall in any event be required to execute such documents and take all such actions as may be required to extend the terms of the BLM Easements beyond the Lease Term, it being understood, however, that nothing set forth herein shall obligate Lessor or Lessee for the payment of any extension fee or similar charge imposed by BLM as a condition to such extension other than *de minimus* costs and expenses.

For the avoidance of doubt, but subject to the introductory language of this <u>Section 11.1</u> and the immediately preceding paragraph, (i) Lessor shall not in any event engage in any written or verbal communication with any Governmental Authority, including the Bureau of Land Management and the Wyoming Oil and Gas Conservation Commission, regarding or in any way relating to the ownership, use or operation of the Lessee Property or any Lessee Other Activities, (ii) for so long as an Ultra Entity Person is the Lessee under this Lease, Lessor shall not engage in any written or verbal communication with any Governmental Authority, including the Bureau of Land Management and the Wyoming Oil and Gas Conservation Commission, regarding the ownership, use or operation of the Leased Property, and (iii) for so long as a Person other than an Ultra Entity Person is the Lessee under this Lease, Lessor shall not engage in any written or verbal communication with any Governmental Authority, including the Bureau of Land Management and the Wyoming Oil and Gas Conservation Commission, regarding the ownership, use or operation of the Leased Property, and (iii) for so long as a Person other than an Ultra Entity Person is the Lessee under this Lease, Lessor shall not engage in any written or verbal communication with any Governmental Authority, including the Bureau of Land Management and the Wyoming Oil and Gas Conservation Commission, regarding the use or operation of the Leased Property without Lessee's prior written consent, which consent may be conditioned, to the extent permitted by Applicable Legal Requirements, upon Lessee's review and approval of any such written communications. Lessee shall promptly provide Lessor with a copy of any notices or demands received by Lessee from any Governmental Authority which concerns any Applicable Legal Requirement.

Lessor acknowledges and stipulates that Lessee may suffer irreparable harm in the event of a breach of the provisions of this<u>Section 11.1</u> by Lessor for which Lessee has no adequate remedy at law. Therefore, in addition to all other remedies available pursuant to the terms of this Lease or at law, Lessee shall have the right to obtain immediate injunctive or other equitable relief upon a breach of this <u>Section 11.1</u> by Lessor.

11.2 Lessor's Rights. Notwithstanding the foregoing Section 11.1, but subject in any event to Lessee's right to pursue Tax Challenges and Permitted Lessee Contests, at such time as an Ultra Entity Person is not the Lessee under this Lease, if Lessor determines, in Lessor's reasonable judgment, that Lessee has not taken sufficient action to protect the interests of Lessor in regard to any such regulatory matter (including, without limitation, regulatory proceedings) with respect to the Leased Property, upon prior

written notice to Lessee, Lessor may take such action as may be necessary in order for Lessor to be and remain in compliance with Applicable Legal Requirements as they relate to Lessor and the Leased Property. Lessor shall promptly provide Lessee with a copy of any notices or demands received by Lessor from any Governmental Authority which concerns any Applicable Legal Requirement.

11.3 **No Application to Condemnation**. Lessor and Lessee acknowledge and agree that the provisions of this<u>Article XI</u> shall not prohibit Lessor from cooperating in condemnation proceedings as required by <u>Article XVI</u>.

ARTICLE XII. MECHANIC'S LIENS

12.1 **Lessee's Obligations.** If any mechanic's, materialmen's or other Liens against any of the Lessor's interest in the Leased Property in connection with any materials, labor or equipment furnished or claimed to have been furnished to or for any Lessee Party are filed against Lessor's interests in the Leased Property on account of action or inaction by any Lessee Party (including by Lessee in its capacity as Lessor's construction agent for Removable Additions or Non-Removable Additions under <u>Article X</u>, other than any such Liens filed as a result of materials, labor or services for Removable Additions or Non-Removable Additions and for which Lessor has failed to meet its obligation to pay the cost of such materials, labor or services), then Lessee shall promptly cause such Lien to be discharged (whether by payment or bond), or, if Lessee desires to contest the Lien, Lessee may do so as long as the enforcement of the Lien is stayed pursuant to a Permitted Lessee Contest, and (b) if a Lessor Lender with respect to Permitted Indebtedness requires Lessor's interest in the Leased Property to be free of such mechanic's or materialmen's liens, then upon request of Lessor or such Lessor Lender, Lessee shall either discharge the Lien or post a bond sufficient to cover the amount of the Lien and interest, penalties and costs that will be payable to discharge the Lien assuming its validity. Notwithstanding the foregoing, Lessee shall have no obligation whatsoever for or with respect to Liens arising from Lessor's failure to advance payments for Non-Removable Additions or Removable Additions or Removable Additions or sort cause as Lessor's construction agent as provided in Section 10.1

12.2 **Lessor's Obligations**. Lessor shall not permit any mechanic's, materialmen's or other Liens against any Leased Property in connection with any materials, labor or equipment furnished, or claimed to have been furnished, to or for Lessor, and if any such Liens shall be filed against any Leased Property on account of any action or inaction by Lessor (other than any action or inaction alleged against Lessor resulting from the failure of Lessee to perform its obligations hereunder), Lessor shall cause the Lien to be discharged, provided that if Lessor desires to contest any such Lien (whether by payment or bond), it may do so as long as enforcement of the Lien is stayed pursuant to a Permitted Lessor Contest.

ARTICLE XIII. LESSEE PROPERTY

13.1 No Lien. Lessee is and shall remain the owner of the Lessee Property, other than Non-Removable Additions at the end of the Lease Term as provided in Section 10.1(e). Lessor shall not have any Lien on any of the Lessee Property for the performance of Lessee's obligations under this Lease or otherwise. LESSOR WAIVES AND DISCLAIMS ALL STATUTORY AND CONTRACTUAL LIEN RIGHTS IN LESSEE PROPERTY NOW OR HEREAFTER PLACED AT OR WITH THE LEASED PROPERTY. Lessor agrees to execute an agreement with Lessee or Lessee's lender to evidence such waiver of Lessee's Lien rights in Lessee Property and, following a default by Lessee under any Lessee mortgage, allow Lessee's lender access to the Lesseor, Lesser and remove Lessee Property subject to the terms of such agreement, provided that the form and substance of any such agreement is reasonably satisfactory to Lessor, Lessee and Lessee's lender (including in such agreement reasonable conditions and time periods within which such Lessee Property may be removed). Lessee shall have the right to remove

Lessee Property at any time or times during the Term and for a period of thirty (30) days after the expiration of the Term, without any obligation to pay Lessor holdover rent under Section 26.4.

13.2 **Sole Risk of Lessee**. All of Lessee Property is at Lessee's sole risk and if any of the Lessee Property is destroyed or damaged other than as a result of the negligence or willful misconduct of a Lessor Party, no part of the destruction or damage shall be the responsibility of Lessor. If the damage is caused by the negligence or willful misconduct of a Lessor Party, Lessor shall be responsible for any damage or destruction, subject, however, to the waiver of subrogation provisions of <u>Section 14.4</u>. The term "negligence" as used in this <u>Section 13.2</u> shall not include negligence imputed as a matter of law to Lessor solely by reason of Lessor's failure to act in respect of matters which are or were the obligation of Lessee under this Lease or solely as a result of Lessor's ownership of the Leased Property.

13.3 Lessee's Ability to Grant Liens on Lessee Property. Nothing contained in this Lease shall be deemed to limit or otherwise restrict Lessee's ability to grant a Lien on Lessee's interest in the Leased Property under this Lease or on Lessee Property, in any such case to a lender not Affiliated with Lessee for the purpose of securing indebtedness of Lessee or its Affiliates to such lender. Lessor acknowledges and agrees, following a default by Lessee under any Lessee mortgage, to allow Lessee's lender access to the Leased Property to seize and remove all or any of the Lessee Property and Lessor agrees to execute an agreement with Lessee's lender to evidence such right as provided in Section 13.1.

ARTICLE XIV. INSURANCE

14.1 **Insurance Coverage**. Subject to Section 14.3 below, from and after the Effective Date, Lessee shall maintain with respect to the Leased Property at all times during the Term, at no cost or expense to Lessor, the following types and amounts of insurance:

(a) *Liability Insurance.* Commercial general liability insurance against claims for bodily injury, death or property damage occurring on, in or about the Leased Property, which insurance shall (i) be written on an "Occurrence Basis", and shall provide minimum protection with a combined single limit in the amount of \$1,000,000 per occurrence and \$2,000,000 annual aggregate; and (ii) include premises and operations liability coverage, products and completed operations liability coverage, and blanket contractual liability coverage. In addition, Lessee shall maintain auto liability insurance in an amount not less than \$1,000,000;

(b) *Worker's Compensation and Employer's Liability Insurance.* Worker's Compensation and Employer's Liability Insurance in accordance with all applicable federal and state law, which shall cover all Lessee's personnel, invitees, sublessors, guests, consultants, and similar Persons performing under this Lease, with an Employer's Liability limit shall of not less than one million Dollars (\$1,000,000) or such greater amount as may be required by law; and

(c) **Property Insurance.** "All Risk" Insurance, equal to the full replacement value (as declared) of the Liquids Gathering System, which shall be payable in case of loss and held, paid or applied as provided in <u>Section 15.1</u> hereof. The policy shall provide that no act or omission of any Person named as insured shall invalidate the interest of, or be a defense against, any other Person named as insured.

(d) *Excess Liability Insurance*. Excess Liability Insurance covering liabilities arising out of operations worldwide, with a combined minimum limit of not less than \$25,000,000 over and above the primary liability limits of all underlying insurance policies as required in this Section.

(e) Builders Risk Insurance. For any alteration, addition, repair or replacement to the Leased Property by Lessee (but excluding any Removable Additions or Non-Removable Additions for which Lessee is acting as Lessor's construction agent under <u>Article X</u>) for which the estimated cost exceeds \$500,000,

completed value builder's risk insurance upon the entire work for the full insurable value replacement cost coverage.Lessor and Lessee acknowledge and agree that the coverage required hereunder may be provided through a combination of policies that is different than those described above. In addition, Lessee may, at its election, maintain insurance covering additional casualties and events.

14.2 **Insurance Requirements Generally**. The insurance required by <u>Section 14.1</u> shall be written by companies having an A.M. Best Insurance Reports rating of not less than "A-" and a financial size category of at least "VII", and all such companies shall be authorized to do an insurance business in the State of Wyoming, or as otherwise agreed to by Lessor. All insurance policies required to be maintained pursuant to the terms of this Lease (other than the Worker's Compensation Insurance), shall be endorsed to name Lessor and its successors and assigns (and, at Lessor's written request and to the extent reasonably practicable, the Lessor Lender or its Agent, Administrative Agent or Collateral Agent) as additional insureds and to provide that such insurance shall be primary over any insurance that may be maintained by Lessor, and, except during the Lease Term (when all insurance proceeds are delivered and paid to Lessee, as loss payee). Notwithstanding anything to the contrary in this Lease, for so long as (a) an Ultra Entity Person, or (b) an Investment Grade Person is the Lessee under this Lease, then in any such case all insurance policies may have such deductibles and self-insured retention amounts as Lessee deems reasonable. If the Lessee is not as provided in the immediately preceding sentence, all insurance retention amounts approved by Lessor, which approval shall not be unreasonably withheld so long as the proposed deductibles and self-insurance required by this Article may be blanket with other insurance maintained by Lessee or any Affiliate of Lessee and may be effected by any combination of basic and excess or umbrella coverage provided: (i) the total amount of the insurance available shall be at least the protection equivalent of separate policies in the amounts required; and (ii) in all other respects, all of the policies shall comply with the applicable provisions of this Article.

Lessee shall pay as they become due all premiums for the insurance required by this<u>Article XIV</u>, shall renew or replace each policy if required to comply with the insurance requirements of this Lease, and shall deliver to Lessor a certificate or other evidence of the existing policy and any renewal or replacement policy (on an ACORD 27 form, in the case of property insurance, and on an ACORD 25 form, in the case of liability insurance, and, in either case, otherwise reasonably satisfactory to Lessor, to the extent reasonably available) as soon as available, but in any event not later five (5) Business Days after the date the new coverage is effective (<u>it being understood</u> that in no event shall Lessee allow any insurance coverage to lapse if and to the extent such insurance is necessary to comply with the requirements of this Lessee's failure to comply with any of the foregoing requirements of this <u>Article XIV</u> within five (5) Business Days of the giving of written notice by Lessor to Lessee (or such shorter period as may be required to maintain the insurance required by this <u>Article XIV</u> in full force and effect), Lessor shall be entitled to procure such insurance. Any sums expended by Lessor in procuring such insurance shall be Additional Rent and shall be repaid by Lessee upon written demand therefor by Lessor, together with interest thereon from the applicable Due Date for Other Additional Rent at the Default Interest Rate.

14.3 Self-Insurance. During any part of the Term, for so long as an Ultra Entity Person or another Investment Grade Person is the Lessee under this Lease, Lessee may self-insure for some or all insurance obligations contained hereunder. If the foregoing sentence is not applicable, the amount of any self-insurance retention shall be subject to the limitations set forth in <u>Section 14.2 above</u>. If Lessee desires to self-insure pursuant to this<u>Section 14.3</u>, Lessee shall deliver to Lessor prior to self-insuring a notice that it intends to self-insure hereunder and a brief summary of the extent to which it intends to self-insure. Ultra Wyoming, on behalf of itself and Ultra Resources and any successor Lessee who is an Ultra Entity Person, hereby notifies Lessor that it intends (and they intend) to self-insure with respect to fire, casualty and

property insurance and builder's risk insurance with respect to the Leased Property. The beneficiaries of Lessee's self-insurance shall be afforded no less protection than if such self-insured portion was fully insured by an insurance company of the quality and caliber required hereunder, including the provision of a legal defense and the payment of claims within the same time period that a third-party insurance carrier would have paid such claims. The waiver of subrogation provisions of <u>Section 14.4</u> shall be applicable to any self-insured exposure.

14.4 **Waiver of Subrogation**. Each of Lessor and Lessee hereby releases the other from any and all liability for loss or damage caused by fire or any of the extended coverage casualties or any other casualty which shall be brought about by the fault or negligence of the other party or any persons for whom such other party is responsible, <u>provided</u> that this release shall be in force and effect only with respect to loss or damage occurring during such time as the respective party's policies of property insurance shall contain a clause to the effect that this release shall not affect such policies or right of the releasing party to recover thereunder, and the release shall apply only to the extent of the releasing party's insurance coverage, provided that, with respect to Lessee's release of Lessor hereunder, such release shall also apply with respect to any loss occurring during such time as the Lessee is self-insured in accordance with the terms of <u>Section 14.3</u>. Any insurance policy held by Lessee or Lessor, to the extent that waiver is reasonably and customary for the policy in question, shall include a waiver of subrogation clause so long as the same is obtainable. Nothing herein shall be deemed an obligation on the part of Lessor to maintain any insurance maintained by Lessor shall serve as excess coverage.

14.5 **No Release from Liability**. Except as provided in <u>Section 14.4</u>, nothing in this <u>Article XIV</u> shall be deemed to release either party from liability for damages resulting from the negligence or willful misconduct of such party or from responsibility for repairs necessitated by such negligence or willful misconduct. By requiring insurance as provided herein, neither Lessor nor Lessee represents that coverage and limits will necessarily be adequate to protect Lessee or Lessor, and such coverage and limits shall not be deemed as a limitation of the liability of Lessee or Lessor under any indemnification provisions in this Lease. Failure of Lessor to demand such certificate or other evidence of full compliance with these insurance requirements or failure of Lessor to identify a deficiency from evidence that is provided shall not be construed as a waiver of Lessee's obligation to maintain such insurance.

ARTICLE XV. FIRE AND OTHER CASUALTY

15.1 Fire and Other Casualty.

(a) *General Provisions.* If all or any part of the Leased Property is damaged or destroyed by fire or other casualty during the Term, then Lessee shall give prompt notice of such event to Lessor and, except as otherwise provided in this Article, Lessee shall, at Lessee's cost and expense, promptly thereafter repair, restore or replace the Leased Property, or any such part thereof, to substantially the same condition it was in immediately prior to the casualty (subject to any changes to all or any such part of the Leased Property that Lessee intends to make to the extent permitted under this Lease). Notwithstanding the foregoing or anything to the contrary in this <u>Article XV</u> or otherwise, if this Lease is terminated under this <u>Article XV</u> or <u>Section 25.5</u> (Burdensome Buyout Event) as a result of such fire or other casualty, (i) Lessee shall not be obligated to repair, restore or replace the Leased Property after such fire or other casualty or to pay for such repairs, restoration or replacement, and (ii) neither Lessor nor any other Lessor Indemnified Party shall have any claim against Lessee with respect to the repair, restoration or replacement of the Leased Property pursuant to this <u>Section 15.1</u> or <u>Section 25.5</u>. Unless this Lease is terminated pursuant to this <u>Section 15.1</u> or <u>Section 25.5</u> as a result of such fire or other casualty, (A) this Lease shall remain in effect following such fire or other casualty, (B) Lessee shall, at Lessee's cost and expense, promptly repair, restore or replace the Leased Property, or any part thereof, as a result of such fire or other casualty, to substantially the same

condition it was in immediately prior to the casualty (subject to changes permitted herein), (C) the rights and obligations of Lessor and Lessee shall remain in effect without abatement, (D) all Base Rent shall continue unabated, and (E) subject to <u>Section 15.2</u>, below, all Casualty Proceeds recovered on account of any damage or destruction to all or any part of the Leased Property by fire or other casualty shall be paid to Lessee.

(b) Lease Termination Option and Purchase Offer. Notwithstanding the above, if the value, as determined by Lessee in good faith, of the damaged or destroyed portions of the Leased Property (determined as of the Casualty Date as if such damage or destruction had not occurred) is greater than fifty percent (50%) of the Insured Value, then Lessee may, by written notice delivered to Lessor within sixty (60) days after such determination, either (i) propose to continue this Lease in effect, or (ii) propose to terminate this Lease and purchase the Leased Property for the purchase price as specified by Lessee in such notice (any such written notice by Lessee electing under <u>subsection (ii)</u> to terminate this Lease and purchase the Leased Property shall be referred to as a "Lessee Casualty Termination Notice"). Lessee's failure to deliver its Lessee Casualty Termination Notice within such sixty (60) day period shall constitute Lessee's election to continue the Lease in effect and to repair, restore or replace the damaged or destroyed Leased Property as provided in Section 15.1(a).

(c) *Casualty Response Notice*. If Lessee delivers its Lessee Casualty Termination Notice to Lessor, then Lessor shall have a period of no more than fifteen (15) days in which to notify Lessee in writing of Lessor's acceptance or rejection of the Lessee Casualty Termination Notice ("*Casualty Response Notice*"). Lessor's failure to provide its Casualty Response Notice within such fifteen (15) day period shall be deemed a rejection of the Lessee Casualty Termination Notice and the Lease shall continue in effect.

(d) Acceptance of Lessee Casualty Termination Notice and Termination of Lease. If Lessor accepts the Lessee Casualty Termination Notice, (i) Lessee shall purchase the Leased Property from Lessor as provided in Section 15.1(e) on a date no later than sixty (60) days after Lessor's Casualty Response Notice and this Lease shall terminate on such date (the "Casualty Termination Date"), (ii) except the terms and provisions of this Lease which survive termination, neither Lessor nor Lessee shall have any rights or obligations under this Lease arising from and after the Casualty Termination Date, (iii) no further Base Rent or Additional Rent shall accrue with respect to the period after the Casualty Termination Date, (iv) Lessee shall have no repair, restoration or replacement obligations with respect to the fire or other casualty giving rise to the Lessee Casualty Termination Notice and no obligation to pay for such repair, restoration of replacement, (v) neither Lessor nor any other Lessor Indemnified Party shall have any claim against Lessee with respect to the repair, restoration of replacement of liability and obligations under its Lessee Guaranty, Ultra Resources Guaranty, and each Lessor Guarantor shall be released of liability and obligations under its Lessor Guaranty with respect to (x) as to Lessee Guarantors and Ultra Resources Guaranty, matters for which Lessee is released in <u>subsections (iv)</u> and (y) matters arising or accruing from and after the Casualty Termination Date, it being understood and agreed, however, that Lessee Guarantors, Ultra Resources and Lessor Guaranty shall be released shall the respective of the section 15.1(d), and (vii) all Casualty Proceeds shall be paid to Lessee.

(e) Closing of Purchase Offer. In addition, if Lessor accepts the Lessee Casualty Termination Notice, on the Casualty Termination Date, (i) Lessee shall pay to Lessor the purchase price specified by Lessee in the Lessee Casualty Termination Notice or such other purchase price as Lessor and Lessee may mutually agree, (ii) Lessor shall convey the Leased Property to Lessee or its designee on an AS IS WHERE IS basis, without any express or implied warranties from Lessor other than special warranties of title, provided, however, that such conveyance shall in any event be made by Lessor to Lessee or its designee free and clear of (A) all Permitted Lessor Liens, (B) any other Liens placed on the Leased Property by, through or under Lessor; and (C) any Other Recorded Documents executed by, through or under Lessor in violation of

this Lease, and (iii) Lessor will deliver to Lessee or its designee such conveyance documents, each in form and substance reasonably satisfactory to Lessee, as Lessee may reasonably request, to transfer the ownership of the Leased Property to Lessee free and clear of the Liens and other items described in subsections (A)-(C) above.

(f) *Continuation of Lease if Lessor Rejects the Lessee Casualty Termination Notice* If Lessor rejects or is deemed to have rejected the Lessee Casualty Termination Notice, such Lessee Casualty Termination Notice with respect to termination of the Lease and Lessee's offer to purchase the Leased Property shall be deemed to be automatically rescinded and this Lease shall remain in full force and effect in accordance with its terms.

(g) No Waiver. Upon each occurrence of a fire or other casualty for which Lessee has the right under this <u>Section 15.1</u> to give a Lessee Casualty Termination Notice, Lessee shall be entitled to give its Lessee Casualty Termination Notice with respect to such fire or other casualty, even if Lessee previously elected not to give such notice with respect to the occurrence of a prior fire or other casualty and even if prior Lessee Casualty Termination Notice(s) were rejected or deemed rejected by Lessor hereunder.

15.2 **Restoration Conditions**. Provided that this Lease has not been terminated pursuant to this <u>Article XV</u> above, then Lessee's repair, restoration or replacement of the Leased Property shall be subject to the following requirements:

(a) **Payment of Casualty Proceeds to Lessee.** All Casualty Proceeds shall be paid by the insurer to Lessee.

(b) **Repair Costs in Excess of \$2,000,000**. If the cost of such repairs, restoration or replacement as a result of a casualty exceeds \$2,000,000 with respect to such casualty, (A) Lessee shall provide to Lessor and Lessor Lender (or the Agent for such Lessor Lender if requested in writing by Lessor) such reasonable documentation related to such repairs, restoration or replacement as Lessor and Lessor Lender (or the Agent for such Lessor Lender if requested in writing by Lessor) may reasonably request to confirm that such repairs, restoration or replacements have been made in the manner required by this Lease, (B) if such repairs, restoration or replacement have not been completed within ninety (90) days following the date of such casualty, Lessee shall pay to the Lessor Lender (or the Agent for such Lessor Lender if requested in writing by Lessor) the amount by which the total cost of such repairs, restoration or replacement exceeds \$2,000,000, reduced by repairs, restoration or replacements progress, and (C) Lessor and Lessor Lender (or the Agent for such Lessor Lender if requested in writing by Lessor). Lessor Lender if requested in writing by Lessor lender in writing by Lessor) will have the right to inspect the Leased Property during normal business hours and after prior reasonable written notice, with Lessee's representative present and upon the condition that Lessor and Lessor Lender (or the Agent for such Lessor Lender if requested in writing by Lessor) comply with Lessee's rules and regulations, including safety, (I) to inspect the damage, and (II) to confirm that such repairs, restoration or replacements are being or have been made in the manner required by this Lease; provided that no such inspection shall interfere with the normal operation of the Leased Property or the business of Lessee or its Affiliates.

ARTICLE XVI. CONDEMNATION

16.1 **Condemnation Damages and Awards.** Except to the extent Lessor may be required to make the same available to Lessee under this<u>Article XVI</u> and as otherwise set forth in this <u>Article XVI</u>. Lessee assigns to Lessor any and all rights it may have to any Taking Proceeds and agrees to execute such instruments as may be requested by Lessor to evidence the assignment of such Taking Proceeds to Lessor. Notwithstanding the foregoing, Lessee may make a separate claim for any damages payable for any of the

Lessee Property, any so-called special damages to Lessee for interruption to Lessee's operations or otherwise, or any damages for relocation (collectively, the 'Lessee Condemnation Proceeds'), and none of such Lessee Condemnation Proceeds shall constitute Taking Proceeds. Unless this Lease is terminated pursuant to this <u>Article XVI</u> or <u>Section 25.5</u> (Burdensome Buyout Event) in connection with a Taking, this Lease shall remain in effect, and the rights and obligations of Lesser and Lessee shall remain in effect without abatement. Lessor and Lessee shall cooperate in good faith to maximize the Taking Proceeds (but without impairing Lessee's claim for Lessee Condemnation Proceeds) consistent with the terms of this Lease.

16.2 Apportionment of Award, Termination and Purchase Offer.

(a) *General Taking Provisions.* If all or any part of the Leased Property is subject to a Taking during the Term, then Lessee shall give prompt notice of such event to Lessor and except as otherwise provided in this Article, Lessee shall, at Lessee's cost and expense, promptly thereafter repair, restore or replace the Leased Property, or any such part thereof, to substantially the same condition it was in immediately prior to the Taking (subject to any changes to all or any such part of the Leased Property that Lessee intends to make to the extent permitted under this Lease). Notwithstanding the foregoing or anything to the contrary in this <u>Article XVI</u> or otherwise, if this Lease is terminated under this <u>Article XVI</u> or Section 25.5 (Burdensome Buyout Event) as a result of such Taking, (i) Lessee shall not be obligated to repair, restore or replace the Leased Property after such Taking or to pay for such repairs, restoration or replacement, and (ii) neither Lessor nor any other Lessor Indemnified Party shall have any claim against Lessee with respect to such repair, restoration or replacement of the Property or Losses arising from termination of this Lease or Lessee's purchase of the Leased Property pursuant to this <u>Section 16.2</u> or <u>Section 25.5</u>. Unless this Lease is terminated pursuant to this <u>Section 16.2</u> or <u>Section 25.5</u> in connection with such Taking, this Lease shall remain in effect following such Taking, the rights and obligations of Lessor and Lessee shall remain in effect without abatement, all Base Rent shall continue unabated, and all Lessee Condemnation Proceeds and, subject to <u>Section 16.3</u>, below, Taking Proceeds recovered on account of such Taking shall be paid to Lessee.

(b) *Lease Termination Option and Purchase Offer*. Notwithstanding the above, if as a result of the Taking, in Lessee's good faith judgment, the Leased Property can no longer be profitably operated or is no longer suitable for its intended use, then Lessee may, in lieu of repairing, restoring or replacing the same, by written notice delivered to Lessor within sixty (60) days after such determination, either (i) propose to continue this Lease in effect, or (ii) propose to terminate this Lease and purchase the Leased Property for the purchase price specified by Lessee in such notice (any such written notice by Lessee electing under <u>subsection (ii)</u> to terminate this Lease and purchase the Leased Property shall be referred to as a "*Lessee Taking Termination Notice*"). Lessee's failure to deliver its Lessee Taking Termination Notice within such sixty (60) day period shall constitute Lessee's election to continue the Lease in effect and to restore, repair or replace the damaged or destroyed Leased Property as provided in <u>subsection (i)</u> of this <u>Section 16.2(b)</u>.

(c) *Taking Response Notice*. If Lessee delivers its Lessee Taking Termination Notice to Lessor, then Lessor shall have a period of no more than fifteen (15) days in which to notify Lessee in writing of Lessor's acceptance or rejection of the Lessee Taking Termination Notice ("*Taking Response Notice*"). Lessor's failure to provide its Taking Response Notice within such fifteen (15) day period shall be deemed a rejection of the Lessee Taking Termination Notice and the Lease shall continue in effect.

(d) Acceptance of Lessee Taking Termination Notice and Termination of Lease. If Lessor accepts the Lessee Taking Termination Notice, (i) Lessee shall purchase the Leased Property from Lessor as provided in <u>Section 16.2(e)</u> on a date no later than sixty (60) days after Lessor's Taking Response Notice and this Lease shall terminate on such date (the "Taking Termination Date"), (ii) except as provided in <u>Sections 25.2, 25.3</u> and <u>25.4</u>, and the other terms and provisions of this Lease which survive termination, neither Lessor nor Lessee shall have any rights or obligations under this Lease arising from and after the Taking

Termination Date, (iii) no further Base Rent or Additional Rent shall accrue with respect to the period after the Taking Termination Date, (iv) Lessee shall have no repair, restoration or replacement obligations or payment obligations with respect thereto with respect to such Taking or damages caused thereby, (v) neither Lessor nor any other Lessor Indemnified Party shall have any claim against Lessee with respect to repair, restoration or replacement of the Leased Property or Losses arising from termination of this Lease or Lessee's purchase of the Leased Property pursuant to this <u>Section 16.2</u>, (vi) each Lessee Guarantor shall be released of liability and obligations under its Lessee Guaranty, Ultra Resources shall be released of liability and obligations under its Lessee Guaranty, and each Lessor Guarantor shall be released of liability and obligations under its Lessee Guaranty, and each Lessor Guarantor shall be released of liability and obligations under its Lessor Guaranty with respect to (x) as to Lessee Guarantors and Ultra Resources only, matters for which Lessee is released in <u>subsections (iv</u>) and (<u>v</u>) of this <u>Section 16.2</u>(d), preceding, and (y) matters arising or accruing from and after the Taking Termination Date, it being understood and agreed, however, that Lessee Guarantors, Ultra Resources and Lessor Guarantor shall remain liable with respect to matters or claims arising or accruing prior to the Taking Termination Date except those matters described in <u>subsections (iv</u>) and (<u>v</u>), and (<u>v</u>), and (<u>v</u>), and (<u>v</u>) all Taking Proceeds shall be paid to Lessee. Following a termination of this Lease under this <u>Section 16.2</u>, Lessee's right of first refusal under, and the other terms and provisions of this Lease which survive termination, shall remain in full force and effect.

(e) *Closing of Purchase Offer.* In addition, if Lessor accepts the Lessee Taking Termination Notice, on the Taking Termination Date (i) Lessee shall pay to Lessor the purchase price specified by Lessee in the Lessee Taking Termination Notice or such other purchase price as may be agreed by Lessor and Lessee, (ii) Lessor shall convey the Leased Property to Lessee or its designee on an **AS IS WHERE IS** basis, without any express or implied warranties from Lessor other than special warranties of title, provided, however, that such conveyance shall in any event be made by Lessor to Lessee or its designee free and clear of (A) all Permitted Lessor Liens, (B) any other Liens placed on the Leased Property by, through or under Lessor; and (C) any Other Recorded Documents executed by, through or under Lessor in violation of this Lease, and (iii) Lessor will deliver to Lessee or its designee such conveyance documents, each in form and substance reasonably satisfactory to Lessee, as Lessee may reasonably request, to transfer the ownership of the Leased Property to Lessee free and clear of the Liens and other items described in <u>subsections (A)-(C)</u> above

(f) *Continuation of Lease if Lessor Rejects the Lessee Taking Termination Notice*. If Lessor rejects or is deemed to have rejected the Lessee Taking Termination Notice, such Lessee Taking Termination Notice with respect to termination of the Lease and Lessee's offer to purchase the Leased Property shall be deemed to be automatically rescinded and this Lease shall remain in full force and effect in accordance with its terms

(g) *No Waiver.* Upon each occurrence of a Taking for which Lessee has the right under this <u>Section 16.2</u> to give a Lessee Taking Termination Notice, Lessee shall be entitled to give its Lessee Taking Termination Notice with respect to such Taking, even if Lessee previously elected not to give such notice with respect to the occurrence of a prior Taking and even if prior Lessee Taking Termination Notice(s) were rejected or deemed rejected by Lessor hereunder.

16.3 **Restoration Conditions**. Provided that this Lease has not been terminated pursuant to <u>Section 16.2</u>, above, in connection with a Taking, then Lessee shall repair, restore or replace the Leased Property damaged as a result of such Taking, as and to the extent required by <u>Section 16.2</u> and the Taking Proceeds shall be paid as follows:

(a) *Payment of Casualty Proceeds to Lessee*. All Takings Proceeds shall be paid to Lessee.

(b) *Repair Costs in Excess of \$2,000,000.* If the cost of repairs, restoration or replacement of the Taking exceeds \$2,000,000 with respect to such Taking, (A) Lessee shall provide to Lessor and Lessor

Lender (or the Agent for such Lessor Lender if requested in writing by Lessor) such reasonable documentation related to such repairs, restoration or replacement as Lessor and Lessor Lender (or the Agent for such Lessor Lender if requested in writing by Lessor) may reasonably request to confirm that such repairs, restoration or replacements have been made in the manner required by this Lease, (B) if such repairs, restoration or replacement have not been completed within ninety (90) days following the date of such Taking, Lessee shall pay to the Lessor Lender (or the Agent for such Lessor Lender if requested in writing by Lessor) the amount by which the total cost of such repairs, restoration or replacement exceeds \$2,000,000, reduced by repairs, restoration and replacement costs in excess of \$2,000,000 completed and paid as of such date, which amounts shall be held and disbursed by such Lessor Lender to Lessee as such repairs, restoration or replacements progress, and (C) Lessor and Lessor Lender (or the Agent for such Lessor Lender if requested in writing normal business hours and after prior reasonable written notice, with Lessee's representative present and upon the condition that Lessor and Lessor Lender (or the Agent for such Lessor Lender (or the Agent for such Lesse's rules and regulations, including safety, (I) to inspect the damage, and (II) to confirm that such repairs, restoration or replacements are being or have been made in the manner required by this Lease; provided that no such inspection shall interfere with the normal operation of the Leased Property or the business of Lessee or its Affiliates.

(c) *Payment of Taking Proceeds.* If not already paid to Lessor or Lessor's Lender as required by <u>Section 16.3(b)</u>, above, upon completion of the repairs, restoration and replacement of the Leased Property damaged as a result of a Taking, Lessee shall pay to Lessor Lender (or the Agent for such Lessor Lender if requested in writing by Lessor) the amount by which such Taking Proceeds exceed the cost of repairs, restoration and replacement of the Leased Property by Lessee.

ARTICLE XVII. LIMITS ON TRANSFERS, ASSIGNMENTS, LEASES AND LIENS

17.1 Assignment and Subletting by Lessee.

(a) *Lease Assignments.* At any time and from time-to-time, and without any consent from or approval by Lessor, Lessee may cause or permit to occur a Permitted Lease Assignment. For purposes hereof, the term "*Permitted Lease Assignment*" means either (i) a Lease Assignment for which (A) the successor Lessee is a Permitted Lessee Transferee, and (B) within ten (10) Business Days after the date of such Lease Assignment, Lessee provides to Lessor (I) a copy of the written assignment of lease or similar agreement, including an assumption by the successor Lessee of the obligations of Lessee under this Lease arising from and after the date of such Lease Assignment, (II) if such assignee successor Lessee qualifies as a Permitted Lessee Transferee because its Affiliate is an Investment Grade Person or otherwise qualifies as a Permitted Lessee Transferee because its Affiliate qualifies under <u>subsection (f)</u> of the definition of the term "Permitted Lessee Transferee", a Lease Guaranty executed by such Affiliate, and (III) evidence of the insurance required by this Lease, or (ii) any other Lease Assignment approved in writing by Lessor. Lessee shall not enter into any Lease Assignment other than a Permitted Lease Assignment shall be null and void *ab initio*.

(b) *Subleases.* At any time and from time-to-time, and without any consent from or approval by Lessor, Lessee may enter into a Permitted Sublease. For purposes hereof, the term "*Permitted Sublease*" means a Sublease which is either (i) a Sublease with a Permitted Lessee Transferee so long as within ten (10) Business Days after Lessee's entering into any such Sublease with a Permitted Lessee Transferee, Lessee shall provide Lessor with a copy of the executed sublease agreement, or (ii) any other Sublease approved in writing by Lessor. Lessee shall not enter into any Sublease other than a Permitted Sublease, and any Sublease other than a Permitted Sublease shall be null and void *ab initio*. Renewals of any Permitted Sublease shall be deemed to be a new Sublease for purposes of this<u>Section 17.1(b)</u> and shall not constitute a

Permitted Sublease unless such renewal otherwise qualifies as a Permitted Sublease at the time of such renewal.

(c) Lessor Consent to Certain Proposed Permitted Lessee Transferees

(i) <u>No Lessor Consent Required</u>. No Lessor consent shall be required with respect to any Permitted Lessee Transferee described in <u>subsections (a), (b), (c),</u> (d), or (e) of the definition of the term "Permitted Lessee Transferee".

(ii) <u>Limited Lessor Consent Rights for Investment Grade Persons</u>. So long as an Ultra Entity Person is the Lessee, Lessor's consent shall be required with respect to an Investment Grade Person described in <u>subsection (f)</u> of the definition of the term "Permitted Lessee Transferee," but Lessor's approval shall not be unreasonably withheld, and Lessor's approval shall be based solely on the following: (A) a review of such Investment Grade Person's experience in the oil and gas exploration and production business, and (B) the proposed use by such Investment Grade Person would not (I) constitute a breach of <u>Article IX</u>, or (II) in Lessor's reasonable judgment, impair the structural integrity, functionality or value of the Leased Property in any material respect.

(iii) <u>Scope of Additional Lessor Consent Required for Certain Other Permitted Lessee Transferees</u>. Lessor consent shall be required with respect to a proposed Person described in <u>subsection (g)</u> of the definition of the term "Permitted Lessee Transferee", but Lessor shall not unreasonably withhold its consent to any such proposed Permitted Lessee Transferee, and Lessor's approval shall be based solely on the following: (A) such Person's current and future projected financial strength and creditworthiness, management style, financial philosophy and reputation, (B) such Person's experience in the oil and gas exploration and production business, (C) such Person's ability to recover costs or pass them on to its customers and other Persons, and (D) the proposed use by such Person would not (I) constitute a breach of <u>Article IX</u>, or (II) in Lessor's reasonable judgment, impair the structural integrity, functionality or value of the Leased Property in any material respect.

(iv) <u>Approval Request Process</u>. In the event Lessor consent is required pursuant to <u>subsections</u> (f) or (g) of the Permitted Lessee Transferee definition, Lessor shall in any event either approve or disapprove a proposed Permitted Lessee Transferee for which Lessor consent is required in writing as soon as practicable but no later than fifteen (15) Business Days after receipt of Lessee's written notice to Lessor requesting approval of such proposed Permitted Lessee Transferee and such information with respect to such Permitted Lessee Transferee (consistent with the scope of Lessor's consent as provided herein) as may be reasonably requested by Lessor (by written notice given to Lessee within five (5) days after receipt of Lessee's aforesaid written notice) to evaluate the proposed transaction and the affected parties. Lessor shall be deemed to have rejected any such proposed Permitted Lessee Transferee if a written request for approval is delivered to Lessor, and Lessor does not respond in writing within fifteen (15) Business Days after Lessor's receipt of such written request for approval and all such reasonable information timely requested by Lessor as provided in the immediately preceding sentence. If Lessor does not consent in writing to such proposed Permitted Lessee Transferee within such within fifteen (15) Business Day period, Lessor agrees to provide Lessee with a reasonably detailed written explanation as to the reasons for withholding such consent to such proposed Permitted Lessee Transferee on or before the end of such fifteen (15) Business Day period.

(v) <u>No Expansion of Lessor Approval Rights to a Proposed Permitted Lessee Transferee</u>. This <u>subsection (c)</u> shall not in any event be deemed to give Lessor approval rights as to any proposed Permitted Lessee Transferee where such approval rights are not expressly given in the definition of such term, nor shall this <u>subsection (c)</u> be deemed to expand the scope or basis for any

such Lessor approval beyond the scope provided in the definition of the term "Permitted Lessee Transferee".

Lessor Consent to Lease Assignments and Subleases. Exclusive of and separate from Lessor consent rights with respect to a proposed Permitted Lessee (d) Transferee, which are governed by the definition of the term "Permitted Lessee Transferee" and subsection (c), above, Lessor shall not unreasonably withhold its consent for any Lease Assignment or for any Sublease as provided in Sections 17.1(a)(ii) or 17.1(b)(ii), above, and Lessor's approval of a Lease Assignment or Sublease to the extent required by Sections 17.1(a)(ii) or 17.1(b)(ii), above shall be based solely on the following: (i) the use contemplated under any such Lease Assignment or Sublease does not breach the provisions of Article IX and would not, in Lessor's reasonable judgment, impair the structural integrity, functionality or value of the Leased Property in any material respect, and (ii) the proposed sublessee or assignee enters into an agreement in a form reasonably satisfactory to Lessor recognizing that such Lease Assignment or Sublease, and any renewals or extensions thereof, are subordinate and subject to the terms of this Lease. Lessor shall approve or deny Lessee's request for consent to a Lease Assignment or a Sublease as and to the extent required by Sections 17.1(a)(ii) or 17.1(b)(ii), above, as soon as practicable but no later than fifteen (15) Business Days after receipt of Lessee's notice to Lessor requesting consent with such information as may be reasonably requested by Lessor (by notice given to Lessee within five (5) days after receipt of Lessee's aforesaid notice) to evaluate the proposed transaction and the proposed assignee or sublessee. Lessor shall be deemed to have rejected any such Lease Assignment or Sublease if a written request for approval is delivered to Lessor, and Lessor does not respond in writing within fifteen (15) Business Days after Lessor's receipt of such written request for approval and all information timely requested by Lessor as provided in the immediately preceding sentence. If Lessor does not consent to such proposed Lease Assignment or Sublease in writing within such within fifteen (15) Business Day period, Lessor agrees to provide Lessee with a reasonably detailed written explanation as to the reasons for withholding such consent to such Lease Assignment or Sublease on or before the end of such fifteen (15) Business Day period. This subsection (d) shall not in any event be deemed to give Lessor approval rights as to any Sublease or Lease Assignment except as expressly provided in Sections 17.1(a)(ii) and 17.1(b)(ii), above, nor shall this subsection (d) be deemed to expand the scope or basis for any such approval under such Sections 17.1(a)(ii) and 17.1(b)(ii) beyond the issues set forth in the first sentence of this subsection (d).

(e) *Effect of Lease Assignment or Sublease on Continuing Lessee Liability.* Except as set forth in <u>Section 17.4</u>, no Lease Assignment or Sublease shall relieve Lessee of any of its obligations under or with respect to this Lease, relieve Lessee Guarantor of any of its obligations under the Lessee Guaranty or release Ultra Resources of any of its obligations under the Resources Guaranty.

(f) *Prohibited Lease Assignments and Subleases Void.* Any purported Lease Assignment which is not a Permitted Lease Assignment and any purported Sublease which is not a Permitted Sublease shall be null and void null and void *ab initio*.

(g) Subsequent Lease Amendments with Prior Lessee Consent If (i) Lessor and any assignee or sublessee of Lessee's interest in this Lease modify or amend this Lease so as to increase the obligations of such prior Lessee, and (ii) such modification or amendment is made without a prior Lessee's consent, then the liability of such prior Lessee, its Lessee Guarantor and Ultra Resources shall not be increased, but shall continue as it existed prior to such modification or amendment.

17.2 No Restrictions on Indebtedness or Liens of Lessee or Lessee Guarantors Notwithstanding any provision to the contrary in this Lease or otherwise, and for the avoidance of doubt, there shall be no restrictions or limitations on the ability of Lessee, Lessee Guarantor or Ultra Resources to incur Indebtedness or to grant a Lien on all or any portion of Lessee's rights under this Lease or with respect to the Leased Property, or with respect to any of the Lessee Property.

17.3 **Cure Rights Upon Assignee Default**. Notwithstanding anything to the contrary set forth in<u>Article XXIII</u> of this Lease or otherwise, and except where a prior Lessee and its Lessee Guarantor and Ultra Resources have been released from liability under this Lease in accordance with <u>Section 17.4</u>, upon the occurrence of an event or circumstance which would, with the giving of notice or passage of time, or both, constitute a Lessee Event of Default, Lessor shall not declare a Lessee Event of Default or exercise any rights or remedies on account of such Lessee Event of Default unless and until Lessor gives notice of such default to such prior Lessee(s), their respective Lessee Guarantor(s), and Ultra Resources and the opportunity to cure such default, within the period of time permitted under <u>Section 23.1</u> of this Lease for curing such event or circumstance, but with such cure period as to such prior Lessee, such prior Lessee Guarantor and Ultra Resources beginning only upon receipt by such prior Lessee(s), its Lessee Guarantor(s) or Ultra Resources, as applicable, of such notice

17.4 **Release of Lessee and Lessee Guarantor.** From and after the date of a Permitted Lease Assignment to a Permitted Lessee Transferee of the type, described in <u>subsections (c). (e). (f)</u> or (g) of the definition thereof, then (a) the assigning Lessee and each prior Lessee shall be automatically released from any and all obligations and liabilities arising or accruing under this Lesse from and after the date of such Permitted Lease Assignment, (b) each Lessee Guarantor shall be automatically released from any and all obligations and liabilities arising or accruing under the tessee Guaranty with respect to obligations or liabilities arising or accruing under the Resources Guaranty with respect to obligations and liabilities arising or accruing from and after the date of such Permitted Lease Assignment, and (c) Ultra Resources shall be automatically released from any and all obligations or liabilities arising or accruing from and after the date of such Permitted Lease Assignment to a Permitted Lessee Transferee described in <u>subsection (c)</u> of the definition thereof, (i) if the Permitted Lessee Transferee is owned directly or indirectly by a Permitted Ultra Petroleum Successor, as a condition to such release Lessor may require the ultimate parent company of such Permitted Lessee Transferee to execute a Lessee Guaranty. Neither Lessee Guarantor nor Ultra Resources shall be released of its obligations or liabilities under this Lease, its Lessee Guaranty or the Resources Guaranty, as applicable, (A) in connection with any Lease Assignment except as provided in the immediately preceding sentence, or (B) in connection with any Sublease.

17.5 Transfers and Liens by Lessor and Equity Investors.

(a) **Prohibitions on Indebtedness, Transfers, Leases and Liens.**

(i) <u>Lessor Transfers</u>. Lessor shall not Transfer or permit the Transfer of all or any portion of the Lessor Interests except a Transfer of all the Lessor Interests pursuant to a Permitted Sale. Any Transfer of all or any portion of the Lessor Interests except a Transfer of all the Lessor Interests pursuant to a Permitted Sale shall be null and void *ab initio*

(i i) <u>Lessor Liens</u>. Lessor shall not grant or permit the granting of, or suffer to exist, a Lien upon all or any portion of the Lessor Interests except a Permitted Lessor Lien to secure Permitted Indebtedness. Any Lien upon all or any portion of the Lessor Interests other than a Permitted Lessor Lien to secure Permitted Indebtedness shall be null and void *ab initio*.

(iii) <u>Lessor Lease Transactions</u>. Lessor shall not at any time during the Lease Term enter into a Lessor Lease Transaction. Any Lessor Lease Transaction during the Lease Term shall be null and void *ab initio*.

(iv) <u>Lessor Indebtedness</u>. Lessor shall not incur Indebtedness other than Permitted Indebtedness. Any Indebtedness incurred by Lessor other than Permitted Indebtedness shall be null and void *ab initio*.

(v) <u>Equity Investor Transfers and Other Actions</u>. No Lessor Equity Interest Owner shall Transfer beneficial ownership of (including, for the avoidance of doubt, indirectly by allowing a change in the Beneficial Owner of) any portion of a Lessor Equity Interest to any Person that would result in possession of Controlling Lease Rights by any Person or Persons except pursuant to a Permitted Equity Transfer, and any such Transfer which is not a Permitted Equity Transfer shall be null and void *ab initio* and Lessor shall not recognize any such Transfer.

(vi) Equity Investor Liens. No Lessor Equity Interest Owner shall grant or permit the granting of, or suffer to exist, a Lien on any Lessor Equity Interest beneficially owned by such Lessor Equity Interest Owner except a Permitted Lessor Lien with respect to such Lessor Equity Interest. Any Lien on any Lessor Equity Interest other than a Permitted Lessor Lien with respect to any Lessor Equity Interest shall be null and void *ab initio* and Lessor shall not recognize any such Lien on any Lessor Equity Interest.

(b) *Permitted Sales; Release Upon a Permitted Sale.* For purposes hereof, the term "*Permitted Sale*" means a Transfer by Lessor which satisfies all of the following requirements in all respects:

(i) <u>Transfer of All Lessor Interests for Cash Consideration</u>. Such Transfer is a Transfer of all of Lessor's right, title and interest under this Lease and all of Lessor's right, title and interest in the Leased Property for an all-cash consideration (or cash paid to Lessor at closing and the remainder of the purchase price to be paid to Lessor by such transferee after such closing by deferred cash payments) to be paid to Lessor therefor and prior to such Transfer, Lessor has complied with <u>Section</u> <u>25.3</u> hereof with respect to such Transfer; an

(ii) <u>Notice</u>. Not less than fifteen (15) Business Days prior to the date of such Transfer (ten (10) Business Days in the case of a Transfer to an Affiliate of Lessor), Lessor has timely provided to Lessee notice of the proposed Transfer and evidence reasonably acceptable to Lessee that all the requirements with respect to a Permitted Sale shall be satisfied with respect to the proposed Transfer, including copies of the Governing Documents of such proposed transferee, the documents and instruments evidencing any Indebtedness or Liens arising from or created in connection with such Transfer, and information regarding the owners of the Equity Interests with respect to such proposed transferee and their Lessor Lenders; an

(iii) <u>Successor Lessor Requirements</u>. After giving effect to such Transfer, (A) the Lessor representations in <u>Section 4.1</u> shall be true and correct in all respects with respect to such transferee Lessor (other than the specific information regarding the state of organization and Equity Owners of Pinedale as set forth in <u>Section 4.1(a)</u>, which information shall be updated as applicable to describe the owners of the successor Lessor), (B) so long as an Ultra Entity Person is Lessee, no Person other than a Permitted Controller of Lease Rights shall possess Controlling Lease Rights, (C) no Disqualified Person shall possess Controlling Lease Rights, (D) Disqualified Persons, in the aggregate, do not possess beneficial ownership of twenty-five percent (25%) or more of the Lessor Equity Interests, and (E) such transferee Lessor and its Governing Documents comply in all respects with <u>Sections 27.1</u> and <u>27.2</u>; and

(iv) <u>Lessor Indebtedness</u>. As a result of such Transfer, there shall be no Lessor Indebtedness other than Permitted Indebtedness and there shall exist no Liens on any Lessor Interests other than Permitted Lessor Liens; and

(v) Equity Investor Agreement. In connection with such Transfer, the successor Lessor, each direct owner of a Lessor Equity Interest in such Lessor, and any Affiliate of any such Person required to guaranty the obligations of such Person under the Equity Investor Agreement as a result of the application of the definition of the term "Permitted Controller of Lease Rights" to such Transfer, must execute and deliver to Lessee the Equity Investor Agreement; and

(vi) <u>U.S. Transferee</u>. After giving effect to such Transfer, the transferee Lessor and its Lessor Guarantor, if applicable, shall be organized in the United States or if the transferee or its Lessor Guarantor, if applicable, is organized in a jurisdiction outside the United States, the transferee and its Lessor Guarantor, if applicable, shall indemnify Lessee for any taxes that may be imposed on Lessee as a result thereof, and

(vii) <u>Assumption of Lessor Lease Obligations</u>. Contemporaneous with such Transfer, the transferee Lessor must assume in writing the obligations of Lessor under this Lease arising from and after the date of such Transfer and the other obligations of Lessor under the Easements arising from and after the date of such Transfer; and

(viii) <u>Delivery of Lessor Guaranty and SNDA</u>. Contemporaneous with such Transfer (A) if otherwise required by this Lease in order for such Transfer to be a Permitted Sale, the Lessor Guarantor of such transferee Lessor must execute and deliver to Lessee a Lessor Guaranty, and (B) if applicable, such transferee Lessor and its Lessor Lenders must execute and deliver to Lessee the SNDA; and

(ix) <u>No Lessor Event of Default or Default Under Governing Documents</u> After giving effect to such Transfer, no event or circumstance shall have occurred at the time of or as a result of such Transfer which with the giving of notice or passage of time would constitute a Lessor Event of Default or a breach, default or failure of performance under the Governing Documents of the successor Lessor; and

(x) <u>Compliance with Laws and Lease</u>. Such Transfer complies with all Applicable Legal Requirements, including securities laws, and the documents evidencing such Transfer or other action are provided to Lessee to confirm, to the reasonable satisfaction of Lessee, compliance with the provisions of this Lease; and

(xii) <u>No Change in Lessee's Rights</u>. The documents evidencing such Transfer or other action do not alter the rights of Lessee under the Lease or as to the Leased Property; and

(xiii) <u>No Change in Lessee's Obligations</u>. After giving effect to such Transfer or other action, Lessee shall have no greater obligation or liability under this Lease as a result of such Transfer or other action, based on Applicable Legal Requirements in effect at the time of such Transfer or other action, than it would have had if the Transfer or such other action had not taken place (other than obligations indemnified by Lessor under this Lease).

Upon a Permitted Sale in compliance with the terms of this <u>Section 17.5</u>, the transferring Lessor and its Lessor Guarantor shall be released and relieved of further liability or obligation under this Lease and its applicable Lessor Guaranty, in each such case only with respect to matters arising or accruing from and after the date of such Permitted Sale.

(c) *Permitted Equity Transfers*. For purposes of this Agreement "*Permitted Equity Transfer*" means any Transfer of beneficial ownership of all or a portion of a Lessor Equity Interest which satisfies all of the following requirements in all respects:

(i) <u>Disqualified Persons and Controlling Parties</u> Such Transfer would not result in: (A) a Disqualified Person or Disqualified Persons possessing Controlling Lease Rights, (B) Disqualified Persons, in the aggregate, having beneficial ownership of 25% or more of the Lessor Equity netrests, and (C) for so long as an Ultra Entity Person is the Lessee under the Lease, a Person who is not a Permitted Controller of Lease Rights possessing Controlling Lease Rights; and

(ii) Notices. Such Lessor Equity Interest Owner shall have given written notice to Lessee of any proposed Transfer of beneficial ownership of all or any portion of a Lessor Equity Interest. Such notice shall be given to Lessee at least fifteen (15) Business Days (ten (10) Business Days in the case of a Transfer to an Affiliate of the transferor) prior to executing a definitive agreement for such Transfer. Such notice shall describe in detail the material terms of such proposed Transfer, the name of each Person proposed to acquire beneficial ownership of all or any portion of a Lessor Equity Interest or to possess Controlling Lease Rights, and provide reasonably sufficient information to enable Lessee to notify, within ten (10) Business Days of Lessee's receipt of such notice, whether such Person(s) is/are Disqualified Person(s) and to verify within such ten (10) Business Day period that such Transfer complies with the requirements of <u>Section 17.5(c)(i)</u>. Such Lesser Equity Interest Owner shall also provide to Lessee copies of all material documents evidencing such Transfer, in their then current form, at least ten (10) Business Days prior to the closing date or effective date of any Transfer and any material revisions or amendments to such documents promptly after they become available; and

(iii) Addendum to Equity Investor Agreement; Guaranty if Required. Each Person acquiring direct ownership of Lessor Equity Interests shall have executed an Addendum to the Equity Investor Agreement agreeing to be bound by the Equity Investor Agreement, and any Affiliate of any such Person required to guaranty the obligations of such Person under the Equity Investor Agreement as a result of the application of the definition of the term "Permitted Controller of Lease Rights" to such Transfer shall have executed such guaranty; and

(iv) <u>United States Persons; Lessee Indemnity</u>. Unless each Person acquiring beneficial ownership is a United States Person within the meaning of Section 7701(a)(3) of the Code, the transferee Lessor Equity Interest Owner and each Person acquiring beneficial ownership shall have agreed to indemnify Lessee, in a manner reasonably acceptable to Lessee, against any amounts payable by Lessee, as Lessee, pursuant to this Lease or otherwise to the extent such liabilities arise out of the status of such transferee as a Person other than a United States Person; and

(v) <u>Compliance with Laws and Lease</u>. Such Transfer complies with all Applicable Legal Requirements, including securities laws, and the documents evidencing such Transfer are provided to Lessee to confirm, to the reasonable satisfaction of Lessee, compliance with the provisions of this Lease; and

(vi) <u>No Default Under Lessor's Governing Documents</u>. No default or event of default attributable to such Lessor Equity Interest Owner or Lessor shall occur under Lessor's Governing Documents as a result of such Transfer; and

(vii) <u>No Change in Lessee's Rights</u>. The documents evidencing such Transfer do not alter the rights of Lessee under the Lease or as to the Leased Property; and

(viii) <u>No Change in Lessee's Obligations</u>. Immediately after giving effect to such Transfer, Lessee, as Lessee, shall have no greater obligation or liability under this Lease as a result of such Transfer, based on Applicable Legal Requirements in effect at the time of such Transfer, than it would have had if the Transfer had not taken place (other than obligations indemnified by

Lessor, the transferee Lessor Equity Interest Owner and such Person acquiring such beneficial ownership pursuant to subsection (iv), above). No Lessor or Lessor Guarantor shall be released or relieved of any liability or obligation under this Lease or its applicable Lessor Guaranty in connection with or as a result of a Transfer of any Lessor Equity Interest or any beneficial ownership of any Lessor Equity Interest.

ARTICLE XVIII. LESSEE'S SURRENDER OF LEASED PROPERTY

18.1 **Surrender**. Upon the expiration of the Term or the earlier termination of this Lease, Lessee shall surrender to Lessor in Good Condition and Repair the Leased Property and any Permitted Capital Improvements and Additional Lines that become Lessor's property upon termination of this Lease as provided in <u>Section 10.1(e)</u> hereof, in all such cases subject to reasonable and ordinary wear and tear, and subject to the provisions of <u>Article XV</u> and <u>Article XVI</u> and <u>Section 25.5</u>. Any removal of all or any of the Lessee Property by Lessee shall be done in material compliance with Environmental Laws and other Applicable Legal Requirements, including Lessee's obligation to file all reports and documentation required to be filed with any Governmental Authority in connection with the removal of Lessee Property, and in accordance with the requirements of <u>Article X</u> hereof. Lessee shall repair any material damage to the Leased Property caused by the removal of Lessee Property. After the expiration of the Term or the earlier termination of this Lease, neither Lessee nor Lessor shall have any rights, liabilities or obligations hereunder (except for any that survive the expiration or any earlier termination of this Lease) and including, without limitation, the terms and provisions of <u>Sections 25.2, 25.3</u> and <u>25.4</u> if and to the extent applicable.

ARTICLE XIX. ASSIGNMENT OF LEASE

19.1 **Assignment of Lease**. Lessor shall not (a) Transfer all or any portion of its right, title or interest under this Lease except pursuant to a Permitted Sale, or (b) create or suffer to exist a Lien on all or any portion of its right, title or interest under this Lease except (i) a Permitted Lessor Lien, (ii) a Lien which Lessee is contesting in accordance with a Permitted Lessor Contest.

ARTICLE XX.

MORTGAGE SUBORDINATION AND NON-DISTURBANCE AND NOTICE TO MORTGAGEE

20.1 Mortgage Subordination. Except as and to the extent applicable with respect to the Underlying Lien Claims, this Lease and the right, title and interest of Lessee under this Lease shall be prior and superior in all respects to any Lien on all or any portion of the Leased Property by, through or under Lessor, irrespective of the time of execution or time of recording of such Lien, whether placed on the Leased Property prior to, on or after the Effective Date, and whether or not such Lien constitutes a Permitted Lessor Lien or secures Permitted Indebtedness. Except as and to the extent applicable with respect to the Underlying Lien Claims, no Lien placed or permitted on the Leased Property by, through or under Lessor shall be superior or prior to this Lease or Lessee's right, title and interest under this Lease, whether by date of execution, time of recording, tacking, subrogation, or any other legal or equitable principles or operation of law. Concurrently with the execution of this Lease, Lessor Lessor Lessor Lender and Lessee have entered into the SNDA. Lessor shall cause any subsequent Person holding or claiming any Lien in or to all or any portion of the Lessor Interests or Lessor Equity Interests, as a condition to the granting of same, to execute and deliver to Lessee the SNDA.

ARTICLE XXI. INDEMNIFICATION

21.1 Indemnification by Lessee.

(a) Lessee's Indemnification Obligations. Subject to the waiver of subrogation provisions in Section 14.4, the remainder of this Article XXI, and Section 26.3, and except for Losses for which Lessor is responsible pursuant to Section 21.2. Lessee, to the fullest extent permissible by Applicable Legal Requirements, agrees to indemnify, hold harmless and defend Lessor Indemnified Parties and Lessor Lender (in Lessor Lender's capacity as such) from and against any and all Losses proximately caused by (i) a Lease Assignment or Sublease made by Lessee in violation of this Lease, (ii) a Lessee Event of Default, (iii) any negligent acts or omissions of Lessee or any other Lessee Party with respect to this Lease, the Leased Property or the SNDA, (iv) the fraud, gross negligence or willful misconduct of Lessee or any Lessee Party in connection with the Leased Property or the transactions contemplated by this Lease or in connection with the SNDA, (v) the operation, possession, use, non-use, maintenance, modification, alteration, construction, restoration, condition, design or replacement of the Leased Property or any other Lessee Party, or (vi) the business and activities of Lessee or of any other Person permitted on or about the Leased Property by Lessee (whether as an invitee, sublessee, licensee or otherwise); provided, however, to the extent and in the proportion such Losses also arise out of or are based upon or arise from or are attributable to any of the following (collectively, and together with the matters described in Section 21.1(b), below, the "Exclusions from Lessee's Indemnification Obligations"), Lessee's indemnification under this subsection shall, to such extent, not apply:

(A) the negligence, willful misconduct or fraud of any Lessor Indemnified Party or, in the case of indemnification of a Lessor Lender, the negligence, willful misconduct or fraud of such Lessor Lender;

(B) except to the extent arising by reason of a Lessee Event of Default, breach of any covenant, representation or warranty by any Lessor Party, Lessor Indemnified Party or Lessor Lender contained in this Lease, any Lessor Loan Document, the SNDA, the Equity Investor Agreement or any other document entered into in connection herewith or therewith;

(C) any dispute (A) between or among any Lessor Indemnified Parties, (B) between or among Lessor Lenders or (C) between or among any one or more Lessor Indemnified Parties, on the one hand, and any Lessee Party or Lessor Lender on the other hand;

(D) the negotiation, preparation or administration of this Lease, any Lessor Loan Document, the SNDA, the Equity Investor Agreement or any other document entered into in connection herewith or therewith;

(E) except to the extent arising by reason of a Lessee Event of Default, a violation of any Applicable Legal Requirement or any other legal, regulatory, judicial or similar requirement by any Lessor Party, Lessor Indemnified Party or Lessor Lender;

(F) a Transfer or other disposition by any Lessor Party, Lessor Indemnified Party or Lessor Lender of any interest in all or any portion of the Leased Property;

(G) to the extent imposed with respect to any period (except during the exercise of remedies pursuant to this Lease) after the expiration of the Term or earlier termination of this Lease, except to the extent such Losses relate to (A) events or matters occurring prior to the expiration of the Term or earlier termination of the Lease, (B) the exercise of remedies relating to a Lessee Event of Default, or (C) the failure by Lessee to return the Liquids Gathering System to Lessor in accordance with the terms of this Lease;

(H) arising as a result of a default or event of default under the Lessor Loan Documents, the SNDA or the Equity Investor Agreement or arising otherwise in connection with or under any Lessor Loan Document, except to the extent caused by a Lessee Event of Default; or

(I) a Lessor Event of Default.

(b) Additional Limits on Lessee's Indemnification Obligation. Notwithstanding the foregoing, Losses arising from Lessee's indemnification obligations in this Section 21.1 shall not include (i) any taxes or costs or expenses associated with contesting taxes, it being understood that Lessee's liability with respect to taxes is governed by Article VI, (ii) principal, interest, default interest, fees, charges or penalties owed in connection with Indebtedness of any Lessor Party or any Lessor Indemnified Party, provided, however, that default interest on Permitted Indebtedness imposed as a result of a Lessor event of default thereunder (A) other than a default based on the failure to make a payment or payments under or with respect to such Indebtedness and (B) which arises solely and directly as a result of a Lessee Event of Default other than a failure to pay Base Rent or Additional Rent, shall not be excluded from "Losses", or (iii) Losses related to any funding or hedging arrangements entered into or obtained in connection with Lessor Indebtedness or debt or equity investments in Lessor.

(c) *Environmental Matters Not Included.* For the avoidance of doubt, this <u>Section 21.1</u> is not intended to provide any indemnification to Lessor or any Lessor Indemnified Party for any matters related to Environmental Laws or Hazardous Materials or any other matters covered by <u>Article XXII</u>.

2 1 . 2 **Release and Indemnification by Lessor**. Subject to the waiver of subrogation provisions in<u>Section 14.4</u>, and except for Losses for which Lessee is responsible pursuant to <u>Section 21.1</u>, and subject to the provisions in<u>Article XIV</u> and <u>Section 26.3</u>, Lessor, to the fullest extent permissible by Applicable Legal Requirements:

(a) **Release of Lessee Indemnified Parties.** Releases Lessee and each other Lessee Indemnified Party from any Losses from any source arising out of or based upon, in whole or in part, (i) any acts or omissions by any Lessor Party, Lessor Indemnified Party or Lessor Lender in violation of the terms of this Lease, the SNDA or the Equity Investor Agreement, (ii) any acts or omissions of any Lessor Party, Lessor Indemnified Party or Lessor Lender with respect to this Lease, the Leased Property, the SNDA or the Equity Investor Agreement, (iii) any Lessor Event of Default, or (iv) the fraud, negligence or willful misconduct of any Lessor Party, Lessor Indemnified Party or Lessor Lender in connection with the Leased Property or the transactions contemplated by this Lease, or in connection with the Equity Investor Agreement or the SNDA; and

(b) Lessor Indemnification of Lessee and Lessee Indemnified Parties. Agrees to indemnify, hold harmless and defend Lessee and each other Lessee Indemnified Party from and against any and all Losses, from any source arising out of or based upon, in whole or in part, (i) any acts or omissions by any Lessor Party or Lessor Indemnified Party in violation of the terms of this Lease, the SNDA or the Equity Investor Agreement, (ii) any acts or omissions of any Lessor Party or Lessor Indemnified Party with respect to this Lease or the Leased Property, the SNDA or the Equity Investor Agreement, (iii) any Lessor Event of Default, or (iv) the fraud, negligence or willful misconduct of any Lessor Party or Lessor Indemnified Party in connection with the Leased Property or the transactions contemplated by this Lease, or in connection with the Equity Investor Agreement or the SNDA;

Provided, however, that in the case of <u>subsections (a)</u> and (b) above, to the extent and in the proportion such Losses also arise out of or are based upon (A) any breach of this Lease by Lessee, (B) the fraud, negligence or willful misconduct of Lessee or any other Lessee Party with respect to the Leased Property, this Lease, the SNDA or the Equity Investor Agreement, (C) Taxes for which Lessee is responsible pursuant to this Lease, or (D) breach by Lessee of any representations made by it in this Lease, then Lessor's release and indemnification under this <u>subsection (b)</u> shall not apply.

2 1 . 3 **Concurrent Negligence**. Notwithstanding the provisions of <u>Section 21.1</u> and <u>Section 21.2</u>, in the event of the concurrent negligence or intentional misconduct of Lessee, any other Lessee Party or any Lessee Indemnified Party on the one hand and the concurrent negligence or intentional misconduct of Lessor, any other Lessor Party or Lessor Indemnified Party, on the other hand, a party's (the "*Indemnifying Party*") obligation to indemnify the other as set forth in this<u>Article XXI</u> shall be limited to the extent of the Indemnifying Party's negligence and/or intentional misconduct (and that of the Lessee Indemnified Parties (if Lessor is the Indemnifying Party)), including the Indemnifying Party's proportionate share of reasonable costs, attorneys' fees, and expenses incurred in connection with any claim, action, or proceeding brought with respect to such injury or damage.

21.4 Survival. The obligations of Lessee and Lessor under this Article XXI shall survive the expiration or earlier termination of this Lease.

21.5 **Claims Procedure.** In the case of any Losses asserted by an Indemnified Party under this <u>Article XXI</u>, such Indemnified Party shall give prompt notice thereof to the Indemnifying Party and in any event within no less than 60 days after the Indemnified Party receives notice of such assertion; <u>provided</u> that failure to so notify the Indemnifying Party shall not reduce the Indemnifying Party's obligations to indemnify any Indemnified Party hereunder except to the extent such failure adversely affects the Indemnifying Party's rights, or materially compromises such Indemnifying Party's ability, to defend such Losses or results in additional liability on such Indemnifying Party's part. The Indemnifying Party shall be entitled, at its expense, acting through counsel selected by it (and reasonably satisfactory to such Indemnified Party), to participate in or to assume and control (if it promptly so elects upon notice of the Losses), the negotiation, litigation and/or settlement of any such Losses. Such Indemnified Party may (but shall not be obligated to) participate at its own expense and with its own counsel in any proceeding conducted by the Indemnifying Party in accordance with the foregoing, in which case the Indemnifying Party shall keep such Indemnified Party and its counsel fully informed of all proceedings and filings. Notwithstanding the foregoing, but subject to <u>Article XI</u> and the right of Lesser to pursue Tax Challenges and Permitted Lessee Contests and the right of Lessor to pursue Permitted Lessor Contests, the Indemnifying Party shall not be indemnifying Party, a Level I Lessor Default has occurred and is continuing, (b) the proceeding involves possible imposition of any criminal liability or unindemnified evil penalty on the Indemnified Party, or (c) the proceeding involves the granting of injunctive relief against the Indemnified Party not related to this Lease.

ARTICLE XXII. ENVIRONMENTAL LAWS

22.1 **Environmental Undertakings.** Lessee's operations on the Leased Property from and after the Effective Date shall comply with Environmental Laws in all material respects. Lessor will conduct no operations on the Leased Property. After either party discovers or is informed of the existence of a material violation of Environmental Laws with respect to the Leased Property, or receipt of any notices alleging non-ordinary course of business liabilities under Environmental Laws with respect to the Leased Property, that party shall give prompt notice to the other party of such event.

22.2 Environmental Covenants.

(a) Lessee's Obligations for Certain Corrective Actions. Except as otherwise provided in Section 25.5, Lessee shall be responsible for performing Corrective Action required under Environmental Law to the extent arising from Lessee Environmental Liabilities, such Corrective Actions to be performed in material compliance with Environmental Laws. Lessee's obligations to conduct Corrective Action shall be limited to those activities designed to achieve the least stringent remediation permitted under Environmental Laws for a property in similar use as the property requiring Corrective Action.

(b) Lessor's Obligations for Certain Corrective Actions. Upon Lessee's written request to Lessor requesting that Lessor take such Corrective Action, Lessor shall be responsible for performing any Corrective Action required under Environmental Laws to the extent arising from Lessor Environmental Liabilities, such Corrective Actions to be performed in material compliance with Environmental Laws. Lessor's obligations to conduct Corrective Action shall be limited to those activities designed to achieve the least stringent remediation permitted under Environmental Laws for a property in similar use as the property requiring Corrective Action.

22.3 **Lessee Environmental Indemnity.** Lessee shall, at no cost to Lessor, protect, defend, indemnify, release and hold harmless each of the Lessor Indemnified Parties and each Lessor Lender for, from and against any and all Losses (including engineers' fees, environmental consultants' fees, and costs of investigation (including sampling, testing, and analysis of soil, water, air, building materials and other materials and substances whether solid, liquid or gas)) imposed upon or incurred by or asserted against any Lessor Indemnified Parties, but only to the extent arising out of any one or more of the following:

- (a) Lessee Environmental Liabilities; or
- (b) Lessee breach of its covenants in this Article XXII (including Lessee's obligation to take Corrective Action under Section 22.2(a)).

Lessee's indemnification obligation does not cover any Losses to the extent attributable to (i) a breach of this Lease by Lessor or any other Lessor Party, (ii) the fraud, negligence or willful misconduct of Lessor, any other Lessor Party or any Lessor Lender, (iii) Lessor Environmental Liabilities, or (iv) Exclusions from Lessee's Indemnity Obligations. Lessor's and Lessor Indemnified Parties' sole and exclusive remedy, either at law, under statute or in equity, against Lessee for Lessee Environmental Liabilities, violations of Environmental Law by Lessee or any other Lessee Party or Releases of Hazardous Materials by Lessee or any other Lessee Party are the remedies in this <u>Section 22.3</u> and no other rights or remedies shall be applicable thereto, except that Lessee's failure to perform the Corrective Actions required by Lessee under<u>Section 22.2(a)</u> shall, if not cured within the notice and cure periods provided in <u>Section 23.1(f)</u>, or such earlier date as such Corrective Action must be completed to comply with applicable Environmental Law (but subject to any pending Permitted Lessee Contests with respect to same), entitle Lessor to exercise its self-help remedies under <u>Section 23.2</u> for so long as such Corrective Action remains incomplete.

22.4 **Lessor Environmental Indemnity.** Lessor shall, at no cost to Lessee, protect, defend, indemnify, release and hold harmless each of the Lessee Indemnified Parties for, from and against any and all Losses (including engineers' fees, environmental consultants' fees, and costs of investigation (including sampling, testing, and analysis of soil, water, air, building materials and other materials and substances whether solid, liquid or gas)) imposed upon or incurred by or asserted against any Lessee Indemnified Party, but only to the extent arising out of any one or more of the following:

- (a) Lessor Environmental Liabilities; or
- (b) Lessor's breach of its covenants in this Article XXII (including Lessor's obligation to take Corrective Action under Section 22.2(b)).

Lessor's indemnification obligation does not cover any Losses to the extent attributable to (i) a breach of this Lease by Lessee or any other Lessee Party, (ii) the fraud, negligence or willful misconduct of Lessee or any other Lessee Party, or (iii) Lessee Environmental Liabilities. Lessee's and Lessee Indemnified Parties' sole and exclusive remedy, either at law, under statute or in equity, against Lessor for Lessor's Environmental Liabilities, violations of Environmental Law by Lessor or any other Lessor Party or Releases of Hazardous Materials by Lessor or any other Lessor Party are the remedies in this <u>Section 22.4</u>, except that Lessor's failure to perform the Corrective Actions required by Lessor under <u>Section 22.2(b)</u> shall, if not cured within

the notice and cure periods provided in <u>Section 23.4(i)</u> or such earlier date as such Corrective Action must be completed to comply with applicable Environmental Law (but subject to any pending Permitted Lessor Contests with respect to same), entitle Lessee to exercise its self-help remedies under <u>Section 23.4</u> for so long as such Corrective Action remains incomplete.

22.5 **Claims Procedure**. In the case of any Losses asserted by an Indemnified Party under this Article XXII, such Indemnified Party shall give prompt notice thereof to the Indemnifying Party and in any event within no less than 60 days after the Indemnified Party receives notice of such assertion; provided that failure to so notify the Indemnifying Party shall not reduce the Indemnifying Party's obligations to indemnify any Indemnified Party hereunder except to the extent such failure adversely affects the Indemnifying Party's rights, or materially compromises such Indemnifying Party's ability, to defend such Losses or results in additional liability on such Indemnifying Party's part. The Indemnifying Party shall be entitled, at its expense, acting through counsel selected by it (and reasonably satisfactory to such Indemnified Party may (but shall not be obligated to) participate at its own expense and with its own counsel in any proceeding conducted by the Indemnifying Party in accordance with the foregoing, in which case the Indemnifying Party shall keep such Indemnified Party and its counsel fully informed of all proceedings and filings. Notwithstanding the foregoing, but subject to <u>Article XI</u> and Lessee's right to pursue Tax Challenges and Permitted Lessee Contests and Lessor's right to pursue Permitted Lessor Contests, the Indemnifying Party shall not be entitled at is continuing, (b) the proceeding involves possible imposition of any criminal liability or unindemnified civil penalty or the Indemnifying Party, or (c) the proceeding involves the granting of injunctive relief against the Indemnified Party not related to this Lesse.

2 2 . 6 **Survival**. It is expressly understood and agreed that Lessee's and Lessor's obligations under this <u>Article XXII</u> shall survive the expiration or earlier termination of this Lease.

ARTICLE XXIII. DEFAULTS AND REMEDIES

23.1 **Lessee Events of Default**. Each of the following shall be an event of default under this Lease (each, subject to the additional notice and cure provisions of <u>Section 17.3</u>, a "Lessee Event of Default"):

(a) Lessee fails to make any payment of Base Rent when due and such failure continues for five (5) Business Days after Lessee's receipt of written notice from Lessor of such failure; or

(b) Lessee fails to make any Additional Rent (other than a failure to pay Base Rent and except for such payments as Lessee is contesting in good faith or pursuant to a Tax Challenge or a Permitted Lessee Contest) and such failure continues for fifteen (15) Business Days after Lessee's receipt of written notice of such failure from Lessor; or

(c) any representation and warranty made by Lessee under this Lease is false at the time made and which, individually or in the aggregate with respect to each other such false representation or warranty, is material; or

(d) Lessee makes or permits a Lease Assignment or a Sublease in violation of <u>Article XVII</u> of this Lease, and with respect to a Lease Assignment or Sublease to or with an Affiliate of Lessee, Lessee fails to unwind or terminate such Lease Assignment or Sublease within sixty (60) days after Lessee's receipt of written notice from Lessor of such breach; or

(e) Lessee fails to maintain property insurance on the Leased Property in material compliance with this Lease, and such failure continues for fifteen (15) Business Days after Lessee's receipt of written notice of such failure from Lessor; or

(f) Lessee fails to observe or perform any other covenant of Lessee under this Lease and that failure continues for sixty (60) days after Lessee's receipt of written notice of that breach from Lessor (or if the cure of that failure reasonably requires more than sixty (60) days to complete, if Lessee fails to commence the cure within such sixty (60) day period and thereafter diligently pursue such cure to completion), but with respect to the foregoing, no such uncured failure shall constitute a Lessee Event of Default unless individually or in the aggregate with each other such uncured failure, such failure is material; or

(g) the filing by or against Lessee or Lessee Guarantor of a petition for relief under any Debtor Relief Laws (unless, in the case of a petition filed against Lessee or Lessee Guarantor, the same is dismissed within ninety (90) days after filing), or the appointment of a trustee or receiver to take possession of all, or substantially all, of Lessee's or Lessee Guarantor's assets or of Lessee's interest in this Lease, where such appointment is not discharged in ninety (90) days after appointment of said trustee or receiver, or the voluntary filing of a petition for the appointment of the same; or

(h) Lessee or Lessee Guarantor (i) fails to make any payment of principal or interest with respect to any Material Debt after giving effect to any applicable cure period or (ii) fails to observe or perform any other agreement contained in any agreement or instrument relating to that Material Debt that is a default (other than a failure to pay specified in subsection (i) of this paragraph) and such default continues after the applicable grace or cure period, if any, specified in such agreement or instrument, if the effect of the failure specified in <u>subsection (i) or (ii)</u> is to accelerate the maturity of that Material Debt; or

(i) a breach or default has occurred and is continuing by Lessee under the SNDA and such breach or default continues for thirty (30) days after Lessee's receipt of written notice of such breach or default from Lessor.

23.2 Lessor's Remedies for a Lessee Event of Default. Upon the occurrence and during the continuance of a Lessee Event of Default, with or without notice or demand, except such notice as may be required by statute and cannot be waived by Lessee and such notice as is specifically required by the terms of this Lease (all other notices being hereby waived), Lessor shall be entitled to exercise, at its option, the following remedies, Lessor hereby waiving all other rights and remedies to which Lessor may be entitled at law or in equity:

(a) *Lessor's Remedies for a Level 1 Lessee Default.* Following the occurrence and during the continuance of a Level 1 Lessee Default only, Lessor shall have, with respect to each such Level 1 Lessee Default either, but not both, of the rights and remedies specified in <u>subsection (i)</u> and <u>subsection (ii)</u>, below.

(i) Option to Terminate the Lease. Upon the occurrence and during the continuance of a Level 1 Lessee Default, Lessor may give Lessee written notice of Lessor's intention to terminate this Lease as a result of such Level 1 Lessee Default (except for the terms and provisions hereof which survive termination) on a date specified in such notice (which date shall be no sooner than sixty (60) days after the date of the notice), in which case, upon the date therein specified, the Term and the estate hereby granted shall expire and terminate as if such date were the date hereinabove fixed for the expiration of the Term except the terms and provisions hereof which survive the Lease Term. Upon such termination, Lessee shall surrender possession of the Leased Property to Lessor and Lessor may take possession of the same on the termination date without being deemed guilty in any manner of trespass or becoming liable for any loss or damage resulting therefrom, without resort to legal or judicial process, procedure or action. If Lessor so

terminates this Lease, Lessor may, as full and complete damages for such Level 1 Lessee Default, bring an action against Lessee for any or all of the following:

(A) <u>Accrued and Unpaid Base Rent</u>. All Base Rent accrued and unpaid to the termination date and not otherwise paid by Lessee; and

(B) <u>Accrued and Unpaid Additional Rent</u>. All Additional Rent accrued and unpaid to the termination date and not otherwise paid by Lessee (including Lessee's indemnity obligations under <u>Sections 21.2</u> and <u>22.3</u> with respect to matters or circumstances arising prior to the date of Lease termination, whether or not Lessor has asserted such indemnity claims prior to the date of Lease termination), but without duplication for amounts for which Lessor seeks damages under this Lease; and

(C) <u>Failure to Surrender Leased Property</u>. Losses resulting from Lessee's failure to surrender the Leased Property as required by <u>Section 18.1</u>, but without duplication for amounts for which Lessor seeks damages or indemnification under this Lease; and

(D) <u>Collection Costs</u>. All reasonable out-of-pocket expenses incurred by Lessor in enforcing its remedies under this <u>Section 23.2(a)(i)</u> with respect to such Level 1 Lessee Default, including reasonable attorneys' fees, court costs, expert witness fees, costs of tests and analyses, travel and accommodation expenses, deposition and trial transcripts, copies and other similar costs and fees, paid or incurred by Lessor as a result of its exercise of remedies under this <u>subsection (a)(i)</u> with respect to such Level 1 Lessee Default, regardless of whether or not legal proceedings are actually commenced, subject to <u>Section 26.9 of</u> this Lease, and

(E) <u>Termination Damages</u>. The amount, if any, by which (1) the unpaid Base Rent (as calculated below) payable under this Lease which would have been earned after termination for the balance of the Term (had this Lease not been terminated and without giving effect to any unexercised extension options), discounted to present value at the Discount Rate exceeds (2) the Fair Market Rent for such period, also discounted to present value at the Discount Rate. For purposes of calculating Base Rent which would have been earned after termination for the balance of the Term under (1) above, the Variable Rent component of Total Rents shall be calculated based on the average Variable Rent payable during the one year period prior to the termination of this Lease.

Notwithstanding the foregoing or anything to the contrary set forth in this Lease, the amount required to be paid by Lessee pursuant ts $(\underline{Sections 23.2(a)(i)(A), (B), (C), (D))})$ and (\underline{E}) shall be limited as follows: The Inception Date Present Value of the amount paid by Lessee pursuant to $\underline{Sections 23.2(a)(i)(A), (B), (C), (D)})$ and (\underline{E}) plus the Inception Date Present Value of the Aggregate Minimum Rent shall not in any event exceed 200,250,000. As used herein, (I) "*Aggregate Minimum Rent*" means the Minimum Rent paid by Lessee during the period beginning on Effective Date and continuing through the date of termination of the Lease Term, (II) "*Inception Date Present Value*" of an amount means the present value of such amount as of the Effective Date, calculated using the Specified Discount Rate, and (III) "*Specified Discount Rate*" means eight percent (8%).

In addition, and notwithstanding the foregoing, in the event that as of the Lease termination date specified in Lessor's notice to Lessee such Level 1 Lessee Default has been cured, such termination notice shall be deemed to have been rescinded, this Lease shall remain in effect and Lessor shall only be entitled to Base Rent and Additional Rent which has accrued under the Lease under <u>subsections (A)</u> and (<u>B</u>), above, and Lessor's reasonable out-of-pocket expenses under <u>subsection (D)</u> above.

(i i) Option to Exercise Remedies Other Than Lease Termination Upon a Level 1 Lessee Default. In the event that Lessor elects NOT to exercise its remedies under subsection (i) above with

respect to a Level 1 Lessee Default, and in lieu of its remedies under<u>subsection (i)</u>, above, upon the occurrence and during the continuance of a Level 1 Lessee Default, Lessor may exercise any or all of its rights or remedies under <u>subsection (b)</u>, below, with respect to such Level 1 Lessee Default.

(b) Lessor's Remedies for a Lessee Event of Default Other Than Termination Available for a Level 1 Lessee Default Except as otherwise provided in Section 23.2(a) above with respect to the

exclusivity of such remedies, following the occurrence and during the continuance of any Lessee Event of Default, but only so long as either (x) such Lessee Event of Default does not constitute a Level 1 Lessee Default, or (y) such Lessee Event of Default does constitute a Level 1 Lessee Default and Lessor has elected not to pursue its termination rights as set forth in <u>subsection (a)(i)</u>, above, this Lease shall remain in full force and effect, Lessee shall be entitled to remain in possession of the Leased Property hereunder and shall remain obligated and liable for Lessee's obligations hereunder, and Lessor may exercise the following remedies, concurrently, successively, or in any combination (except NOT in combination with any of the remedies set forth in <u>subsection (a)(i)</u>, above, and excluding any other or additional remedies):

(i) <u>Accrued, Unpaid Base Rent</u> Lessor may bring an action against Lessee for all Base Rent accrued, unpaid and due and owing to the date of such action and not otherwise paid by Lessee; and

(ii) <u>Accrued, Unpaid Additional Rent</u>. Lessor may bring an action against Lessee for all Additional Rent accrued, unpaid and due and owing to the date of such action and not otherwise paid by Lessee (including Lessee's indemnity obligations under <u>Sections 21.2</u> and <u>22.3</u>), but in any case without duplication for claims made under <u>subsection (iii)</u> of this <u>Section 23.2(b)</u>, below; and

(iii) <u>Damages</u>. Lessor may bring an action against Lessee for Lessor's damages which are proximately caused by such Lessee Event of Default, but without duplication for amounts for which Lessee seeks damages or indemnification under this Lease; and

(iv) <u>Specific Performance</u>. Lessor may bring an action against Lessee for specific performance by Lessee of its unperformed obligations under this Lease, but without duplication for other amounts for which Lessor seeks damage or indemnification under this Lease. All covenants and agreements of Lessee in this Lease shall be deemed special, unique and extraordinary, and any breach of any covenant or agreement by Lessee shall be deemed to cause Lessor irreparable injury not properly compensable by damages in an action at law, such that the rights and remedies of Lessor may be enforced both at law or in equity

(v) <u>Self-Help Remedies</u>. Lessor may perform, on Lessee's behalf, any unperformed covenant or obligation under this Lease constituting such Lessee Event of Default if such covenant or obligation remains unperformed on the date that is not less than ten (10) Business Days after entry of a final and non-appealable court order confirming such Lessee Event of Default, in which event, Lessee shall reimburse Lessor for all reasonable costs, expenses and disbursements incurred by Lessor in doing so, plus together with interest thereon at the Default Interest Rate from the Due Date for Additional Rent, but in any event without duplication for amounts for which Lessor seeks damages or indemnification under this Lease; provided, however, that any action taken by Lessor in accordance with this <u>Section 23.2(b)(v)</u> shall be made in compliance with Lessee's rules and regulations with respect to the Leased Property; and

(vi) <u>Costs of Collection</u>. Lessor shall be entitled to recover from Lessee all out-of-pocket costs and expenses, including reasonable attorneys' fees, court costs, expert witness fees, costs of tests and analyses, travel and accommodation expenses, deposition and trial transcripts, copies and other similar costs and fees, paid or incurred by Lessor in exercising its remedies under

this Section 23.2(b) as a result of such Lessee Event of Default, regardless of whether or not legal proceedings are actually commenced, subject to Section 26.9 of this Lease.

(c) No Waiver or Election of Remedies. No delay or omission of Lessor to exercise any right or power accruing upon the occurrence and during the continuance of any Lessee Event of Default shall impair any other or subsequent Lessee Event of Default or impair any rights or remedies consequent thereto unless such Lessee Event of Default is cured. Except with respect to the exclusivity of certain remedies as provided in <u>Sections 23.2(a)</u> and (b), every power and remedy given by this Section to Lessor may be exercised from time-to-time, and as often as may be deemed expedient, by Lessor, subject at all times to Lessor's right to change any course of action undertaken by Lessor. Lessor hereby irrevocably waives and releases any and all rights and remedies with respect to any Lessee Event of Default other than as expressly granted in this <u>Section 23.2</u> provided, however, that this sentence shall not be deemed to prohibit collection by Lessor of Base Rent and Additional Rent which has accrued or arisen during the Term, including indemnification claims which arose during the Term pursuant to <u>Sections 21.1 and 22.3</u>, but without duplication of any amounts for which Lessor seeks damages or indemnification under this Lease.

23.3 Lessor Events of Default. Each of the following shall be an event of default under this Lease (each, a 'Lessor Event of Default'):

(a) *Monetary Defaults.* Lessor fails to make any payment due to Lessee under this Lease and such failure continues for fifteen (15) Business Days after Lessor's receipt of written notice of such failure from Lessee; or

(b) *Certain Representations.* Any representation or warranty made by Lessor under <u>Sections 4.1(f)</u> or (g) is false in any material respect at the time made and such representation or warranty continues to be untrue for thirty (30) days after Lessor's receipt of written notice of such matter; or

(c) Other Representations. Any other representation and warranty made by Lessor under this Lease is false at the time made and which, individually or in the aggregate with respect to each other such false representation or warranty, is material; or

(d) *Certain Specified Lessor Defaults* Any breach, default or failure of performance by Lessor or its Governing Documents under (i) Section 3.2(a) (*Prohibited Other Record Agreements*), (ii) Section 17.5(a)(i), (ii), (iii) or (iv) (*Prohibited Lessor Transfers, Indebtedness, Liens and Leases*), (iii) Section 27.1(a), (d), (e), (f) or (g) (SPE & Governing Document Requirements), or (iv) Sections 27.2(a), (b), (c), or (f) (*Loan Related Requirements*), and such breach, default or failure of performance remains uncured for sixty (60) days after Lessor's receipt of written notice of such breach, default or failure of performance; or

(e) Lessor Default under the SNDA. Any breach, default or failure of performance by Lessor under the SNDA, and such breach, default or failure of performance remains uncured for thirty (30) days after Lessor's receipt of written notice of such breach, default or failure of performance; or

(f) Equity Investor Defaults. Any breach, default or failure of performance under (i) Section 17.5(a)(v), (ii) Section 17.5(a)(v), or (iii) Sections 27.1(b), (c) or (h), and such breach, default or failure of performance remains uncured for sixty (60) days after Lessor's receipt of written notice of such breach, default or failure of performance, provided, however, that with respect to any such breach, default or failure of performance, Lessor shall be deemed to have cured such breach, default or failure of performance so long as (A) the Transfer which gave rise to such breach, default or failure of performance is null and void *ab initio* under Lessor's Governing Documents, (B) Lessor does not recognize any Lessor Equity Interest Owner who (or whose Beneficial Owner(s)) gave rise to such breach, default or failure of performance is null and void, (D) Lessor does not pay or otherwise permit to be made any

distribution or other funds or amounts to (I) the transferee of a Lessor Equity Owner if the Transfer to such transferee gave rise to such breach, default or failure of performance, or (II) to a Lessor Equity Interest Owner if one or more of its Beneficial Owner(s) gave rise to such breach, default or failure of performance, (E) Lessor does not make or permit to be made any allocations or other adjustments to the capital or similar account of a Lessor Equity Interest Owner(s) as a result of Lessor income, revenue, gains or otherwise if one or more of its Beneficial Owners gave rise to such breach, default or failure of performance, and (F) Lessor does not permit or allow a Lessor Equity Interest Owner(s) (if one or more of its Beneficial Owners gave rise to such breach, default or failure of performance) or its Beneficial Owner(s) to: (I) vote or have any influence or involvement in any governance matter, approval, consent or election with respect to Lessor, the Leased Property, Lessee or this Lease, (II) receive from Lessor or any of Lessor's beneficial owners any financial or other this Lease; <u>provided further</u>, that even if the conditions in <u>subsections (A)-(F)</u> above remain satisfied, if any Person receives a final judgment holding that (X) such Transfer giving rise to such breach, default or failure of performance is enforceable and remains in effect, or (Y) any of the actions or conditions in <u>subsections (A)-(F)</u>, above must cease or are otherwise unenforceable, then such breach, default or failure of performance shall be deemed not to be cured for purposes of this Lease; or

(g) *Failure to Deliver Documents in Connection with a Required Sale to Lessee or its Designee* Lessor fails to execute and deliver the documents required to Transfer the Leased Property to Lessee or its designee if Lessor has accepted Lessee's purchase offer under <u>Article XV</u>, <u>Article XVI</u> or <u>Section 25.5</u>, and such failure remains uncured for ten (10) Business Days after Lessor's receipt of written notice of such failure from Lessee; or

(h) Breach of ROFR Provisions. Any breach, default or failure of performance by Lessor under Section 25.3 (Transfer of Leased Property ROFR) or Section 25.4 (Lease of Leased Property ROFR) and such breach, default or failure of performance remains for ten (10) Business Days after Lessor's receipt of written notice of such failure from Lessee; or

(i) Other Lessor Covenant Defaults. Lessor fails to observe or perform any other covenant of Lessor under this Lease in any material respect (other than those specified in other subsections of this Section 23.3 which failure continues for sixty (60) days after Lessor's receipt of written notice of such breach from Lessee (or if the cure of such failure reasonably requires more than sixty (60) days to complete, if Lessor fails to commence the cure within such sixty (60) day period and thereafter diligently pursue such cure to completion); or

(j) **Debtor Relief Laws**. The filing by or against Lessor of a petition for relief under any Debtor Relief Laws (unless the same is dismissed within ninety (90) days after filing), or the appointment of a trustee or receiver to take possession of all, or substantially all, of Lessor's assets or of Lessor's interest in this Lease, where such appointment is not discharged in ninety (90) days after appointment of said trustee or receiver, or the voluntary filing of a petition for the appointment of the same.

23.4 Lessee's Remedies for a Lessor Event of Default

(a) *Lessee Remedies.* Upon the occurrence and during the continuance of a Lessor Event of Default, with or without notice or demand, except such notice as may be required by statute and cannot be waived by Lessor and such notice as is specifically required by the terms of this Lease (all other notices being hereby waived), this Lease shall remain in full force and effect, Lessee shall be entitled to remain in possession of the Leased Property hereunder and shall remain obligated and liable for Lessee's obligations hereunder (subject to Lessee's offset rights set forth below), and Lessee shall have and may exercise the following remedies, concurrently, successively, or in any combination, Lessee hereby waiving all other rights and remedies to which Lessee may be entitled at law or in equity:

(i) <u>Amounts Due by Lessor under the Lease</u>. Lessee may collect and bring an action against Lessor for all amounts due, owing and unpaid by Lessor as of the date of such action, including Lessor's indemnity obligations under this Lease with respect to matters or circumstances arising prior to the date of such action, but in any case without duplication for claims made under <u>subsection (a)(ii)</u> below; and

(ii) <u>Damages</u>. Lessee may bring an action against Lessor for Lessee's damages which are proximately caused by such Lessor Event of Default, but without duplication for other amounts and indemnities for which Lessor may be obligated under this Lease; and

(iii) <u>Specific Performance</u>. Lessee may bring an action against Lessor for specific performance by Lessor of its unperformed obligations under this Lease, but without duplication for other amounts, damages and indemnities for which Lessor may be obligated under this Lease. All covenants and agreements of Lessor in this Lease shall be deemed special, unique and extraordinary, and any breach of any covenant or agreement by Lessor shall be deemed to cause Lessee irreparable injury not properly compensable by damages in an action at law, such that the rights and remedies of Lessee may be enforced both at law or in equity; and

(iv) <u>Self-Help</u>. Lessee may perform, on Lessor's behalf, any unperformed covenant or obligation under this Lease constituting such Lessor Event of Default if such covenant or obligation remains unperformed on the date that is not less than ten (10) Business Days after entry of a final and non-appealable court order confirming such Lessor Event of Default, in which event, Lessor shall reimburse Lessee for all reasonable costs, expenses and disbursements incurred by Lessee in doing so, plus together with interest thereon at the Default Interest Rate from the date which is ten (10) Business Days after Lessee makes demand on Lessor for payment of same, but in any event without duplication for amounts for which Lessor seeks damages or indemnification under this Lease; and

(v) <u>Enforcement Costs</u>. Lessee shall be entitled to recover from Lessor all out-of-pocket costs and expenses, including reasonable attorneys' fees, court costs, expert witness fees, costs of tests and analyses, travel and accommodation expenses, deposition and trial transcripts, copies and other similar costs and fees, paid or incurred by Lessee in exercising its remedies under this <u>Section 23.4(a)</u> as a result of such Lessor Event of Default, regardless of whether or not legal proceedings are actually commenced, subject to <u>Section 26.9 of</u> this Lease; and

(vi) Special Payment for a Level 1A Lessor Default. So long as an Ultra Entity Person is the Lessee under this Lease and no Level 1 Lessee Default has occurred and is continuing, upon the occurrence and during the continuance of one or more Level 1A Lessor Defaults, beginning immediately upon notice from Lessee to Lessor, Lessor shall pay to Lessee an amount equal to \$10,000 per day until all Level 1A Lessor Defaults are cured (the "Level 1A Lessor Default Payment Amount"). Lessor may choose to pay the Level 1A Lessor Default Payment Amount due hereunder on a monthly basis, on or before the date five (5) Business Days prior to the end of each calendar month, in the amount of the aggregate unpaid Level 1A Lessor Default Payment Amount due through the end of such calendar month. If on the date five (5) Business Days prior to the end of any calendar month Lessor has not paid to Lessee the aggregate Level 1 A Lessor Default Payment Amount due through the end of such calendar month, then Lessee may offset against the monthly payment of Base Rent for the immediately succeeding calendar month the unpaid Level 1A Lessor Default Payment Amount due through the adjust any monthly payment of Base Rent in an amount which, when added to any offsets to such monthly payment of Base Rent under <u>subsection (vii)</u> below, would cause such monthly payment of Base Rent thereunder to be less than the Monthly Debt Service for such payment date. Following the cure of all Level 1A Lessor Defaults hereunder, within ten (10) Business Days following delivery of written notice from Lessor to Lessee of (a) evidence reasonably acceptable to

Lessee that all Level 1A Lessor Defaults have been cured, and (b) a correct calculation of the total Level 1A Lessor Default Payment Amounts paid by Lessor to Lessee or offset against Base Rent with respect to such Level 1A Lessor Defaults (collectively, with respect to such Level 1A Lessor Defaults, the "Level 1A Default Payments"), then (x) the Level 1A Lessor Payment Amount that has accrued and remains unpaid shall automatically be extinguished, and (y) Lessee shall pay to Lessor (or as otherwise directed by Lessor under this Lease or by Lessor Lender with respect to payment of Base Rent) the amount of such Level 1A Lessor Default Payments, less any amounts owing to Lessee under the preceding <u>subsections (i)-(v)</u> as of such date, and without payment by Lessee of interest at the Default Interest Rate thereon. Neither payment by Lessor to Lessee of any Level 1A Lessor Default Payment Amount nor Lessee's offset of any Level 1A Lessor Default Payment against Base Rent as permitted by this <u>subsection (vi)</u> shall be exclusive of any other remedies available under this <u>Section 23.4(a)</u>, and Lessor Default would be impractical and extremely difficult, if not impossible, to estimate or to determine, and the Level 1A Lessor Default Payment Amount is a fair and reasonable estimate of those damages which has been agreed to in an effort to cause the amount of such damages to be certain; and

(vii) Right to Offset. So long as no Level 1 Lessee Default has occurred and is continuing, in the event Lessor fails to pay to Lessee any amounts due to Lessee pursuant to <u>subsections (i)</u>, (iii), (iv), or (v), above, within ten (10) Business Days after written demand from Lessee, Lessee may offset such delinquent amounts, including interest thereon at the Default Interest Rate, from Base Rent otherwise due and owing by Lessee, whether previously or thereafter due and owing provided, however, that Lessee shall not offset such amounts against any monthly payment of Base Rent in an amount which, when added to any offsets under <u>subsection</u> (vi), above, with respect to such monthly payment of Base Rent, would cause any payment of monthly Base Rent hereunder to be less than the Monthly Debt Service for such payment date.

(b) **Delays and Waivers of Other Remedies.** No delay or omission of Lessee to exercise any right or power accruing upon the occurrence and during the continuance of any Lessor Event of Default shall impair any other or subsequent Lessor Event of Default or impair any rights or remedies consequent thereto unless such Lessor Event of Default is cured. Subject to the provisions of <u>subsection (a)</u>, above, which provide that such remedies are available without duplication of certain other claims, every power and remedy given by this <u>Section 23.4</u> to Lessee may be exercised from time-to-time, and as often as may be deemed expedient, by Lessee, subject at all times to Lessee's right to change any course of action undertaken by Lessee. Lessee hereby irrevocably waives and releases any and all rights and remedies with respect to any Lessor Event of Default other than as expressly granted in this <u>Section 23.4</u>, provided, however, that this sentence shall not be deemed to prohibit collection by Lessee of amounts due and owing from Lessor which payment obligation has accrued or arisem with respect to facts or circumstances during the Term, including indemnification claims which arose during the Term pursuant to <u>Sections 21.2</u> and <u>22.4</u>, but without duplication of any such amounts or claims and such remedies.

23.5 **Mitigation of Damages**. Notwithstanding anything to the contrary contained in this <u>Article XXIII</u>, (a) upon the occurrence of a Lessee Event of Default, Lesser shall be obligated to mitigate its damages hereunder, and (b) upon the occurrence of a Lessor Event of Default, Lessee shall be obligated to mitigate its damages hereunder.

ARTICLE XXIV. NOTICE

24.1 **Notices.** Except where otherwise specifically provided in this Lease, all notices, demands, requests or other communications (each, a "*Notice*") which either party is required to or may desire to give to



the other shall be in writing and shall be given by (a) personal delivery, (b) mailing a copy thereof by certified or registered mail, postage prepaid, return receipt requested, or (c) nationally recognized overnight courier service (such as Federal Express or UPS), all charges prepaid, furnishing a receipt upon delivery, in each case addressed to the party to whom the notice is directed at the Notice Address of such party set forth below. A notice given by Lessee may be given, in addition to the methods set forth in the preceding sentence, via electronic mail, and any Notice sent by Lessee via electronic mail shall be deemed to have been received by the addressee upon Lessee sending the email to the addressee's designated email address below. The Notice Address of each party is:

Lessee:

Ultra Wyoming LGS, LLC 400 North Sam Houston Parkway East, Suite 1200 Houston TX 77060 Attention: Marshall D. Smith, Senior Vice President and Chief Financial Officer

With a copy to:

Ultra Petroleum Corp. 400 North Sam Houston Parkway East, Suite 1200 Houston TX 77060 Attention: Legal Department

Lessor:

Pinedale Corridor, LP 4200 W. 115th Street, Suite 210 Leawood, KS 66211 Email Address: rick.green@corridortrust.com

With a copy to:

Husch Blackwell LLP 4801 Main Street, Suite 1000 Kansas City, MO 64112 Attn: Steve Carman Email Address: <u>steve.carman@huschblackwell.com</u>

The addresses to which notices and demands shall be delivered or sent may be changed from time-to-time by notice served by either party upon the other as provided

above.

24.2 **Deemed Receipt**. Unless otherwise provided in this Lease, and except for deemed receipt of email notices as provided in <u>Section 24.1</u>, notice shall be deemed to have been received by the addressee as follows: (a) if a Notice is delivered in person, or sent by registered or certified mail, or nationally recognized overnight courier, upon receipt by the addressee or delivery to the address of the addressee; and (b) if the addressee rejects or otherwise refuses to accept the Notice, or if the Notice cannot be delivered because of a change in address for which no Notice was given, then upon the rejection, refusal, or inability to deliver the Notice.

24.3 Delivery; Time of Notice. Notwithstanding the foregoing, if any Notice is received after 5:00 p.m. on a Business Day where the

addressee is located, or on a day that is not a Business Day where the addressee is located, then the Notice is deemed received at 9:00 a.m. local time on the next business day where the addressee is located.

ARTICLE XXV.

RENEWAL AND END OF TERM OPTIONS, RIGHT OF FIRST REFUSAL, AND OTHER PREFERENTIAL PURCHASE RIGHTS

25.1 Renewal Option.

(a) **Renewal Rights.** Provided no Level 1 Lessee Default has occurred and remains uncured as of the end of the Current Lease Term End, Lessee shall have the right and option (the "**Renewal Option**") to extend the term of the Lease with respect to all (but not less than all) of the Leased Property for successive renewal terms, each for a length of time equal to five (5) years, provided, however, that the maximum aggregate Term shall in no event exceed 99 years. Base Rent during each Renewal Term shall be the Fair Market Rent, determined as set forth below.

(b) *Lessee Renewal Notice.* Prior to the date that is twenty-one (21) months before the Current Lease Term End (the "*Renewal Notice Date*"), Lessee may exercise its Renewal Option if Lessee and Lessor agree in writing upon the Fair Market Rent for the applicable Renewal Term (the "*Agreed FMV Rent*").

(c) Determination of Agreed Fair Market Rent. Lessor and Lessee shall negotiate in good faith to reach agreement as to the Fair Market Rent and, if requested by Lessee, one or more Responsible Officers of Lessor shall meet with one or more Responsible Officers of Lessee at Lessee's offices in Houston, Texas or such other location as the parties shall mutually agree to conduct such negotiations in person. The Agreed FMV Rent shall be the Base Rent for the Leased Property for the applicable Renewal Term. If Lessor and Lessee are unable to agree upon the Fair Market Rent on or prior to the Renewal Notice Date, Lessee may notify Lessor in writing of its intent to exercise its Renewal Option subject to an acceptable determination of Fair Market Rent on or prior to the Renewal Notice Date (such notice, a "Fair Market Rent Determination Notice"). If the Lessee does not deliver the Fair Market Rent Determination Notice on or before the Renewal Notice Date, Lessee will be deemed to have decided not to elect to exercise the Renewal Term Option and the Lease will expire on the Current Lease Term End. If Lessee delivers a Fair Market Rent Determination Notice to Lessor, then each of Lessor and Lessee shall submit to the other in writing (the "Designation Notice"), no sooner than fifteen (15) Business Days after Lessee's delivery of such Fair Market Rent Determination Notice, (i) its good faith estimate of the Fair Market Rent during the applicable Renewal Term (collectively referred to as the "Estimates"), and (ii) the name of the Independent Appraiser appointed by it to determine which of the two (2) Estimates most closely reflects the Fair Market Rent for the Leased Property during the applicable Renewal Term, and Fair Market Rent shall be determined as follows:

(i) If either Lessor or Lessee fails to appoint an Independent Appraiser within the fifteen (15) Business Day period referred to above, the Independent Appraiser appointed by the other party shall be the sole Independent Appraiser for the purposes hereof and shall determine which of the two (2) Estimates most closely reflects the Fair Market Rent within twenty (20) Business Days after the Designation Notice appointing such Independent Appraiser, and the Estimate so chosen shall be binding on both Lessor and Lessee as the Base Rent for the Leased Property during the applicable Renewal Term.

(ii) If both Lessor and Lessee appoint an Independent Appraiser within the fifteen (15) Business Day period referred to above, Lessor's and Lessee's Independent Appraisers shall work together in good faith to agree upon which of the two (2) Estimates most closely reflects the Fair Market Rent for the Leased Property within twenty (20) Business Days after the Designation Notices appointing such Independent Appraisers (or the later of the two Designation Notices). The Estimate

chosen by such Independent Appraisers shall be binding on both Lessor and Lessee as the Base Rent for the Leased Property during the applicable Renewal Term.

(iii) If the two (2) Independent Appraisers have not agreed upon which of the two (2) Estimates most closely reflects the Fair Market Rent by midnight on the date which is twenty (20) Business Days after the Designation Notices appointing such Independent Appraisers (or the later of the two Designation Notices), then, within ten (10) Business Days after the expiration of such twenty (20) Business Day period, the two (2) Independent Appraisers shall select a third Independent Appraiser. Once the third Independent Appraiser has been selected, then, as soon thereafter as practicable but in any case within fifteen (15) Business Days, the third Independent Appraiser shall make his or her determination of which of the two (2) Estimates most closely reflects the Fair Market Rent and such Estimate shall be binding on both Lessor and Lessee as the Base Rent for the Leased Property during the applicable Renewal Term. If the two (2) Independent Appraisers are unable to agree upon a third Independent Appraiser within such fifteen (15) Business Day period, the third Independent Appraiser shall be appointed as soon as possible by the American Arbitration Association (or any successor organization, or if no successor organization shall then exist, by a court of competent jurisdiction residing in Harris County, Texas, or such other jurisdiction as Lessor and Lessee mutually agree), subject to the qualification requirements set out herein. In the event of the failure, refusal or inability of any Independent Appraiser to act, a new Independent Appraiser.

(iv) Notwithstanding the foregoing procedures of this <u>Section 25.1(c)</u>. Lessor and Lessee may terminate the foregoing procedures to determine the Fair Market Rent for the applicable Renewal Term, if at any time during the process, they agree in writing as to what constitutes the Fair Market Rent for the applicable Renewal Term and such determination shall be binding on both Lessor and Lessee as the Base Rent for the Leased Property during the applicable Renewal Term.

(v) The parties shall share equally in the costs of the third Independent Appraiser. The fees of any appraiser, counsel or experts engaged directly by Lessor or Lessee shall be borne by the party retaining the appraiser, counsel or expert.

(d) Lessee Right to Rescind Upon Final Determination of Fair Market Rent. If Fair Market Rent is determined pursuant to <u>subsection (c)</u>, above, then on or before the date which is thirty (30) days following the determination of the Fair Market Rent pursuant to <u>subsection (c)</u>, above, Lessee shall elect, by written notice to Lessor, whether to (i) rescind Lessee's election to extend the Lease Term for the applicable Renewal Term, in which case Lessee shall be deemed to have elected not to extend the Lease Term for the applicable Renewal Term, the Lease shall be deemed to have been extended for such period and Lessee's possession during such period shall not constitute a default or "holding over" after the Lease Term), or (ii) renew the Lease for the applicable Renewal Term for the Fair Market Rent as so determined, in which case during such Renewal Term all of the terms and conditions of this Lease will continue in full force and effect, with Base Rent being determined as set forth in <u>subsection (c)</u>, above. Any failure of the parties to determine Fair Market Rent within the timeframes set forth in this <u>Section 25.1</u> shall not invalidate any of Lessee's rights under this <u>subsection (d)</u> or any other provision of this <u>Section 25.1</u>. If Lessee fails to timely notify Lessor in writing of Lessee's decision to rescind its extension notice or to renew the Lease as provided in this <u>subsection (d)</u>, Lessee will be deemed to failure of Lessee to extend the Lease Term for such Renewal Term as provided in <u>subsection (i)</u>, above. Upon expiration of the Lease Term due to failure of base to extend the Lease Term for a Renewal Term or actual or deemed rescission by Lessee of its election to so extend for the Renewal Term, (A) Lessor may (but shall not be obligated to) market the Lease Property for sale or lease, subject in each case to <u>Sections 25.3</u>

and 25.4 hereof, and (B) the Lease shall terminate except the provisions hereof which survive termination, includingSections 25.3 and 25.4 hereof.

(e) Payment of Renewal Term Base Rent Prior to Final Determination of Fair Market Rent In the event the procedures outlined in <u>subsections (c)</u> and (d), above, are not completed and, therefore, the Fair Market Rent for the applicable Renewal Term is not determined as of the Current Lease Term End, then Lessee shall pay Base Rent for the Leased Property during the applicable Renewal Term based on the average of the Estimates provided by Lessor's Independent Appraiser and Lessee's Independent Appraiser. If it is thereafter determined that the Fair Market Rent for the applicable Renewal Term is an amount different than the amount paid by Lessee, the Fair Market Rent as so determined shall be effective from the date of commencement of the applicable Renewal Term, and Lessee or Lessor, as the case may be, shall pay to the other party, within twenty (20) Business Days after the determination of such Fair Market Rent for the applicable Renewal Term, the difference between the rent payable based upon the Fair Market Rent for the applicable Renewal Term, the difference between the rent payable based upon the Fair Market Rent for the applicable Renewal Term, and Lessee to Lessor, as the case may be, shall pay to the other party, within twenty (20) Business Days after the determination of such Fair Market Rent for the applicable Renewal Term, the difference between the rent payable based upon the Fair Market Rent for the applicable Renewal Term, the difference between the rent payable based upon the Fair Market Rent for the applicable Renewal Term and the total Base Rent actually paid to the date of such payment. If the amount due by Lesser to Lesser to Lessor, as the case may be, under the foregoing sentence is not paid within such twenty (20) Business Day period, such amount shall thereafter bear interest from the Due Date for Additional Rent (with respect to payments by Lessee) or ten (10) Business Days following written demand from Lessee to Lessor (with respect to payments by Lesser) at a rate equal to the Default I

25.2 Lessee as Operator After Term. If Lessor elects by written notice to Lessee given no later than 90 days prior to the end of the Lease Term, then following the termination of this Lease, Lessee or its Affiliate will continue to operate the Leased Property after the Term as an operator and not a lessee, pursuant to the Services Agreement and, in such event, Lessee or its Affiliate may elect to use the Leased Property during any such period pursuant to the Gathering Agreement, in each case with such revisions as Lessor and Lessee shall mutually agree (together and with such mutually acceptable revisions, the "Operating Agreements"), for a Monthly Operating Fee equal to the fair market value for applicable services and use under the Operating Agreements as Lessor and Lessee shall negotiate in good faith, for a period of up to two (2) years following the end of the Lease Term, as specified in the notice from Lessor, but subject in any event to the termination rights provided in the Operating Agreements. Upon timely written request from Lessor as provided in this <u>Section 25.2</u>, Lessor and Lessee shall execute and deliver the Operating Agreements and pay or cause to be paid the Monthly Operating Fee in connection therewith.

25.3 Right of First Refusal With Respect to the Transfer of All of the Leased Property.

(a) Limitations on Lessor Transfers. Lessor covenants and agrees that (i) during the ROFR Transfer Period, Lessor shall not Transfer less than all of its right, title or interest in the Leased Property or this Lease, (ii) any Transfer of all of Lessor's right, title and interest in the Leased Property and this Lease during the Lease Term shall be subject to Section 17.5 and this Section 25.3, and (iii) any Transfer of all of Lessor's right, title and interest in the Leased Property and this Lease after the end of the Lease Term, but during the remainder of the ROFR Transfer Period, shall be subject to this Section 25.3.

(b) **Transfers Subject to ROFR Rights.** If at any time or from time-to-time during the ROFR Transfer Period Lessor desires to Transfer or agrees to Transfer all of its right, title and interest in and to the Leased Property and in and under this Lease, in any such case to a Person that is not an Affiliate of Lessor and Lessor has received a Third Party Asset Offer with respect to such Transfer, then (subject to the immediately succeeding sentence of this <u>Section 25.3(b)</u>), Lessor may not consummate such Transfer unless Lessor first complies with the terms and conditions of this <u>Section 25.3</u>. Notwithstanding the foregoing or anything in this <u>Section 25.3</u> to the contrary, (i) Lessor shall not be permitted to consummate any Transfer of all or any portion of its right, title and interest in and to the Leased Property and in and under this Lease Term which is not a Permitted Sale even if Lessee does not elect to exercise its right of first

refusal under this <u>Section 25.3</u> with respect to such Transfer, (ii) the restrictions on Transfers by Lessor under <u>Section 17.5</u> shall not be applicable to any Transfer of Lessor's right, title and interest in and to the Leased Property and in and under this Lease after the Lease Term, although the prohibition on partial Transfers as set forth in <u>subsection (a)</u>, above, shall remain applicable during the remaining ROFR Transfer Period, and (iii) this <u>Section 25.3</u> shall not be applicable, and Lessor shall not be required to comply with this <u>Section 25.3</u> with respect to a proposed Transfer by Lessor of all or any portion of its right, title and interest in and to the Leased Property and in and under this Lease after the Lease Term if either (A) this Lease has been terminated by Lessor pursuant to <u>Section 23.2(a)(i)</u> as a result of a Level 1 Lessee Default, or (B) as of the end of the Lease Term an Ultra Entity Person is not the Lessee under this Lease. Lessee's rights under this <u>Section 25.3</u> are absolute, appurtenant and shall run with the land with respect to the Leased Property and shall survive termination of this Lease for the ROFR Transfer Period.

(c) **ROFR** Asset Sale Notice. If Lessor desires to accept a Third Party Asset Offer (or if the Third Party Asset Offer is a binding contract that Lessor desires to execute), then Lessor shall provide written notice to Lessee of the Third Party Asset Offer (an "**ROFR** Asset Sale Notice"). The ROFR Asset Sale Notice will include the following information:

(i) a copy of the Third Party Asset Offer and any related documents (such as an agreed upon form of contract);

(ii) a package containing any information about the Leased Property that Lessor has provided to the proposed buyer;

(iii) an identification of the direct and indirect owners of the proposed buyer (and each Person that will have Controlling Lease Rights) and information as to the business, character, reputation and financial capacity of the proposed buyer and the principals associated therewith to carry out the terms of the Third Party Asset Offer and, during the Lease Term, to comply with all requirements of this Lease and the Equity Investor Agreement in the event such Transfer is consummated; and

(iv) during the Lease Term, such other documents and information as Lessee may reasonably require to evidence that the proposed Transfer will qualify as a Permitted Sale hereunder.

(d) **ROFR** Asset Offer. The delivery of the ROFR Asset Sale Notice constitutes an offer (an "**ROFR** Asset Offer") by Lessor to sell the Leased Property to Lessee in accordance with the terms set forth in this Section 25.3.

(e) Lessee Election. Lessee will have fifteen (15) Business Days from the date of its receipt of the ROFR Asset Offer (the 'ROFR Asset Election Period') to elect to acquire the Leased Property pursuant to the same terms the proposed buyer would acquire the Leased Property under the Third Party Asset Offer. If Lessee desires to acquire the Leased Property pursuant to such terms, then prior to the end of the ROFR Asset Election Period, Lessee must deliver a notice of acceptance (the "ROFR Asset Acceptance Notice") to Lessor. If Lessee rejects the Third Party Asset Offer or does not timely accept such offer, then Lessor may proceed to close on the Third Party Asset Offer with the proposed buyer in accordance with the substantially unmodified terms of the Third Party Asset Offer or J if such Transfer is completed within one hundred fifty (150) days after the last day of the ROFR Asset Election Period. If the terms of such Transfer are modified pursuant to a Material Amendment or such Transfer is not completed within such one hundred fifty (150) day period, then Lessee shall again have the opportunity to be the buyer under such transaction, meaning that Lessor must send Lessee a new ROFR Asset Sale Notice and comply again with this <u>Section 25.3</u>. At no time during the Lease Term may Lessor elect to Transfer all or any portion of the Lessor Interests or any interest therein except pursuant to a Permitted Sale, whether or not Lessee has elected or rejected the ROFR

Asset Offer and any such Transfer by Lessor in violation of this Section 25.3 or Section 17.5 shall be void *ab initio*. This restriction on Transfer and Lessee's right of first refusal rights under this Section 25.3 are absolute, appurtenant, and shall run with the land with respect to the Leased Property and this Lease.

(f) *Closing*. The closing of the purchase of the Leased Property by Lessee under this <u>Section 25.3</u> will be held within sixty (60) days after the delivery of the ROFR Asset Acceptance Notice or such longer period as required to complete the process described in <u>Subsection (g)</u> below, if applicable. At the closing, Lessor will deliver conveyance documents, each in form and substance contemplated by the Third Party Asset Offer, to transfer the ownership of the Leased Property to Lessee.

(g) **Determination of FMV of Non-Cash Consideration.** If any part of the consideration to acquire the Leased Property as described in the Third Party Asset Offer is not in cash, then before the end of the ROFR Asset Election Period, Lessee may also elect to have the fair market value of such non-cash consideration (the "Non-Cash Consideration") determined pursuant to this Section.

(i) Lessor and Lessee will attempt to agree on the Fair Market Value of the Non-Cash Consideration as of the date of the ROFR Asset Offer (the "Non-Cash Consideration Value"). If Lessor and Lessee do not agree within ten (10) Business Days following the ROFR Asset Election Period as to the deemed Non-Cash Consideration Value, then Lessor and Lessee will attempt to agree on an Independent Appraiser to determine such Non-Cash Consideration Value. If Lessor and Lessee agree on such Independent Appraiser on or before fifteen (15) Business Days following the end of the ROFR Asset Election Period, then such Independent Appraiser will determine such Non-Cash Consideration Value on or before thirty (30) Business Days following the end of the ROFR Asset Election Period, which determination will be binding and conclusive on each of Lessor and Lessee. If Lessor and Lessee do not agree on such Independent Appraiser on or before fifteen (15) Business Days following the end of the ROFR Asset Election Period, then (15) Business Days following the end of the ROFR Asset Election Period, which determination will be binding and conclusive on each of Lessor and Lessee. If Lessor and Lessee do not agree on such Independent Appraiser on or before fifteen (15) Business Days following the end of the ROFR Asset Election Period, then Lessee and Lessor will comply with the following procedure to determine the Independent Appraiser and the Non-Cash Consideration Value as of the date of the ROFR Asset Offer:

(ii) On or before twenty (20) Business Days following the end of ROFR Asset Election Period, each of Lessee and Lessor will submit to each other in writing a list of three proposed Independent Appraisers. If either Lessor or Lessee fails to timely provide such list to the other, then the Independent Appraiser will be decided by the other from its list of proposed Independent Appraisers.

(iii) If the name of exactly one proposed Independent Appraiser appears on both the list submitted by Lessor and the list submitted by Lessor, and the name so drawn will be the Independent Appraiser. Such drawing will occur twenty-five (25) Business Days following the ROFR Asset Election Period.

(iv) The Independent Appraiser, as selected by the process described in this Section, will, within forty (40) Business Days following the end of the ROFR Asset Election Period, (i) determine the Non-Cash Consideration Value as of the date of the Third Party Asset Offer; and (ii) notify in writing Lessor and Lessee of such determination.

(v) The conclusions of the Independent Appraiser as to the Non-Cash Consideration Value as of the date of the Third Party Asset Offer will be binding on each of Lessor and Lessee. The fees and expenses of the Independent Appraiser (and any appraisers engaged by it) will be paid one-half by each of Lessor and Lessee.

(vi) Once the Non-Cash Consideration Value as of the date of the Third Party Asset Offer is determined pursuant to the procedures above and such determination is made known in writing to Lessee, then Lessee will have a right to do any of the following: (A) rescind the ROFR Asset Acceptance Notice (in which event Lessor may proceed under <u>Section 25.3(e)</u> as if the ROFR Asset Acceptance Notice has not been given except that the 150 day period referred to in <u>Section 25.3(e)</u> shall commence as of the date that Lessee elects to rescind the ROFR Asset Acceptance Notice), (B) not rescind the ROFR Asset Acceptance Notice but elect to not pay the Non-Cash Consideration to Lessor and instead pay the Non-Cash Consideration Value to Lessor, or (C) not rescind ROFR Asset Acceptance Notice and pay the Non-Cash Consideration to the cash consideration as specified in the Third Party Asset Offer) to Lessor. Such election shall be made within ten (10) Business Days after the date the Independent Appraiser advises Lessee and Lessor in writing of the Non-Cash Consideration Value, and if not timely made, Lessee shall be a deemed to have made an election under <u>subsection (A)</u> above.

25.4 Right of First Refusal With Respect to a Lessor Lease Transaction.

(a) *Limitations on Lessor Lease Transactions.* Lessor covenants and agrees that (i) Lessor shall not enter into (A) a Lessor Lease Transaction having a lease term which includes all or any portion of the Lease Term, or (B) at any time during the ROFR Lease Period, any Lessor Lease Transaction with respect to less than all of the Leased Property, (ii) any Lessor Lease Transaction with respect to all of the Lease Term shall be subject this <u>Section 25.4</u>, and (iii) any Lessor Lease Transaction during the ROFR Lease Period with respect to all of the Leased Property shall be subject to this <u>Section 25.4</u>, and (iii) any Lessor Lease Transaction during the ROFR Lease Period with respect to all of the Leased Property shall be subject to this <u>Section 25.4</u>.

(b) Lease Transactions Subject to ROFR. If at any time or from time-to-time (i) either (A) during the Lease Term Lessor desires to enter into a Lessor Lease Transaction with a lease term which will begin after the end of the Lease Term, or (B) during the ROFR Lease Period Lessor desires to enter into a Lessor Lease Transaction, in the case of either such <u>subsections (A)</u> or (B) with respect to all or any portion of the Leased Property (the portion of the Leased Property (which may be all of the Leased Property) that Lessor desires to lease is referred to as the "Applicable Leased Assets") to a Person that is not an Affiliate of Lessor and Lessor has received a Third Party Lease Offer with respect thereto, and (ii) the Lease was or is not terminated by Lessor pursuant to <u>Section 23.2(a)(i)</u> as a result of a Level 1 Lessee Default, and (iii) an Ultra Entity Person was or is the Lessee under this Lease as of the end of the Lease Term, then (I) Lessor may consummate such Lessor Lease Transaction of the Applicable Leased Assets only if Lessor complies with this <u>Section 25.4</u>. In addition, Lessor shall not enter into a Lessor Lease Transaction within two (2) years after the end of the Term without complying with this <u>Section 25.4</u>. In addition, Lessor shall not enter into a Lessor Lease Transaction with a subject to this<u>Section 25.4</u> except for all cash consideration. For the avoidance of doubt, if (X) this Lease is terminated by Lessor pursuant to <u>Section 23.2(a)(i)</u> as a result of a Level 1 Lessee Default, or (Y) as of the end of the Lease Term an Ultra Entity Person was not or is not the Lessee under this Lease, the provisions of this <u>Section 25.4</u> shall not apply and Lessor may consummate a Lessor Lease Transaction without complying with the terms and conditions of this <u>Section 25.4</u>. This restriction on Lessor Lease Transactions and Lessee's right of first refusal rights under this <u>Section 25.4</u> are absolute, appurtenant, shall run with the land with respect to the Leased Property and this Lease, and su

(c) ROFR Lease Notice. If Lessor desires to accept a Third Party Lease Offer (or if the Third Party Lease Offer is a binding contract that Lessor desires to execute), then Lessor shall provide written

notice to Lessee of the Third Party Lease Offer (a "ROFR Lease Notice"). The ROFR Lease Notice will include the following information:

- (i) a description of the Applicable Leased Assets;
- (ii) a copy of the Third Party Lease Offer and any related documents (such as an agreed upon form of lease); and
- (iii) a package containing any information about the Leased Property that Lessor has provided to the proposed lessee, licensee or sublessee; and

(iv) an identification of the principals associated with the proposed lessee, licensee or sublessee under such Third Party Lease Offer and information as to the business, character, reputation and financial capacity of the proposed lessee, licensee or sublessee and the principals associated therewith to carry out the terms of the Third Party Lease Offer.

(d) **ROFR Lease Offer.** The delivery of the ROFR Lease Notice constitutes an offer (a "**ROFR Lease Offer**") by Lessor to lease the Applicable Lease to Lessee in accordance with the terms set forth in this <u>Section 25.4</u>.

(e) Lessee ROFR Election. Lessee will have a period of fifteen (15) Business Days from the date of its receipt of the ROFR Lease Offer (this 15-Business Day period is the "ROFR Lease Election Period") to elect to enter into the Lessor Lease Transaction pursuant to the same terms the proposed lessee would lease the Applicable Lease Assets under the Third Party Lease Offer (and, to the extent that the Applicable Lease Offer provides for any Non-Cash Consideration, Lessor and Lessee shall apply the same procedures to determine the Non-Cash Consideration Value a is set forth in Section 25.3). If Lessee desires to lease the Applicable Lease Assets pursuant to such terms, then prior to the end of the ROFR Lease Election Period, Lessee must deliver a written notice of acceptance (the "ROFR Lease Acceptance Notice") to Lessor. If Lessee rejects the Third Party Lease Offer or does not timely accept such offer, then Lessor may proceed to close on the Third Party Lease Offer with the proposed lessee, licensee or sublessee in accordance with the substantially unmodified terms of the Third Party Lease Offer, provided that if such Lessor Lease Transaction does not occur within one hundred fifty (150) days after the date of delivery of the ROFR Asset Acceptance Notice or if the terms of such Lessor Lease Transaction are modified pursuant to a Material Amendment, then Lessee shall again have the opportunity to be the lessee, licensee or sublessee under such Lessor Lease Transaction, meaning that Lessor must send Lesser a new ROFR Lease Transaction in violation of this Section 25.4 shall be void *ab initio*. This restriction on Lessor Lease Transactions is absolute, appurtenant, and shall run with the land with respect to the Leased Property.

(f) **Closing.** A Lessor Lease Transaction of the Applicable Lease Assets by Lessee under this Section (i) will be evidenced by a lease in the form included in and which is a part of the ROFR Lease Notice, but if no such lease form is included in and a part of such ROFR Lease Notice, then in substantially the form of this Lease, revised as necessary or appropriate to conform to the terms of ROFR Lease Notice, and (ii) such lease and any other documents required thereby shall be executed by Lessor and Lessee within thirty (30) days after the date of delivery of the ROFR Lease Acceptance Notice. Notwithstanding the foregoing, Lessor and Lessee agree that no lessee guarantor, lessee guaranty or Resources Guaranty, or lessor guarantor or lessor guaranty shall be required unless as and to the extent required by the ROFR Lease Notice.

25.5 Burdensome Buyout.

(a) Burdensome Buyout Notice. At any time and from time-to-time, upon the occurrence and during the continuance of a Burdensome Event, Lessee shall have the option, but not the obligation, to



provide written notice (the "Burdensome Buyout Notice") to Lessor proposing a termination of this Lease and offering to purchase all of the Leased Property as provided in this Section 25.5 for the purchase price as specified by Lessee in such Burdensome Buyout Notice.

(b) Burdensome Buyout Response Notice. If Lessee delivers its Burdensome Buyout Notice to Lessor, then Lessor shall have a period of no more than fifteen (15) days in which to either notify Lessee in writing of Lessor's acceptance or rejection of Lessee's Burdensome Buyout Notice ("Burdensome Buyout Response Notice"). Lessor's failure to provide its Burdensome Buyout Response Notice within such fifteen (15) day period shall be deemed a rejection of Lessee's Burdensome Buyout Notice and the Lease shall continue in effect.

(c) *Termination of Lease*. If Lessor accepts the Burdensome Buyout Notice, (i) this Lease shall terminate on a date no later than sixty (60) days after the Burdensome Buyout Response Notice specified in the Burdensome Buyout Notice and this Lease shall terminate on such date (the "*Burdensome Buyout Lease Termination Date*"), (ii) except as provided in <u>Sections 25.2, 25.3</u> and <u>25.4</u>, and the other terms and provisions of this Lease which survive termination, neither Lessor nor Lessee shall have any rights or obligations under this Lease arising from and after the Burdensome Buyout Lease Termination Date, (iii) no further Base Rent or Additional Rent shall accrue with respect to the period after the Burdensome Buyout Lease Termination Date, (iv) Lessee shall have no obligation to take any Corrective Action with respect to the circumstances giving rise to the Burdensome Event, (v) neither Lessor Indemnified Party shall have any claim against Lessee with respect to any Corrective Action with respect to the circumstances giving rise to the Burdensome Event, or Losses arising from termination of this Lease or Lessee's purchase of the Leased Property pursuant to this <u>Section 25.5</u>, (vi) all Lessee Guarantors shall be released of liability and obligations under their Lessor Guarantees with respect to (x) as to Lessee Guarantors and Ultra Resources only, matters for which Lessee is released in <u>subsections (iv)</u> and (y) matters arising or accruing from and after the Burdensome Buyout Lease Termination Date except those matters described in <u>subsections (iv)</u> and (y), preceding, and (yi) if the circumstances giving rise to the Burdensome Buyout Lease Termination Date except those matters described in <u>subsections (iv)</u> and (yi) if the circumstances giving rise to the Burdensome Event include a casualty or a Taking, then the Casualty Proceeds or the Taking Proceeds, as applicable, shall be paid to Lessee.

(d) Closing of Purchase Offer. If Lessor accepts the Burdensome Buyout Notice, on the Burdensome Buyout Lease Termination Date (i) Lessee shall pay to Lessor the purchase price specified by Lessee in the Burdensome Buyout Notice or such other purchase price as Lessor and Lessee mutually agree, (ii) Lessor shall convey the Leased Property to Lessee or its designee on an AS IS WHERE IS basis, without any express or implied warranties from Lessor other than special warranties of title, provided, however, that such conveyance shall in any event be made by Lessor to Lessee or its designee free and clear of (A) all Permitted Lessor Liens, (B) any other Liens placed on the Leased Property by, through or under Lessor, and (C) any Other Recorded Documents executed by, through or under Lessor in violation of this Lease, and (iii) Lessor will deliver to Lessee or its designee such conveyance documents, each in form and substance reasonably satisfactory to Lessee, as Lessee may reasonably request, to transfer the ownership of the Leased Property to Lessee free and clear of the Liens and other items described in subsections (A)-(C) above.

(e) Continuation of Lease if Lessor Rejects Lessee's Burdensome Buyout Notice. If Lessor rejects or is deemed to have rejected Lessee's Burdensome Buyout Notice, such Burdensome Buyout Notice with respect to termination of the Lease and Lessee's offer to purchase the Leased Property shall be deemed to be automatically rescinded and this Lease shall remain in full force and effect in accordance with its terms.

(f) *No Waiver*. Upon each occurrence of a Burdensome Event, Lessee shall be entitled to give its Burdensome Buyout Notice with respect to such Burdensome Event, even if Lessee previously elected not to give such notice with respect to the occurrence of a prior Burdensome Buyout and even if Lessee's prior Burdensome Buyout Notice(s) were rejected by Lessor hereunder.

25.6 **Application of Certain Right of First Refusal Rights to a Foreclosure Sale**. Notwithstanding anything to the contrary herein, Lessee's right of first refusal rights under <u>Section 25.3</u> hereof shall not be applicable to a Transfer of the Leased Property made pursuant to a foreclosure or deed in lieu of foreclosure with respect to a Permitted Lessor Lien. For the avoidance of doubt, Lessee's rights under both <u>Section 25.3</u> and <u>25.4</u> shall survive any foreclosure sale or deed in lieu of foreclosure or similar conveyance with respect to the Leased Property and this Lease.

ARTICLE XXVI. MISCELLANEOUS PROVISIONS

26.1 **Memorandum/Notice of Lease**. Simultaneously with the execution of this Lease, Lessor and Lessee shall execute and deliver the Memo of Lease and cause it to be recorded in the Official Public Records. Upon termination of this Lease, upon the request of either party, the other party will execute an instrument in recordable form indicating that this Lease has been terminated. No mortgages, deeds of trust, fixture filings, UCC financing statements or other Lien filings will be recorded by Lessor or any Lessor Lender in the Official Public Records or in personal property UCC records naming Lessee as a debtor with respect to any of the Leased Property.

26.2 **Force Majeure**. If either party shall be delayed or hindered in, or prevented from, the performance of any act required under this Lease by reason of strikes, lockouts, labor troubles, riots, insurrection, war or other reasons of a like nature beyond the reasonable control of the party delayed in performing work or doing acts required under the terms of this Lease (any such delay, hindrance or prevention being referred to as "*Force Majeure*"), then performance of such act shall be excused for the period of delay, and the period of the performance of any such act shall be extended for a period equivalent to the period of such delay unless otherwise specifically provided to the contrary in this Lease. The provisions of this Section shall not apply to delays in the payment of amounts due and owing by Lessee or Lessor hereunder or resulting from the inability of a party to obtain financing or to satisfy its obligations under this Lease because of a lack of funds.

2 6 . 3 **Consequential Damages**. Notwithstanding anything in this Lease to the contrary, in no event shall Lessor or Lessee be liable or responsible for (a) consequential, punitive, special or indirect damages, or loss of profits, or (b) except to the extent not excluded from "Losses" in the definition of such term in this Lease, diminution in value under this Lease, and including with respect to both <u>subsections (a)</u> and (b), any holding over by Lessee under <u>Section 26.4</u> of this Lease.

26.4 **Holding Over**. Except for continuations of Lessee's occupancy after the Lease Term pending determination of rent for a Renewal Term as provided in <u>Section</u> 25.1(e), if Lessee remains in possession of any Leased Property after the expiration of the Term without having timely executed its right, if any, to extend the Term, such continuing possession shall create a month to month tenancy on the terms of this Lease except that the monthly Base Rental shall be 125% of the Base Rent applicable as of the end of the Term, and such tenancy may be terminated at the end of any month thereafter by either party giving at least sixty (60) days' notice to the other party.

26.5 **Quiet Enjoyment**. Lessor agrees that so long as this Lease is in effect, Lessee shall and may peaceably and quietly have, hold and enjoy the Leased Property and all rights of Lessee hereunder during the Term without any manner of hindrance or molestation from Lessor, Lessor Lender, or anyone claiming by, through or under Lessor or Lessor Lender.

26.6 **Cost and Expense**. Wherever it is provided in this Lease that an act is to be undertaken by any Person, such act shall be done by such Person or caused to be done by such Person at no cost or expense to the other party, other than any *de minimus* cost or expense, unless a contrary intent is expressed.

26.7 Access; Reporting.

(a) Access and Inspection Rights. Lessor and Lessor Lender (or the Agent for such Lessor Lender if requested in writing by Lessor) may, at their sole cost and expense, inspect the Leased Property during normal business hours at such locations where the same is located or kept in Lessee's ordinary course of business, after not less than five (5) Business Days prior notice to Lessee; provided, however, neither Lessor, any Lessor Lender nor any Agent for any Lessor Lender shall have any right to inspect any of the Leased Property without a representative of Lessee being present and Lessor and Lessor Lender (or its Agent, if applicable) shall abide by Lessee's reasonable rules and regulations (including those governing matters of health and safety) when making such inspections; provided, further, that, unless (i) Lessee's interest under this Lease has been assigned to a Person that is not a Permitted Lessee Transferee or (ii) a Level 1 Lessee Default has occurred and is continuing, and excluding inspections made with respect to restorations following a casualty or condemnation as such inspections are permitted by and pursuant to Article XV or Article XVI, Lessor Lender (or the Agent for such Lessor Lender if requested in writing by Lessor) may inspect the Leased Property no more frequently than once per calendar year and Lessor may inspect the Leased Property no more frequently than twice per calendar year. Lessee shall use commercially reasonable efforts to cause its representative to be present for Lessor's inspection when notice is given as required by this subsection (a). In the event Lessee is unable to provide a representative of Lessee to be present for a Lessor or Lessor Lender inspection at any time during a thirty (30) consecutive day period, Lessee shall pay to Lessor a \$10,000 penalty. Under no circumstances shall Lessor permit its inspection or any inspection by Lessor Lender (or the Agent for such Lessor Lender if requested in writing by Lessor) to interfere with Lessee's ordinary operation of the Leased Property or with Lessee's business. Under no circumstances may Lessor or Lessor Lender permit any Disqualified Persons to inspect the Leased Property; provided, however, that if Lessee has not elected to renew this Lease beyond the Current Lease Term End, then commencing the last day on which the Renewal Option may be exercised, Lessor may allow a potential purchaser or replacement lessee of all or any part of the Leased Property (including a Disqualified Person) to accompany it in an inspection conducted under this Section, provided that any such Person shall execute a confidentiality agreement containing the restrictions on disclosure herein set forth prior to any such inspection. Lessor acknowledges and stipulates that Lessee may suffer irreparable harm in the event of a breach of the provisions of this Section 26.7(a) for which Lessee has no adequate remedy at law. Therefore, in addition to all other remedies available pursuant to the terms of this Lease or at law, Lessee shall have the right to obtain immediate injunctive or other equitable relief upon a breach of this Section 26.7(a)

(b) *Lessee Reporting Obligations*. Lessee shall:

month;

(i) furnish, not later than 60 days after the end of each calendar month during the Term, a report stating the Actual Daily Product Volume for such prior

(ii) use its commercially reasonable efforts to provide to Lessor the financial statements and Audit Report provided for in paragraphs (iii), (iv) and (v) of this Section prior to the date such financial statements must be filed by Lessor or Lessor Guarantor with the Securities and Exchange Commission, and will provide Lessor its best estimate of the dates on which Lessee Guarantor plans to file its Forms 10-K and Forms 10-Q with the Securities and Exchange Commission at least five days prior to any such filing date;

(iii) not later than the date on which a large accelerated filer (as defined in the rules of the Securities and Exchange Commission) with a fiscal year ending on December 31 is required to file its Form 10-K with the Securities and Exchange Commission, including any extension permitted by Rule 12b-25 of the Exchange Act, cause Lessee Guarantor (or, if there is no Lessee Guarantor, then Lessee) to furnish

to Lessor such Person's audited consolidated balance sheet and related statements of operations, stockholders' equity and cash flows as of the end of and for such its most recently completed fiscal year, as well as related footnotes;

(iv) not later than the date on which a large accelerated filer with a fiscal year ending December 31 is required to file its Form 10-Q with the Securities and Exchange Commission, including any extension permitted by Rule 12b-25 of the Exchange Act, cause Lessee Guarantor (or, if there is no Lessee Guarantor, then Lessee) to furnish to Lessor such Person's unaudited consolidated balance sheet and related statements of operations, stockholders' equity and cash flows as of the end of and for its most recently completed fiscal quarter, as well as related footnote

(v) If Lessor has advised Lessee in writing that Lessor or Lessor Guarantor is required to file an Auditor's Report with respect to Lessee Guarantor's (or Lessee's) financial information delivered under Section 26.7(b)(iii), in filings to be made by Lessor or Lessor Guarantor with the Securities and Exchange Commission, then Lessee or Lessee Guarantor shall use commercially reasonable efforts to cause its auditor to provide the Auditor's Report at the sole cost and expenses of Lessor Guarantor (which costs and expenses will be reimbursed by Lessor and Lessor Guarantor within 30 days of receiving an invoice for such costs and expenses from Lessee or Lessee Guarantor); and

(vi) provide prompt written notice to Lessor of any failure by Lessee or Lessee Guarantor to pay any installment of principal or interest under its Material Debt when due if such failure is not cured or waived within the applicable grace period with respect to such Material Debt.

(c) Financial statements, reports and other information required or permitted to be furnished by Lessee or Lessee Guarantor pursuant to this Lease, including the financial statements and other information furnished pursuant to this <u>Section 26.7(b)(iii)</u> and (iv), may be submitted by Lessee or Lessee Guarantor by email addressed to Lessor. In the event that any information or documents furnished by Lessee or Lessee Guarantor pursuant to this Lease is publicly available on the Securities and Exchange Commission EDGAR database (or any successor database), such documents or information shall be deemed to have been delivered to Lessor by Lessee on the date on which Lessee Guarantor files such financial statements or other information with the Securities and Exchange Commission and provides notice to Lessor of such filing, which notice may be given by email. In addition, Lessor agrees that if the issuer of the financial statements required to be delivered by paragraphs (iii) and (iv) of this Section is not a large accelerated filer, or has a fiscal year other than December 31st, the Lessor and Lessor Guarantor will cooperate in good faith with the Lessee, including making appropriate required to the staff of the Securities and Exchange Commission.

(d) **Periodic Discussions.** In addition to the reports and statements to be delivered by Lessee pursuant to <u>Section 26.7(b)</u>, Lessor and Lessee agree, within fifteen (15) Business Days after the end of each calendar quarter during the Term, to convene a conference call with at least one Responsible Officer of each party to discuss the operations and financial performance of the Liquids Gathering System, the reports and statements delivered by Lessee pursuant to <u>Section 26.7(b)</u> and other relevant information regarding the Leased Property, provided, however, that the Responsible Officers of Lessee may delegate participation in such calls to other representatives of Lessee who are appropriate to participate in discussions of the operation and performance of the Liquids Gathering System.

26.8 Accord and Satisfaction.

(a) *Lessor*. Acceptance by Lessor of any partial payment of any amount payable by Lessee hereunder shall not constitute an accord and satisfaction by Lessor of any of Lessee's obligations hereunder and Lessor shall be entitled to collect from Lessee the balance of any amount remaining due.

(b) *Lessee*. Acceptance by Lessee of any partial payment of any amount due from Lessor hereunder shall not constitute an accord and satisfaction by Lessee of any of Lessor's obligations hereunder and Lessee shall be entitled to collect from Lessor the balance of any amount remaining due

2 6 . 9 **Prevailing Party**. If any action at law is necessary to enforce or interpret the terms of this Lease, the prevailing party shall be entitled to reasonable attorneys' fees and costs of the proceeding up to and including all trial and appellate levels in addition to any other relief to which it may be entitled. If any action in equity is necessary to enforce or interpret the terms of this Lease, the prevailing party shall be entitled to reasonable attorneys' fees and costs up to and including all trial and appellate levels in addition to any other relief to which it may be entitled. If any action in equity is necessary to enforce or interpret the terms of this Lease, the prevailing party shall be entitled to reasonable attorneys' fees and costs up to and including all trial and appellate levels in addition to any other relief to which it may be entitled. For purposes of this Section, a party will be considered to be the "prevailing party" if (a) such party initiated the litigation and substantially obtained the relief which it sought (whether by judgment, voluntary agreement or action of the other party, trial, or alternative dispute resolution process), (b) such party did not initiate the litigation and either (i) received a judgment in its favor, or (ii) did not receive judgment in its favor, but the party receiving the judgment was seeking.

26.10 Confidentiality.

General Confidentiality Provisions. Lessor shall, and shall cause any Person receiving Proprietary Information or other Confidential Information directly or (a) indirectly from Lessor, to hold all Proprietary Information and other Confidential Information in strict confidence. Subject to subj Proprietary Information and other Confidential Information only to the Lessor Parent, Lessor Lenders and prospective Lessor Lenders, holders of Lessor Equity Interests and prospective holders of Lessor Equity Interests, and any prospective purchaser of the Leased Property and, on a need to know basis, to their respective Affiliates, directors, officers, employees, accountants, legal counsel and other advisors who are involved in the administration, analysis and accounting for this Lease, and (ii) in the event Lessee is no longer the operator of the Leased Property, Lessor may disclose to a successor operator of the Liquids Gathering System and its advisors and consultants the Records and such other portions of the Proprietary Information and other Confidential Information as are necessary or prudent to permit them to safely and effectively manage and operate the Leased Property, provided that in the case of both subsections (i) and (ii), Lessor shall be responsible for the compliance by each such Person to whom disclosure is made with the confidentiality provisions of this Section 26.10 (including the acknowledgments and stipulations under subsection (c) of this Section), and Lessor shall remain liable for any breach of the provisions of this Section by any such Person. Notwithstanding the foregoing, Lessor shall not be responsible for any failure by Ross Avenue Investments, LLC, the limited partner of Pinedale Corridor, LP, to comply with the confidentiality requirements of this Section 26.10(a) so long as Ross Avenue Investments, LLC, is in full compliance with Exhibit D to the Equity Investor Agreement dated of even date with this Lease. Neither Lessor nor any Person to whom disclosure is made pursuant to this subsection (a) may disclose Proprietary Information or other Confidential Information to any other Person or entity except (A) to the extent required by any regulatory authority, (B) to the extent required by applicable laws or regulations or by subpoena or similar legal process, in each case after adequate notice to Lessee in order to allow Lessee to seek a protective order or other protection therefor, (C) with the prior written consent of Lessee, (D) to the extent such information becomes publicly available other than as a result of a breach of this Section, or (E) to the extent disclosure of such Proprietary Information or other Confidential Information by CorEnergy Infrastructure Trust, Inc. (or any successor Lessor Parent) is necessary or appropriate pursuant to the provisions of the federal securities laws or the rules or regulations promulgated thereunder. For the avoidance of doubt, nothing contained in this Section 26.10 shall limit the ability of Lessee or any of its Affiliates to disclose Proprietary Information or other Confidential Information.

(b) *Special Restrictions Regarding Proprietary Information*. Notwithstanding anything to the contrary set forth in this<u>Section 26.10</u> or elsewhere in this Lease, no Proprietary Information may, under any

circumstance at any time during this Lease or following its termination, be (i) disclosed to any Person who, at the time of such disclosure, is a Disqualified Person, or (ii) disclosed in any document, agreement or exhibit filed pursuant to the federal securities laws or any rules or regulations promulgated thereunder.

(c) Survival, Specific Performance and Equitable Remedies. The obligations under this <u>Section 26.10</u> of Lessor, and of all Persons to whom Confidential Information or Proprietary Information is disclosed, shall survive the expiration and termination of this Lease. Lessor acknowledges and stipulates (and all Persons to whom Confidential Information or Proprietary Information is disclosed shall be deemed to acknowledge and stipulate) that Lessee may suffer irreparable harm in the event of a breach of the provisions of this <u>Section 26.10</u> by Lessee, or by a disclosure of Confidential Information or Proprietary Information is disclosed have the right to obtain immediate injunctive or other equitable relief upon a breach of this <u>Section 26.10</u> by Lesser or any other Person to whom Confidential Information or Proprietary Information is disclosed.

26.11 **Consent of Lessor and Lessee**. Unless specified otherwise herein and except for consents or approvals for which a specific standard is expressly set forth herein (such as "not unreasonably withheld", "sole discretion", etc.) and specific provisions which describe the issues which may be considered when making or withholding approval (Permitted Lessee Transferee, etc.), Lessor's consent to any request of Lessee shall not be unreasonably withheld, or delayed and Lessee's consent to any request of Lessor shall not be unreasonably withheld, or delayed

26.12 **Permitted Lessee Contest**. Lessee shall not be required to pay any cost, expense or charge or perform any obligation so long as Lessee contests in good faith and at its own expense the amount or validity thereof by appropriate proceedings which shall operate to prevent the collection thereof or realization thereon and the sale, foreclosure or forfeiture of the Leased Property or any part thereof to satisfy the same, and Lessee shall have furnished any security as may be required in the applicable proceeding, and, pending any such proceedings, Lessor shall not have the right to pay or perform the same. In no event shall the manner in which Lessee pursues any such contest exacerbate in any material respect the risk to Lessor of civil or criminal liability, penalty or sanction, in addition to such risks as may exist for the matters that are the subject of such contest prior to such contest, and except for liabilities, penalties or sanctions for which Lessee may, and in fact does, post a bond. Further, the manner in which Lessee pursues any such contest and except for such contest prior to such contest shall not exacerbate in any material respect the risk to Lessor of defeasance of its interest in the Leased Property in addition to the risk of such defeasance as may exist for the matters that are the subject of such contest and except for such risk which Lessee may, and in fact does, bond around. Lessee shall use commercially reasonable efforts to diligently prosecute any such contest to a final conclusion, except that Lessee shall have the right to attempt to settle or compromise such to be payable thereon and/or fully perform any obligations to be performed thereon, together will all penalties, fines, interest, costs and expense resulting from such contest. Upon Lessee's request, Lessor shall prosecute such contest, if required by Applicable Legal Requirements, at no cost or expense to Lessor other than *de minimus* cost or expense. Upon reasonable request of Lessor at any time or from time-to-time, Lessee

26.13 **Waiver**. Failure of either party to complain of any act or omission by the other party, no matter how long the same may continue, shall not be deemed to be a waiver by the party of any of its rights hereunder. No waiver by either party at any time, whether express or implied, of any breach of any provision of this Lease shall be deemed a waiver of a breach of any other provision of this Lease or a consent to any subsequent breach of the same or any other provision. All rights and remedies which either party may have under this Lease or by law upon a breach hereunder shall be distinct, separate and cumulative and shall not

be deemed inconsistent with each other. Except as and to the extent this Lease provides that certain rights or remedies are limited or exclusive, no right or remedy, whether exercised by a party or not, shall be deemed to be in exclusion of any other right or remedy and any two (2) or more or all of such rights and remedies may be exercised at the same time, but without duplication of recovery for any such matter.

26.14 **Interpretation**. If any provision of this Lease or the application of any provision to any Person or any circumstance shall be determined to be invalid or unenforceable, then such determination shall not affect any other provision of this Lease or the application of such provision to any other Person or circumstance, all of which other provisions shall remain in full force and effect. It is the intention of the parties that if any provision of this Lease is capable of two constructions, one of which would render the provision void and the other of which would render the provision shall have the meaning which renders it valid. This Lease shall be interpreted and enforced without the aid of any canon, custom or rule of law requiring or suggesting construction against the party drafting or causing the drafting of the provision in question.

26.15 **No Derivative Liability**. Notwithstanding anything set forth in this Lease or otherwise, no direct or indirect (through tiered ownership or otherwise) advisor, trustee, director, officer, employee, beneficiary, shareholder, participant, partner, member, owner, investor, representative or agent of a party or its applicable Affiliates shall have any personal liability, directly or indirectly, under or in connection with this Lease, the Lessee Guaranty, the Resources Guaranty or the Lessor Guaranty or any amendment or amendments to any of the foregoing made at any time or times, heretofore or hereafter, and the other party and its successors and assigns and, without limitation, all other persons and entities, shall look solely to the assets of such party or its applicable Affiliates for the payment of any claim or for any performance, and each other party, on behalf of itself and its successors and assigns, hereby waive any and all such personal liability. Nothing in this <u>Section 26.15</u> is intended or shall be deemed to impair or limit the liability of any Person under any Lesser Guaranty, the Resources Guaranty or any Guaranty Agreement executed and delivered in connection with the Purchase Agreement or the liability of any Equity Investor or Lessor Parent under the Equity Investor Agreement.

26.16 **Successors and Assigns.** The words "*Lessor*" and "*Lessee*" and the pronouns referring thereto, as used in this Lease, shall mean where the context requires or admits, the Persons named herein as Lessor and as Lessee respectively, and (subject to the provisions hereof prohibiting certain Transfer, leases, subleases, assignments and Liens) their respective successors and assigns, irrespective of whether singular or plural, masculine, feminine or neuter. The agreements and conditions to be performed by Lessor shall be binding upon Lessor and its successors and assigns and shall inure to the benefit of Lessor and its successors and assigns.

26.17 **No Offer; Entire Agreement**. This Lease is transmitted for examination only and does not constitute an offer to lease and shall become effective only upon execution and unconditional delivery by Lessor and Lessee and delivery of the Lessor Guaranty, the Lessee Guaranty and the Resources Guaranty required hereby. This instrument, the Purchase Agreement, the SNDA and the Equity Investor Agreement contain the entire and only agreement between the parties, and no oral statements or representations or prior written matter not contained in this Lease or any such other documents shall have any force or effect. This Lease shall not be modified in any way except by a writing subscribed by both parties.

26.18 **Headings**. The headings for the various articles and sections of this Lease are used only as a matter of convenience for reference and are not to be considered a part of this Lease or used in determining the intent of the parties to this Lease.

26.19 **Counterparts**. This Lease may be executed in one or more counterparts, any one or all of which shall constitute one and the same instrument.

26.20 Governing Law; Venue; Service of Process; Waiver of Jury Trial.

(a) THIS LEASE AND ANY DISPUTES, CLAIMS OR CONTROVERSY ARISING OUT OF OR RELATING TO THIS LEASE (WHETHER SOUNDING IN CONTRACT OR TORT LAW) SHALL BE GOVERNED BY THE LAW OF THE STATE OF WYOMING WITHOUT REGARD TO CONFLICT OF LAW PRINCIPLES.

(b) EACH OF LESSOR AND LESSEE IRREVOCABLY AND UNCONDITIONALLY SUBMITS, FOR ITSELF AND ITS PROPERTY, TO THE NON-EXCLUSIVE JURISDICTION OF ANY WYOMING STATE COURT OR FEDERAL COURT OF THE UNITED STATES OF AMERICA SITTING IN THE STATE OF WYOMING AND ANY APPELLATE COURT FROM ANY THEREOF, AND EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY AND UNCONDITIONALLY AGREES THAT ALL CLAIMS IN RESPECT OF ANY SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN SUCH STATE OR, TO THE EXTENT PERMUTED BY LAW, IN SUCH FEDERAL COURT. EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW.

(c) EACH OF LESSOR AND LESSEE HEREBY IRREVOCABLY CONSENTS TO SERVICE OF PROCESS BY MAIL, PERSONAL SERVICE OR IN ANY OTHER MANNER PERMITTED BY APPLICABLE LEGAL REQUIREMENTS, AT THE ADDRESS SPECIFIED IN <u>Article XXIV</u> HEREOF.

(d) EACH OF LESSOR AND LESSEE HEREBY AGREES TO WAIVE ITS RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING HEREUNDER OR ANY DEALINGS BETWEEN THEM RELATING TO THE SUBJECT MATTER OF THIS LEASE, INCLUDING CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS AND ALL OTHER COMMON LAW AND STATUTORY CLAIMS. EACH PARTY HERETO (I) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, AND (II) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS LEASE BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION. EACH PARTY HERETO FURTHER WARRANTS AND REPRESENTS THAT IT HAS REVIEWED THIS WAIVER WITH ITS LEGAL COUNSEL AND THAT IT KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH LEGAL COUNSEL.

26.21 Time of the Essence. TIME IS OF THE ESSENCE AS TO ANY ACT TO BE PERFORMED WITHIN A SPECIFIC TIME FRAME UNDER THIS LEASE.

26.22 Estoppel Certificates.

(a) Lessee agrees upon not less than fifteen (15) days prior notice by Lessor or any Lessor Lender to execute, acknowledge and deliver to Lessor or the Lessor Lender a statement in writing by Lessee in substantially the form of <u>Exhibit D</u>. If Lessee is required to provide such certificate more than twice in any twelve (12) month period, Lessor shall pay Lessee a processing fee of \$500 for each additional certificate requested in such twelve (12) month period.

(b) Lessor agrees upon not less than fifteen (15) days prior notice by Lessee to execute, acknowledge and deliver to Lessee or to such party as Lessee may designate a statement in writing by Lessor

similar to the form of Exhibit D (with appropriate changes to reflect that it is being signed by Lessor). If Lessor shall be required to provide such certificate more than twice in any twelve (12) month period, Lessee shall pay Lessor a processing fee of \$500 for each additional certificate requested in such twelve (12) month period.

ARTICLE XXVII. OTHER AGREEMENTS OF LESSOR

27.1 Special Purpose Entity; Lessor's Governing Documents. At all times during the Term, Lessor covenants and agrees as follows:

(a) **SPE**. Lessor shall at all times be a Special Purpose Bankruptcy Remote Entity.

(b) Controlling Lease Rights. (i) No Disqualified Person shall possess, and Disqualified Persons shall not possess, Controlling Lease Rights, and (ii) so long an Ultra Entity Person is the Lessee, no Person other than a Permitted Controller of Lease Rights shall possess Controlling Lease Rights.

(c) Notice of Transfer; Ownership by Disqualified Persons.

(i) Notice shall be given to Lessee of any issuance by Lessor of any Equity Investor Interests after the date of this Lease. Such notice shall be given not less than fifteen (15) Business Days prior to such issuance (ten (10) Business Days in the case of issuance to a Lessor Equity Interest Owner) and shall contain the information that is required for notices that are given pursuant to Section 17.5(c)(ii).

(ii) Disqualified Persons, in the aggregate, shall not have beneficial ownership of twenty-five percent (25%) or more of the Lessor Equity Interests.

(d) Compliance with Governing Documents. Lessor and its Equity Investors shall at all times comply with Lessor's Governing Documents

(e) Compliance of Governing Documents with Lease and Other Requirements Lessor's Governing Documents shall at all times contain the provisions required by Part B of Schedule 27.1 (each a "Lessor Governing Document Required Term" and collectively, the "Lessor Governing Document Required Terms").

(f) *Amendment of Governing Documents*. Lessor shall not permit any Lessor Governing Document Required Term to be amended, supplemented, restated or terminated without Lessee's prior written consent.

(g) *Delivery of Governing Document Amendments.* Lessor shall deliver or cause to be delivered to Lessee copies of any proposed amendments, terminations, supplements or restatements of any of its Governing Documents at least ten (10) Business Days prior to the execution thereof.

(h) Equity Investor Agreement. Each Lessor and each direct owner of a Lessor Equity Interest shall at all times be a party to the Equity Investor Agreement.

27.2 Additional Lessor Covenants. At all times during the Term, Lessor covenants and agrees as follows:

(a) *Compliance with Lessor Loan Document Criteria.* Lessor agrees that all times (i) the Lessor Loan Documents shall include each Required Loan Document Provision, and (ii) the Lessor Loan Documents shall not include any Prohibited Loan Document Provision. Lessor shall not enter into or permit

any new Lessor Loan Document or any Lessor Loan Document Amendment (i) in violation of this Section 27.2(a), or (ii) without Lessee's consent if such consent is required by subsection (d), below.

(b) Notice of Lessor Loan Default. Lessor shall promptly advise Lessee in writing of any breach, default or failure of performance under any Lessor Loan Document, and immediately provide Lessee a copy of any written notice of default or intent to enforce remedies given by or on behalf of any Lessor Lender with respect to any Lessor Indebtedness or under any Lessor Loan Document.

(c) *Notice of New Loan Documents and Loan Amendments.* So long as an Ultra Entity Person is the Lessee, Lessor shall deliver to Lessee not less than ten (10) Business Days prior to execution of same, a copy of each proposed new Lessor Loan Document and each proposed Lessor Loan Document Amendment, and within five (5) Business Days following execution of same, a copy of each such executed new Lessor Loan Document and Lessor Loan Document Amendment.

(d) *Lessee Consent to New Loan Documents and Loan Amendments.* So long as an Ultra Entity Person is the Lessee, Lessee's prior written consent shall be required for any new Lessor Loan Document after the Effective Date and for any Lessor Loan Document Amendment after the Effective Date if, as a result of such new Lessor Loan Document or Lessor Loan Document Amendment, the resulting Lessor Loan Documents either (i) fail to include any Required Loan Document Provision, or (ii) include any Prohibited Loan Document Provision. Except as provided in the immediately preceding sentence, Lessee consent is not required for any new Lessor Loan Document or Lessor Loan Document Amendment after the Effective Date.

(e) *Quarterly Information Regarding Calculation of Maximum Amount.* Within forty-five (45) days after the end of each calendar quarter and within eighty (80) days after the end of each calendar year, Lessor shall provide to Lessee a calculation of the Maximum Amount as of the end of such calendar quarter, together reasonably detailed supporting information with respect to such calculation.

(f) *Maintenance of Existence*. Lessor shall at all times (i) maintain its existence in good standing under the laws of the State of its incorporation, (ii) be qualified to do business in the State of Wyoming, and (iii) except to the extent it is prohibited by this Lease from doing so without Lessee's consent and Lessee fails to give such consent, maintain and renew all of its respective rights, powers, privileges and franchises except where the failure to do so a material adverse effect on Lessor's ability to perform its obligations under this Lease or on Lessee's ability to possess and operate the Leased Property in accordance with the terms and conditions of this Lease.

(g) Lessor Compliance with Law; Permitted Lessor Contests Subject to Section 11.1 hereof, Lessor shall comply with all Applicable Legal Requirements relating to Lessor except where (i) Lessee is contesting such Applicable Legal Requirements in accordance with a Permitted Lessee Contest, (ii) such compliance could reasonably be expected to impair in any material respect Lessee's use or operation of the Leased Property unless Lessee consents to such compliance, which consent shall not be unreasonably withheld, delayed or conditioned, or (iii) such failure to do so results from a Lessee Event of Default. For purposes hereof, "Permitted Lessor Contest' means (but subject to Section 11.1) a Lessor contest of (A) an Applicable Legal Requirement, or (B) a Lien on Lessor's right, title or interest in this Lease, in either case in good faith and at its own expense by appropriate proceedings which shall operate to prevent the immediate application of such Applicable Legal Requirements or Lien and the sale, foreclosure or forfeiture of the Leased Property or the Lessor's interest in this Lease or Lessee's leasehold estate therein or hereunder or Lessee's use or operation of the Leased Property, or any part thereof, and Lessor Contest be undertaken in violation of <u>Article XI</u>. Further, in no event shall the manner in which Lessor pursues any such contest exacerbate in any material respect the risk to Lessee of civil or criminal liability, penalty or sanction, in addition to such risks as may

exist for the matters that are the subject of such contest prior to such contest, and except for liabilities, penalties or sanctions for which Lessor may, and in fact does, post a bond. Further, the manner in which Lessor pursues any such contest shall not exacerbate in any material respect the risk to Lessor of defeasance of its interest in the Leased Property or impair its use or operation thereof in addition to the risk of such defeasance or impairment as may exist for the matters that are the subject of such contest prior to such contest and except for such risk which Lessor may, and in fact does, bond around. Lessor shall use commercially reasonable efforts to diligently prosecute any such contest to a final conclusion, except that Lessor shall have the right to attempt to settle or compromise such contest through negotiations and to discontinue any such contest at any time. Lessor shall promptly after the final determination of such contest, fully pay any amounts determined to be payable thereon and/or fully perform any obligations to be performed thereon, together will all penalties, fines, interest, costs and expenses resulting from such contest. Lessor shall prosecute any such Permitted Lessor Contest at no cost or expense to Lessee other than *de minimus* costs and expenses. Upon reasonable request of Lessee at any time or from time-to-time, Lessor shall provide a written report to Lessor regarding the status of any such contests.

(h) **Debtor Release Laws**. Lessor shall not voluntarily take any action that shall, or cause any action to be taken that is intended to, submit Lessor, as debtor, to any proceeding under any Applicable Legal Requirements involving bankruptcy, insolvency, reorganization or other laws affecting the rights of creditors generally.

[Remainder of Page Intentionally Blank Signature Pages Follow]



IN WITNESS WHEREOF, the parties hereto have caused this Lease to be duly executed as of the date first above written.

LESSOR:

PINEDALE CORRIDOR, LP,

a Delaware limited partnership

By:Pinedale GP, Inc., a Delaware corporation, its sole general partner

By:/s/ Richard C. Green Richard C. Green, Chairman LESSEE:

ULTRA WYOMING LGS, LLC, a Delaware limited liability company

By:/s/ Marshall D. Smith

Marshall D. Smith, Senior Vice President and Chief Financial Officer

Exhibit 10.2
EXECUTION VERSION

PINEDALE CORRIDOR, LP

a Delaware Limited Partnership

FIRST AMENDED AND RESTATED

LIMITED PARTNERSHIP AGREEMENT

Dated as of December 20, 2012

TABLE OF CONTENTS

		Page
ARTICLE 1	DEFINITIONS	1
1.1	Definitions	1
1.2	Interpretation	10
ARTICLE II	FORMATION OF THE PARTNERSHIP	10
2.1	Formation; Governmance by the Act	10
2.2	Name and Place of Business	10
2.3	Filings and Registrations	10
2.4	Term	10
2.5	Organizational Expenses and LGS Transaction Expenses	10
2.6	Partner Representations and Warranties	11
2.7	Special Purpose Bankruptcy Remote Entity Provisions	11
2.8	LGS Lease and LGS Equity Investors' Agreement; Indemnification	11
2.9	Indemnification	11
ARTICLE III	PURPOSE	12
3.1	Business Purpose	12
3.2	Powers	12
ARTICLE IV	CAPITAL CONTRIBUTIONS; PARTNERSHIP INTERESTS	12
4.1	Nominal Initial Capital Contributions	12
4.2	Initial Capitalization	12
4.3	Additional Capital Contributions	13
4.4	Capital Accounts	13
4.5	No Right to Return of Capital Contributions; No Priority	13
4.6	Authority to Issue New Units; Participation Rights	14
4.7	Partner Loans	14
4.8	Conversion of General Partnership Interests	14
4.9	Unit Certificates	15
ARTICLE V	COMPENSATION OF GENERAL PARTNER	15
5.1	Reimbursement of Out-of-Pocket Costs	15
5.2	Management Fee	15
ARTICLE VI	ALLOCATIONS AND DISTRIBUTIONS	15
6.1	Non Liquidation Cash Distributions	15
6.2	Liquidation Distributions	15
6.3	Profits, Losses and Distributive Shares of Tax Items	16
6.4	Allocation of Income, Loss and Credits	16
6.5	Special Rules Regarding Allocation of Tax Items	16
6.6	No Priority	19
0.0	Normoliky	17

i

ARTICLE VII MANAGEMENT AND CONTROL 7.1 Management by the General Partner 7.2 Rights, Powers and Authority of General Partner 7.3 Actions Requiring a Class B Limited Partner Majority 7.4 Advisory Board; Observer Rights 7.5 Other Businesses of Partners 7.6 Exculpation and Indemnification 7.7 Partnership Assets to be Used First 7.0 N. D. Control Internet Partner	19 19 19 20 22 23 23 23 24
7.2Rights, Powers and Authority of General Partner7.3Actions Requiring a Class B Limited Partner Majority7.4Advisory Board; Observer Rights7.5Other Businesses of Partners7.6Exculpation and Indemnification7.7Partnership Assets to be Used First	19 20 22 23 23 23 24
7.3Actions Requiring a Class B Limited Partner Majority7.4Advisory Board; Observer Rights7.5Other Businesses of Partners7.6Exculpation and Indemnification7.7Partnership Assets to be Used First	20 22 23 23 23 24
7.4Advisory Board; Observer Rights7.5Other Businesses of Partners7.6Exculpation and Indemnification7.7Partnership Assets to be Used First	22 23 23 23 24
7.4Advisory Board; Observer Rights7.5Other Businesses of Partners7.6Exculpation and Indemnification7.7Partnership Assets to be Used First	23 23 24
7.5Other Businesses of Partners7.6Exculpation and Indemnification7.7Partnership Assets to be Used First	23 24
7.7 Partnership Assets to be Used First	24
7.8 No Preferential Transactions with Partners	24
ARTICLE VIII ACCOUNTING AND BANK ACCOUNTS	24
8.1 Books and Records	24
8.2 Taxable and Fiscal Year	25
8.3 Reporting	25
8.4 Tax Returns and Elections; Tax Matters Partner	25
8.5 Section 754 Election	26
8.6 Bank Accounts	26
ARTICLE IX LIMITED PARTNERS	26
9.1 Rights of Limited Partners	26
9.2 Limitation of Limited Partners' Liability	26
9.3 Power of Attorney	26
9.4 Changes in Addresses	27
9.5 General Partner as Limited Partner	27
ARTICLE X TRANSFERS OF PARTNERSHIP INTERESTS BY LIMITED PARTNERS	27
10.1 General Restrictions	27
10.2 Permitted Transfers	28
10.3 Substitute Limited Partners	28
10.4 Effect of Bankruptcy, Death, Incompetency or Termination of a Limited Partner	28
10.5 Effect of Admission as a Substitute Limited Partner	29
10.6 Withdrawal of a Limited Partner	29
10.7 Right of First Bid	29
10.8 Tag-Along Right in Favor of Class B Limited Partnership Interests	30
ARTICLE XI CHANGES IN THE GENERAL PARTNER AND TRANSFERS OF GENERAL PARTNERSHIP INTERESTS	31
11.1 General Restrictions	31
11.2 Permitted Transfers	32
11.3 Withdrawal or Removal of a General Partner	32
11.4 Effect of Withdrawal or Removal of a General Partner	32
11.5 Additional or Successor General Partners	34
11.6 Changes in General Partners	35

ii

ARTICLE XII	DISSOLUTION AND TERMINATION	35
12.1	Events Causing Dissolution	35
12.2	Effect of Dissolution	35
12.3	Application of Proceeds	36
ARTICLE XIII	DISPUTE RESOLUTION	36
13.1	Disputes	36
13.2	Negotiations to Resolve Disputes	36
13.3	Arbitration	36
13.4	Survival	37
ARTICLE XIV	MISCELLANEOUS	37
14.1	Meetings	37
14.2	Title to Assets	38
14.3	Nature of Interest in the Partnership	38
14.4	Notices	38
14.5	Third Party Beneficiaries	38
14.6	Entire Agreement	38
14.7	Amendments	38
14.8	Governing Law	38
14.9	Severability	39
14.10	Binding Agreement	39
14.11	Headings	39
14.12	Counterparts; Electronic Delivery	39

iii

PINEDALE CORRIDOR, LP

FIRST AMENDED AND RESTATED LIMITED PARTNERSHIP AGREEMENT

THIS FIRST AMENDED AND RESTATED LIMITED PARTNERSHIP AGREEMENT of Pinedale Corridor, LP (the "*Partnership*") is made and entered into as of December 20, 2012 (the "*Effective Date*") by and among the Partnership, Pinedale GP, Inc., a Delaware corporation (the "*General Partner*") and each Person set forth from time to time on <u>Exhibit A</u> attached hereto, as the Limited Partners. The General Partner and the Limited Partners are sometimes collectively referred to herein as the "*Partners*" or, individually, as a "*Partner*."

RECITALS

A. The General Partner formed the Partnership by filing the Certificate with the Secretary of State of the State of Delaware on December 3, 2012 (the 'Formation Date').

B. The General Partner and the Initial Limited Partner entered into the Limited Partnership Agreement for the Partnership (the 'Original Partnership Agreement') on the Formation Date.

C. The parties hereto desire to associate themselves together through the Partnership for the conduct of the Business, and, in connection therewith, to amend and restate the Original Partnership Agreement in accordance with the terms and conditions of this Agreement.

NOW, THEREFORE, in consideration of the mutual agreements, covenants, and undertakings herein contained, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE I

DEFINITIONS

1.1 <u>Definitions</u>. As used herein, the following terms shall have the following meanings unless the context otherwise specifies:

"AAA" has the meaning set forth in Section 13.3(a).

"AAA Rules" has the meaning set forth in Section 13.3(a).

"Acceptance Notice" has the meaning set forth in Section 10.7(c).

"Accountant" means the firm of independent public accountants engaged by the Partnership from time to time.

"Act" means the Delaware Revised Uniform Limited Partnership Act, as amended from time to time.

"Adjusted Capital Account Deficit" means, with respect to any Partner, the deficit balance, if any, in such Partner's Capital Account as of the end of the relevant fiscal year, after giving effect to

the following adjustments: (i) increased for any amounts such Partner is unconditionally obligated to restore and the amount of such Partner's share of Partnership Minimum Gain and Partner Minimum Gain after taking into account any changes during such year; and (ii) reduced by the items described in Treasury Regulation 1.704-1(b) (2)(ii)(d)(4), (5) and (6).

"Affiliate" shall have the meaning ascribed to such term in Rule 12b-2 of the General Rules and Regulations under the Exchange Act. However, for purposes of this Agreement, the Partnership shall not be deemed to be an Affiliate of any of the Partners.

"Agreement" means this Agreement of Limited Partnership of Pinedale Corridor, LP, as amended from time to time.

"Arbitration Notice" has the meaning set forth in Section 13.3(a).

"Assignee" means a Person to whom a Limited Partnership Interest has been transferred, by transfer, assignment, or otherwise, in a manner permitted under this Agreement, but who has not become a Substitute Limited Partner.

"Available Cash" means the gross cash proceeds of the Partnership less any amount used by the Partnership to pay, or to establish reserves for the payment by the Partnership of, operating costs, capital expenditures, debt payments, and any other expenditures or contingencies of the Partnership, all as determined by the General Partner.

"Bankruptcy Action" means, with respect to any Person, if such Person:

- (a) makes an assignment for the benefit of creditors,
- (b) files a voluntary petition in bankruptcy,
- (c) is adjudged a bankrupt or insolvent, or has entered against it an order for relief, in any bankruptcy or insolvency proceedings,

(d) consents to, orchestrates or participates in the orchestration of, or files a petition or answer seeking for itself, any reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any statute, law or regulation,

(c) files an answer or other pleading admitting or failing to contest the material allegations of a petition filed against it in any bankruptcy or insolvency proceeding,

(f) seeks, consents to, orchestrates or participates in the orchestration of, or acquiesces in the appointment of a trustee, receiver, liquidator, sequestrator, custodian or any similar official of or for such Person or of all or any substantial part of its properties,

(g) fails to cause the dismissal of any proceeding against such Person seeking reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any statute, law or regulation within one hundred twenty (120) days after the commencement of such proceeding,

(h) fails to cause the vacation or stay of any appointment of a trustee, receiver or liquidator of such Person or of all or any substantial part of its properties made without such Person's consent or acquiescence, within ninety (90) days after such appointment and, if stayed within such time period, the vacation of such appointment within ninety (90) days after the expiration of such stay, or

(i) takes any action in furtherance of any of the foregoing.

"Beneficial Owner" a Person shall be deemed the "Beneficial Owner" of, and shall be deemed to "beneficially own" and have "beneficial ownership of" any equity interests or other securities:

(a) which such Person or any of such Person's Affiliates, directly or indirectly, has the right to acquire (whether such right is exercisable immediately or only after the passage of time) pursuant to any agreement, arrangement or understanding (whether or not in writing) or upon the exercise of conversion rights, exchange rights, rights, warrants or options, or otherwise;

(b) which such Person or any of such Person's Affiliates, directly or indirectly, has the right to vote or dispose of or has "beneficial ownership" of (as determined pursuant to Rule 13d-3 of the General Rules and Regulations under the Exchange Act), including pursuant to any agreement, arrangement or understanding, whether or not in writing; <u>provided, however</u>, that a Person shall not be deemed the "Beneficial Owner" of, or to "beneficially own," any equity interest or other security under this subparagraph (b) as a result of an agreement, arrangement or understanding to vote such equity interest or other security if such agreement, arrangement or understanding: (i) arises solely from a revocable proxy given in response to a public proxy or consent solicitation made pursuant to, and in accordance with, the applicable provisions of the General Rules and Regulations under the Exchange Act, and (ii) is not also then reportable by such Person on a Schedule 13D under the Exchange Act (or any comparable or successor report); or

(c) which are "beneficially owned," directly or indirectly, by any other Person (or any Affiliate thereof) with which such Person (or any of such Person's Affiliates) has any agreement, arrangement or understanding (whether or not in writing), for the purpose of acquiring, holding, voting (except pursuant to a revocable proxy as described in the proviso to subparagraph (b)) or disposing of any voting equity interests or other securities of the Company;

provided, however, that (x) a Person engaged in business as an underwriter of securities shall not be deemed the "Beneficial Owner" of or to "beneficially own" any equity interests or other securities acquired through such Person's participation in good faith in a firm commitment underwriting and (y) the existence of rights of first bid, pre-emptive rights, drag along rights, or tag along rights set forth in this Agreement shall not themselves cause one Person to be deemed the Beneficial Owner of Company Equity Interests held by one or more other Partners.

"Bid" has the meaning set forth in Section 10.7(b).

"Bid Price" has the meaning set forth in Section 10.7(b).

"Bid Terms" has the meaning set forth in Section 10.7(b).

"Business" has the meaning set forth in Section 3.1.

"Capital Account" means the separate account established and maintained for each Partner by the Partnership pursuant to Section 4.4.

"Capital Contribution" means, with respect to a Partner, the total amount of cash and the Fair Value of property contributed by such Partner (or such Partner's predecessor in interest) to the capital of the Partnership for such Partner's Partnership Interest (which aggregate net amount is set forth opposite the name of such Partner on Schedule A attached hereto).

"Certificate" means the Certificate of Limited Partnership of the Partnership duly filed in accordance with the Act, as amended from time to time.

"Class A Limited Partnership Interests" means, at any time, the Limited Partnership Interests, if any, into which General Partnership Interests have been converted pursuant to Section 4.8 and as may otherwise be issued from time to time, without regard to the identity of the Limited Partner or Limited Partners holding such Limited Partnership Interests at such time.

"Class B Limited Partner Majority" means the affirmative vote or consent of Limited Partners holding a majority of the outstanding Units representing Class B Limited Partnership Interests held at the relevant time by all Limited Partners other than Limited Partners that are Affiliates of the General Partner.

"Class B Limited Partnership Interests" means, at any time, the Limited Partnership Interests described on <u>Schedule A</u> attached hereto as being held by Class B Limited Partners, as such schedule is in effect on the Effective Date and may be amended from time to time, without regard to the identity of the Limited Partner or Limited Partners holding such Limited Partnership Interests at such time.

"Code" means Title 11 of the United States Code, 11 U.S.C. Section 101 et seq., as amended.

"Company Equity Interests" means the equity interests issued by the Partnership, including the General Partnership Interests, the Limited Partnership Interests, or any other interest which would be classified as equity on the Partnership's balance sheet under generally accepted accounting principles and any right to acquire such equity interests in the Partnership, including upon conversion or exercise of securities. References herein to a percentage of Company Equity Interests means Company Equity Interests which confer on the holder thereof the right to receive the applicable percentage of dividends, distributions, including liquidating distributions, or other payments with respect to the Company Equity Interests.

"Credits" means all investment tax credits or other tax credits allowed by the Code with respect to the Partnership Properties or the activities of the Partnership.

"Debf" means (a) any indebtedness for borrowed money or the deferred purchase price of property as evidenced by a note, bonds, or other instruments, (b) obligations as lessee under capital leases, (c) liabilities for the deferred purchase price of property (excluding accounts payable arising in the ordinary course of business but including all liabilities created or arising under any conditional sale or other title retention agreement with respect to any such property), (d) all liabilities which

would appear on a balance sheet prepared in accordance with generally accepted accounting principles in respect of Synthetic Leases assuming such Synthetic Leases were accounted for as capital leases, (e) all liabilities for borrowed money secured by any lien with respect to any property owned (whether or not the owner has assumed or otherwise become liable for such liabilities), (f) all liabilities in respect of letters of credit or instruments serving a similar function issued or accepted by banks and other financial institutions (whether or not representing obligations for borrowed money), and (g) obligations under guarantees of indebtedness or obligations of the kinds referred to in clause (a) through (f) of this definition.

"Dispute" has the meaning set forth in Section 13.1.

"Disqualified Person" has the meaning ascribed to such term in the LGS Equity Investors' Agreement on the date hereof.

"Distributions" means any distributions by the Partnership to the Partners of cash, property, Liquidation Proceeds or other amounts.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Fair Value" of an asset means its fair market value, as reasonably determined by the General Partner (unless otherwise specified herein).

"Former General Partner" has the meaning set forth in Section 11.4(c).

"General Partner" means Pinedale GP, Inc., a Delaware corporation, and any of its successors and permitted assigns that is admitted to the Partnership as a general partner of the Partnership, in each such Person's capacity as a general partner of the Partnership.

"General Partner Parties" has the meaning set forth in Section 7.6(a).

"General Partnership Interest" means the interest in the Partnership held by a General Partner in its capacity as a general partner of the Partnership.

"Governmental Authority" means any foreign, domestic, federal, territorial, state or local governmental authority, court, commission, board, bureau, agency, or instrumentality, or any regulatory, administrative or other department, agency, or any political or other subdivision, department, or branch of any of the foregoing.

"Income" and "Loss" mean, respectively, for each fiscal year or other period, an amount equal to the Partnership's taxable income or loss for such year or period, determined in accordance with Code Section 703(a), except that for this purpose (i) all items of income, gain, deduction or loss required to be separately stated by Code Section 703(a)(1) shall be included in taxable income or loss; (ii) tax exempt income shall be added to taxable income or loss; (iii) any expenditures described in Code Section 705(a)(2) (B) (or treated as Code Section 705(a)(2)(B) expenditures pursuant to Treasury Regulation § 1.704-1(b)(2)(iv)(i)) and not otherwise taken into account in computing taxable income or loss shall be subtracted; and (iv) taxable income or loss shall be adjusted to reflect any item of income, gain, loss or deduction specifically allocated in Article VI.

"Initial Limited Partner" means TTO.

"Leased Property" has the meaning set forth in the LGS Lease, and includes (i) the Liquids Gathering System (as defined in the LGS Lease) and the Personal Property (as defined in the LGS Lease) and (ii) any Permitted Capital Improvements and Additional Lines (as defined in the LGS Lease), Parts (as defined in the LGS Lease), and other assets or properties that are included from time to time in the Leased Property in accordance with the Lease, in all cases irrespective of whether, at the relevant time, the Leased Property is leased by the Partnership to Ultra Newco (or a successor lessee) under the LGS Lease.

"LGS Equity Investors' Agreement' means that certain Equity Investors' Agreement, dated as of the Effective Date, among the Partnership, Ultra Newco, Pinedale GP, Inc., TTO, and Ross Avenue Investments, a copy of which is attached hereto as Exhibit B.

"LGS Financing Documents" means (i) that certain Term Credit Agreement, dated as of December 7, 2012, among the Partnership (as Borrower), KeyBank National Association (as Lender and Agent), the other lender parties thereto from time to time, KeyBanc Capital Markets and Bank of America Merrill Lynch, (as Co-Arrangers, Co-Syndication Agents, and Co-Documentation Agents), KeyBanc Capital Markets (as Sole Book Runner), and the other lender parties thereto from time to time (the "Credit Agreement"), (ii) all ancillary agreements, instruments, certificates, and other documents under which the Partnership's obligations under the Credit Agreement are secured, or that are otherwise delivered in connection with or pursuant to the Credit Agreement, and (iii) all agreements, instruments, certificates, and other documents under which the partnership's obligations under the documents that amend or replace any of the foregoing, pursuant to a refinancing transaction or otherwise.

"LGS Lease" means that certain Lease, dated as of the Effective Date, between the Partnership and Ultra Newco.

"LGS Purchase Agreement" means that certain Purchase and Sale Agreement, dated as of December 7, 2012 (as amended December 12, 2012) between the Partnership and Ultra Wyoming.

"LGS Transaction" means the transactions contemplated by the LGS Purchase Agreement, the LGS Lease, and the LGS Financing Documents. For clarity, the LGS Transaction shall not include any issuance of equity by TTO to fund its Capital Contributions.

"LGS Transaction Documents" means the LGS Purchase Agreement, the LGS Lease, the LGS Financing Documents, the LGS Equity Investors' Agreement, and any other agreement, instrument, or other document to which the Partnership is a party or by which the Partnership is bound in connection with the LGS Transaction, as any such agreements, instruments, or documents may be amended from time to time.

"Lien" means any mortgage lien, deed of trust lien, vendor's lien, security interest, mechanic's or materialman's lien, or other lien or security interest.

"Limited Partner" means each Person executing this Agreement as a limited partner and whose name is set forth from time to time on Exhibit A attached hereto, including any Substitute Limited Partner, in each such Person's capacity as a limited partner of the Partnership.

"Limited Partner Parties" has the meaning set forth in Section 7.6(a).

"Limited Partnership Interest" means the interest in the Partnership held by a Limited Partner in its capacity as a limited partner of the Partnership.

"Liquidation Proceeds" means all Partnership Properties at the time of liquidation of the Partnership and all proceeds thereof.

"Management Fee" has the meaning set forth in Section 5.2.

"Nonrecourse Deduction" shall have the same meaning as nonrecourse deductions set forth in Treasury Regulation § 1.704-2. Generally, the amount of Nonrecourse Deductions for a fiscal year equals the net increase in the amount of Partnership Minimum Gain (determined in accordance with Treasury Regulation § 1.704-2(d)) during such year reduced (but not below zero) by the aggregate distributions made during the year of proceeds of a Nonrecourse Liability that are allocable to an increase in Partnership Minimum Gain, determined according to the provisions of Treasury Regulation § 1.704-2(c) and (h).

"Nonrecourse Liability" means a Partnership liability with respect to which no Partner bears the economic risk of loss as determined under Treasury Regulation § 1.752-1(a)(2) and Treasury Regulation § 1.752-2.

"Partner" or "Partners" has the meaning set forth in the Preamble.

"Partner Loans" has the meaning set forth in Section 4.7.

"Partner Minimum Gain" shall have the same meaning as partner nonrecourse debt minimum gain as set forth in Treasury Regulation § 1.704-2(i)(3). With respect to each Partner Nonrecourse Debt, Partner Minimum Gain shall be determined by computing for each Partner Nonrecourse Debt any gain which the Partnership would realize if the Partnership disposed of the property subject to that liability for no consideration other than full satisfaction of such liability. For purposes of computing gain, the Partnership shall use the basis of such property which is used for purposes of maintaining Capital Accounts under Section 4.4. In any taxable year in which a Revaluation occurs, the net increase or decrease in Partner Minimum Gain for such taxable year shall be determined by: (1) calculating the net decrease or increase in Partner Minimum Gain using the current year's book value and the prior year's amount of Partner Minimum Gain, and (2) adding back any decrease in Partner Minimum Gain arising solely from the Revaluation.

"Partner Nonrecourse Debt" shall have the same meaning as partner nonrecourse debt set forth in Treasury Regulation § 1.704-2(b)(4).

"Partner Nonrecourse Deductions" shall have the same meaning as partner nonrecourse deductions set forth in Treasury Regulation § 1.704-2(i)(2). Generally, the amount of Partner Nonrecourse Deductions with respect to a Partner Nonrecourse Debt for a fiscal year equals the net increase during the year in the amount of the Partner Minimum Gain (determined in accordance with Treasury Regulation § 1.704-2(i)) reduced (but not below zero) by the aggregate distributions made during the year of proceeds of a Partner Nonrecourse Debt and allocable to the increase in Partner Minimum Gain, determined according to the provisions of Treasury Regulation § 1.704-2(i).

"Partnership" has the meaning set forth in the Preamble.

"Partnership Interest" refers to all of a Partner's rights and interests in the Partnership as provided in this Agreement or under the Act, including, without limitation, such Partner's interest in the total capital, profits, and losses of the Partnership, the right to vote on any matter as a Partner, or the right to access the books and records, or other information in the possession, of the Partnership.

"Partnership Minimum Gain" shall have the same meaning as partnership minimum gain set forth in Treasury Regulation § 1.704-1(d)(1). Partnership Minimum Gain shall be determined, first, by computing for each Partnership Nonrecourse Liability any gain which the Partnership would realize if the Partnership disposed of the property subject to that liability for no consideration other than full satisfaction of such liability and, then, aggregating the separately computed gains. For purposes of computing gain, the Partnership shall use the basis of such property which is used for purposes of maintaining Capital Accounts. In any taxable year in which a Revaluation occurs, the net increase or decrease in Partnership Minimum Gain for such taxable year shall be determined by: (1) calculating the net decrease or increase in Partnership Minimum Gain, and (2) adding back any decrease in Partnership Minimum Gain arising solely from the Revaluation.

"Partnership Properties" means all properties and assets (whether real, personal, or mixed) that the Partnership may own or otherwise have an interest from time to time.

"Percentage Interest" means, with respect to a Partner, the number of Units held by such Partner divided by the total number of Units representing all Partnership Interests.

"Person" means any individual, firm, corporation, limited liability company, partnership or other entity.

"Power of Attorney" means the power of attorney described and granted in Section 9.3.

"Purchase Date" has the meaning set forth in Section 11.4(d).

"Purchaser" has the meaning set forth in Section 11.4(d).

"Purchasing Partner" and "Purchasing Partners" have the meanings set forth in Section 10.7(d).

"*Revaluation*" shall mean the occurrence of any event described in clause (x), (y) or (z) of Section 4.4 of this Agreement in which the book basis of Partnership Property is adjusted to its Fair Value.

"Ross Avenue Investments" means Ross Avenue Investments, LLC, a Delaware limited liability company.

"Substitute Limited Partner" means a Person admitted to the Partnership as a Limited Partner pursuant to Section 10.3.

"Subject Interest" has the meaning set forth in Section 10.8.

"Synthetic Lease" means, at any time, any lease (including leases that may be terminated by the lessee at any time) of any property (a) that is accounted for as an operating lease under generally accepted accounting principles and (b) in respect of which the lessee retains or obtains ownership of the property so leased for United States federal income tax purposes, other than any such lease under

which such Person is the lessor; but for clarity not including the LGS Lease (which the parties expect to be treated as a "true lease" for United States federal income tax purposes).

"Tag-Along Election Notice" has the meaning set forth in Section 10.8.

"Tag-Along Interest" has the meaning set forth in Section 10.8.

"Tag-Along Right Holder" has the meaning set forth in Section 10.8.

"Tag-Along Transfer Notice" has the meaning set forth in Section 10.8.

"*Transfer*" means any sale, charge, gift, pledge, encumbrance, mortgage, transfer or any other disposition of Company Equity Interests (or any interest therein) whatsoever, whether voluntary or involuntary. A Transfer shall be deemed to be involuntary if it involves any transaction, proceeding or action by or in which the Partner or other Beneficial Owner of Company Equity Interests shall be involuntarily deprived or divested of any right, title or interest in or to any of the Company Equity Interests (including, without limitation, any seizure under levy of attachment or execution, transfer in connection with bankruptcy or other court proceeding to a trustee or receiver or other officer or agency or any transfer to a state or to a public officer or agency pursuant to any statute pertaining to escheat or abandoned property).

"Transfer Notice" has the meaning set forth in Section 10.7(a).

"Treasury Regulations" mean the regulations promulgated by the Treasury Department with respect to the Code, as such regulations are amended from time to time, or corresponding provisions of future regulations.

"TTO" means CorEnergy Infrastructure Trust, Inc. (f/k/a Tortoise Capital Resources Corporation), a Maryland corporation.

"TTO Guaranty" means any guaranty provided by TTO for any indebtedness of the Partnership or obligations under the LGS Lease.

"Ultra Newco" means Ultra Wyoming LGS, LLC, a Wyoming limited liability company.

"Ultra Wyoming" means Ultra Wyoming, Inc., a Wyoming corporation.

"Unit" means a unit representing a General Partnership Interest or a Limited Partnership Interest.

"Valuation Date" has the meaning set forth in Section 11.4(d).

"Wholly Owned Affiliate" of any Person means:

(a) an Affiliate of such Person, one hundred percent (100%) of the voting stock or Beneficial Ownership of which is owned directly by such Person, or by any Person who, directly or indirectly, owns one hundred percent (100%) of the voting stock or Beneficial Ownership of such Person,

- (b) an Affiliate of such Person that, directly or indirectly, owns one hundred percent (100%) of the voting stock or Beneficial Ownership of such Person,
- and
- (c) any Wholly Owned Affiliate of any Affiliate described in clauses (a) or (b) of this definition.

1.2 Interpretation. Unless the context requires otherwise: (a) the gender (or lack of gender) of all words used in this Agreement includes the masculine and feminine; (b) references to Articles and Sections refer to Articles and Sections of this Agreement unless expressly provided otherwise; (c) references to Schedules and Exhibits, if any, refer to the Schedules and Exhibits attached to this Agreement, each of which is made a part hereof for all purposes; (d) references to laws refer to such laws as they may be amended from time to time, and references to particular provisions of a law include any corresponding provisions of any succeeding law; (e) references to money refer to legal currency of the United States of America; (f) the singular form of nouns, pronouns and verbs shall include the plural and vice versa; (g) "shall" and "will" have equal force and effect; (h) the words "include," "including," or "includes" shall be read to be followed by the words "without limitation" or words having similar import; and (i) the word "or" will have the inclusive meaning represented by the phrase "and/or".

ARTICLE II

FORMATION OF THE PARTNERSHIP

2.1 Formation: Governance by the Act. The Partnership was formed on the Formation Date as described in the Recitals. Except as otherwise provided in this Agreement to the contrary, the rights and obligations of the Partners and the administration and termination of the Partnership shall be governed by the Act.

2.2 <u>Name and Place of Business</u>. The business of the Partnership shall be conducted under the name of "Pinedale Corridor, LP" The principal place of business of the Partnership shall be 4200 W. 115th Street, Suite 210, Leawood, Kansas 66211 or such other place in the United States as the General Partner may from time to time designate by written notice to the Limited Partners. The registered office and the registered agent of the Partnership for service of process, as required by the Act, shall be as designated from time to time by the General Partner.

2.3 Filings and Registrations. The General Partner shall cause the due execution and filing, with the Secretary of State of the State of Delaware and such other proper authorities in each jurisdiction where the Partnership conducts its business, of all appropriate organizational and administrative documents of or with respect to the Partnership. The General Partner may take any action it deems necessary or appropriate for the Partnership to conduct or do business as a limited partnership in any state in which the conduct of business by the Partnership may require such action, provided that in any such event the Partnership shall at all times continue to be a limited partnership formed under and governed by the provisions of the Act.

2.4 <u>Term</u>. The term of the Partnership shall be perpetual, unless sooner terminated in accordance with the provisions of <u>Article XII</u> or as otherwise provided by law

2.5 Organizational Expenses and LGS Transaction Expenses. The Partnership shall pay, by direct payment or reimbursement to the relevant party upon the receipt of appropriate

supporting documentation, (a) all costs and expenses incurred by the Partners and their Affiliates in connection with organizing the Partnership, including the preparation of the Certificate and this Agreement, (b) all costs and expenses of the Partners and their Affiliates incurred in connection with negotiating, documenting, and implementing the LGS Transaction, (c) all costs and expenses incurred by TTO and its Affiliates in forming the General Partner, and (d) all costs and expenses incurred by Ross Avenue Investments and its Affiliates in connection with the acquisition by Ross Avenue Investments of its Partnership Interest and the participation by Ross Avenue Investments in the LGS Transaction as a Limited Partner of the Partnership (including the costs and expenses of due diligence and the review, analysis, negotiation, and drafting of all LGS Transaction Documents, irrespective of whether Ross Avenue Investments is a party thereto), including all such legal, consulting, and advisory fees and costs; provided, however, that (x) the costs and expenses payable to or on behalf of Ross Avenue Investments and its Affiliates pursuant to this Section 2.5 shall be subject to an aggregate maximum amount of \$200,000, and (y) the costs and expenses payable to or on behalf of TTO or the General Partner pursuant to this Section 2.5 shall not include any costs or expenses (including attorneys' fees, underwriting fees, or commissions) incurred in connection with raising public or private equity for TTO or the Partnership (including, for clarity, the equity investment of Ross Avenue Investments in the Partnership).

2.6 Partner Representations and Warranties. Each Partner hereby represents and warrants to the Partnership and to each other Partner that: (a) in the case of a Partner who is not a natural person, that the Partner is duly organized, validly existing, and in good standing under the law of its state of organization and that it has the requisite power and authority to execute this Agreement and to perform its obligations hereunder; (b) the Partner is acquiring its Partnership Interest for such Partner's own account as an investment and without an intent to resell or distribute such Partnership Interest; and (c) the Partner acknowledges that the Interests have not been registered under the Securities Act of 1933, as amended, or any state securities laws, and such Partner's Interest may not be resold or transferred by the Partner without appropriate registration or the availability of an exemption from such requirements.

2.7 <u>Special Purpose Bankruptcy Remote Entity Provisions</u>. For so long as the LGS Lease remains outstanding, reference is hereby made to Part B of Schedule 27.1 of the LGS Lease, and notwithstanding anything to the contrary herein, neither Partner shall cause the Partnership to fail to comply with those provisions in the LGS Lease relating to the status of the Partnership as a special purpose bankruptcy remote entity, and the General Partner shall comply with those provisions in the LGS Lease relating to its status as a special purpose bankruptcy remote entity.

2.8 LGS Lease and LGS Equity Investors' Agreement. The Partners acknowledge and agree that the LGS Lease, the LGS Equity Investors' Agreement and the LGS Financing Documents (including the TTO Guaranty) contain requirements and limitations applicable to the Partnership, the Partners and their Affiliates, including with respect to, among other things, confidentiality and the Transfer and ownership of, and Liens on, Partnership Interests. Each Partner agrees to comply with such Partner's obligations under the LGS Equity Investors' Agreement, and not to take any action, or fail to take any action, that would cause the Partnership to breach any of its obligations under the LGS Equity Investors' Agreement, or the LGS Financing Documents.

2.9 Indemnification. Each Partner agrees to indemnify the Partnership and the other Partners from and against any loss, expense, damage or injury suffered, sustained or incurred by the Partnership or the other Partners caused by (a) any condition, action or failure to act by or involving

the indemnifying Partner or any Affiliate of the indemnifying Partner that results in a breach by such Partner, any of its Affiliates or the Partnership of this Agreement, the LGS Equity Investors' Agreement, the LGS Financing Documents or the LGS Lease, or (b) in the case that the indemnifying partner is the General Partner or an Affiliate of the General Partner, (i) any CORR Default (as defined in the Credit Agreement) or (ii) any of the following: (A) the Capital Contribution by the General Partner of 1,050,420 common units of NGL Energy Partners, LP (the "*NGL Units*"), (B) the transfer of the NGL Units by the Partnership to Ultra Wyoming in partial satisfaction of the Purchase Price under the LGS Purchase Agreement, or (C) any breach by the Partnership of Section 3 of the First Amendment, dated December 12, 2012, to the LGS Purchase Agreement. Such indemnity shall be subject to an aggregate cap equal to the indemnifying Partner's aggregate Capital Contributions, and indemnifable losses, expenses, damages or injuries shall under no circumstances include consequential, special, exemplary, punitive, or noneconomic damages, including lost profits. All indemnity payments shall be made first from Available Cash otherwise distributable to the indemnifying Partner at the time, provided that the unavailability or insufficiency of Available Cash for such payments or back any indemnification payment promptly and in full.

ARTICLE III

PURPOSE

3.1 <u>Business Purpose</u>. The Partnership is organized solely for the purpose of acquiring, owning, holding, maintaining, financing, refinancing, mortgaging, encumbering, renovating, operating, leasing, managing and/or selling the Leased Property (such purposes, collectively the "*Business*").

3.2 <u>Powers</u>. Subject to any express limitations set forth in this Agreement, the Partnership shall have the power and authority to take and perform any and all acts and activities and to execute and deliver such documents, instruments and agreements as may be considered by the General Partner, in its sole discretion, to be necessary, appropriate, or desirable to accomplish or further the Business or to protect or benefit the Partnership.

ARTICLE IV

CAPITAL CONTRIBUTIONS; PARTNERSHIP INTERESTS

4.1 <u>Nominal Initial Capital Contributions</u>. On the Formation Date, the General Partner and the Initial Limited Partner made nominal initial Capital Contribution of \$1.00 and \$99.00, respectively. On the Effective Date, such nominal initial Capital Contributions will be returned to the General Partner and the Initial Limited Partner upon the initial capitalization of the Partnership pursuant to <u>Section 4.2</u>, and the Limited Partnership Interest of the Initial Limited Partner shall be canceled.

4.2 Initial Capitalization. On the Effective Date, the Partners will make the Capital Contributions set forth in Exhibit A, and thereafter (a) Ross Avenue Investments shall be admitted to the Partnership as a Limited Partner and (b) each Partner's Percentage Interest, and the number and class of Units held by each Partner, shall be as set forth in Exhibit A hereto.

4.3 Additional Capital Contributions. Except as provided in Sections 4.1 and 4.2, no Partner shall be obligated to make any Capital Contributions.

4.4 Capital Accounts. A separate Capital Account shall be maintained for each Partner. Each Partner's Capital Account shall be (a) increased by (i) the amount of money contributed by such Partner, (ii) the Fair Value of property contributed by such Partner (net of liabilities secured by such contributed property that the Partnership is considered to assume or take subject to Code Section 752), (iii) allocations to such Partner, pursuant to <u>Article VI</u>, of Partnership income and gain (or items thereof), and (iv) to the extent not already netted out under clause (b)(ii) below, the amount of any Partnership liabilities assumed by the Partner or which are secured by any property distributed to such Partner; and (b) decreased by (i) the amount of money distributed to such Partner, (ii) allocations to such Partner, (iii) allocations to such Partner, pursuant to <u>Article VI</u>, of Partnership liabilities secured by such distributed to such Partner is considered to assume or take subject to Code Section 752), (iii) allocations to such Partner, pursuant to <u>Article VI</u>, of Partnership loss and deduction (or items thereof), and (iv) to the extent not already netted out under clause (a)(ii) above, the amount of any liabilities of the Partner assumed by the Partnership or which are secured by any property contributed by such Partner to the Partnership.

In the event any interest in the Partnership is transferred in accordance with the terms of this Agreement, the transferee shall succeed to the Capital Account of the transferre to the extent it relates to the transferred interest.

In the event of (x) an additional Capital Contribution by an existing or an additional Partner of more than a de minimis amount or a distribution of property which results in a shift in Percentage Interests, (y) the distribution by the Partnership to a Partner of more than a de minimis amount of property (other than money) or a distribution of property in exchange for a partnership interest or (z) the liquidation of the Partnership within the meaning of Treasury Regulation § 1.704-1(b)(2)(ii)(g), the book basis of the Partnership Property shall be adjusted to Fair Value and the Capital Accounts of all the Partners shall be adjusted simultaneously to reflect the aggregate net adjustment to book basis as if the Partnership recognized gain and loss equal to the amount of such aggregate net adjustment; <u>provided</u>, <u>however</u>, that the adjustments resulting from clause (x) or (y) above shall be made only if and to the extent the General Partner determines that such adjustments are reasonably necessary or appropriate to reflect the relative economic interests of the Partners.

In the event that Partnership Property is subject to Code § 704(c) or is revalued on the books of the Partnership in accordance with the preceding paragraph pursuant to § 1.704-1(b)(2)(iv)(f) of the Treasury Regulations, the Partners' Capital Accounts shall be adjusted in accordance with § 1.704-1(b)(2)(iv)(g) of the Treasury Regulations for allocations to the Partners of depreciation, amortization, and gain or loss, as computed for book purposes (and not tax purposes) with respect to such Partnership Property.

4.5 No Right to Return of Capital Contributions; No Priority. Except as otherwise expressly provided in this Agreement, no Partner shall be entitled to (a) withdraw or demand the return of any part of its Capital Contribution, reduce such Partner's Capital Account, or receive any distributions from the Partnership, (b) demand or receive property other than cash in return for its Capital Contribution, or (c) receive or accrue any interest on any part of its Capital Contribution. Except as otherwise expressly provided in this Agreement, no Partner shall have any priority over any other Partner, as to the return of such Partner's Capital Contribution or Capital Account balance. An unrepaid Capital Contribution is not a liability of the Partnership or any Partner.

4.6 Authority to Issue New Units: Participation Rights. Subject to Section 7.3(a), the General Partner may from time to time cause the Partnership to issue to any Person (including, notwithstanding Section 7.3(f), to any Partner or any Affiliate of any Partner) additional Units with such terms and conditions as the General Partner may determine (collectively, "New Units"), and in connection therewith, the General Partner shall have the authority to amend this Agreement to reflect the issuance of the New Units; provided, however, that, for so long as the LGS Equity Investors' Agreement exists, it shall be a condition to any such issuance that the person to whom the New Units are issued shall be a party to the LGS Equity Investors' Agreement and that the provisions of Section 2.3 of the LGS Equity Investors' Agreement shall have been complied with. Furthermore, if the General Partner determines to issue New Units, then the Partnership shall offer to each Partner, in proportion to the relative number of Units held by each such Partner; the right to purchase such New Units on the same terms and conditions as the proposed issuance. Any New Units not then subscribed for by the Partners shall be reoffered to those Partners electing initially to purchase their full proportionate share, in proportion to the relative number of Units held by the Partners may be offered to Persons selected by the General Partner in its sole discretion. The General Partner shall determine the timing and such other procedures as may be neccessary and appropriate to enable the Partners to exercise their rights hereunder; provided, that in no event shall have the right to purchase pursuant to this Section 4.6 to an Affiliate of such Partner.

4.7 Partner Loans. Subject to Section 7.3, the General Partner may permit the Partnership to borrow funds from any Partner or any of its Affiliates, on such terms and conditions as the General Partner may determine (any such borrowing, a "Partner Loan"). However, if the General Partner determines to borrow funds pursuant to a Partner Loan, then the Partnership shall offer to each Partner, in proportion to the relative number of Units held by each such Partner, the right to participate in the Partner Loan on a *pari passu* basis. Any portion of the Partner Loan not initially subscribed for by the Partners pursuant to such participation right shall be reoffered to those Partners electing initially to participate to the full extent of their proportionate share, in proportion to the relative number of Units held by them or as they shall otherwise mutually agree, and such process shall be repeated until the entire proposed Partner Loan is subscribed for by the Partners. The General Partner shall determine the timing and such other procedures as may be necessary and appropriate to enable the Partners to exercise their rights hereunder; provided, that in no event shall the Partners be given less than thirty (30) days' and not more than ninety (90) days' prior written notice before being required to commit to participate in any Partner Loans shall not constitute a contribution to the capital of the Partnership or be credited to the Capital Account of the lending Partner(s) or entitle such lending Partner(s) to any increase in its share of Income or Loss which the Partnership or be credited to the Capital Account of the lending Partner(s) or entitle such lending Partner(s) to any increase in its share of Income or Loss which the Partnership or be credited to the Capital Account of the lending Partner(s) or entitle such lending Partner(s) to any increase in its share of Income or Loss which the Partnership may sustain.

4.8 <u>Conversion of General Partnership Interests</u>. At any time or from time to time, and without the consent of any Limited Partner, the General Partner as of the Effective Date (or any of its successors or permitted assigns) may cause the Company to convert Units representing General Partnership Interests held as of the Effective Date into the same number of Units representing Class A Limited Partnership Interests; provided, however, that, following any such conversion, the General

Partner (or such successor or permitted assign) continues to hold not less than the number of Units representing General Partnership Interests equal to one percent (1%) of all outstanding Units representing Partnership Interests. The Company will promptly notify the Limited Partners of any such conversion, and Exhibit A will be amended accordingly.

4.9 <u>Unit Certificates</u>. The Units shall be represented by certificates, in form and substance approved by the General Partner.

ARTICLE V

COMPENSATION OF GENERAL PARTNER

5.1 <u>Reimbursement of Out-of-Pocket Costs</u>. The General Partner shall be entitled to payment by or reimbursement from the Partnership of all reasonable outof-pocket third-party costs and expenses incurred by the General Partner on behalf of the Partnership or to carry out its responsibilities as the General Partner under this Agreement (e.g., third-party legal and accounting fees).

5.2 <u>Management Fee</u>. As consideration for the performance of its duties and responsibilities as the General Partner from and after the Effective Date, the Partnership will pay the General Partner an annual management fee of \$50,000 (the "*Management Fee*"). The Management Fee will be prorated for the year in which the Effective Date occurs (from the Effective Date through the end of such year) and for any other partial year of the Partnership following the year in which the Effective Date occurs, and will be payable in monthly advance installments on the Effective Date and thereafter on the first day of each month.

ARTICLE VI

ALLOCATIONS AND DISTRIBUTIONS

6.1 <u>Non-Liquidation Distributions</u>. The Partnership shall distribute Available Cash to the Partners from time to time as determined by the General Partner in its sole discretion. Any cash distributed by the Partnership to the Partners shall be distributed in accordance with the Partners' respective Percentage Interests as of the date of the distribution, except that the General Partner shall be entitled to a first priority distribution equal to the aggregate amount paid under a TTO Guaranty, to the extent such payment is not the result of a breach thereunder or hereunder by TTO or the General Partner.

6.2 <u>Liquidation Distributions</u>. Liquidation Proceeds shall be distributed in the following order of priority:

(a) To the payment of debts and liabilities of the Partnership (including the payment of the expenses of dissolution and liquidation and, to the extent permitted by applicable law, the repayment of Partner Loans);

(b) To the setting up of such reserves as the Person required or authorized by law to wind up the Partnership's affairs may reasonably deem necessary or appropriate for any disputed, contingent, or unforeseen liabilities or obligations of the Partnership, provided that any such reserves shall be paid over by such Person to an independent escrow agent, to be

held by such agent or its successor for such period as such Person shall deem advisable for the purpose of applying such reserves to the payment of such liabilities or obligations and, at the expiration of such period, the balance of such reserves, if any, shall be distributed as hereinafter provided; and

(c) The remainder to the Partners in accordance with and to the extent of their respective Capital Account balances after taking into account the allocation of all Income or Loss pursuant to this Agreement for the fiscal year(s) in which the Partnership is liquidated.

6.3 <u>Profits, Losses and Distributive Shares of Tax Items</u>. The Partnership's net income or net loss, as the case may be, for each fiscal year of the Partnership, as determined in accordance with such method of accounting adopted for the Partnership pursuant to <u>Article VIII</u>, shall be allocated to the Partners for both financial accounting and income tax purposes as set forth in this <u>Article VI</u>, except as otherwise provided for herein or unless all Partners agree otherwise.

6.4 <u>Allocation of Income, Loss and Credits</u> Income or Loss and Credits for each fiscal year shall be allocated among the Partners in accordance with their respective Percentage Interests. To the extent there is a change in the respective Percentage Interests of the Partners during the year, Income, Loss and Credits shall be allocated among the pre-adjustment and post-adjustment periods as provided in <u>Section 6.5(k)</u>.

6.5 <u>Special Rules Regarding Allocation of Tax Items</u>. Notwithstanding the foregoing provisions of this <u>Article VI</u>, the following special rules shall apply in allocating the net income or net loss of the Partnership:

(a) <u>Section 704(c) and Revaluation Allocations</u>. In accordance with Code Section 704(c) and the Treasury Regulations thereunder, income, gain, loss, and deduction with respect to any property contributed to the capital of the Partnership shall, solely for tax purposes, be allocated among the Partners so as to take account of any variation between the adjusted basis of such property to the Partnership for federal income tax purposes and its Fair Value at the time of contribution. In the event of a Revaluation, subsequent allocations of income, gain, loss and deduction with respect to such property shall take account of any variation between the book basis of such property to the Partnership and its Fair Value immediately after the adjustment in the same manner as under Code Section 704(c) and the Treasury Regulations thereunder. Any elections or other decisions relating to such allocation shall be made by the General Partner in a manner that reasonably reflects the purpose and intention of this Agreement. Allocations pursuant to this <u>Section 6.5(a)</u> are solely for income tax purposes and shall not affect, or in any way be taken into account in computing, for book purposes, any Partner's Capital Account or share of income or loss pursuant to any provision of this Agreement.

(b) <u>Minimum Gain Chargeback</u>. Notwithstanding any other provision of this <u>Article VI</u>, if there is a net decrease in Partnership Minimum Gain during a Partnership taxable year, each Partner shall be allocated items of income and gain for such year (and, if necessary, for subsequent years) in an amount equal to that Partner's share of the net decrease in Partnership Minimum Gain during such year (hereinafter referred to as the "Minimum Gain Chargeback Requirement"). A Partner's share of the net decrease in Partnership Minimum Gain is the amount of the total decrease multiplied by the Partner's percentage share of the Partnership Minimum Gain at the end of the immediately preceding taxable year.

A Partner is not subject to the Minimum Gain Chargeback Requirement to the extent: (1) the Partner's share of the net decrease in Partnership Minimum Gain is caused by a guarantee, refinancing or other change in the debt instrument causing it to become partially or wholly recourse debt or a Partner Nonrecourse Liability, and the Partner bears the economic risk of loss for the newly guaranteed, refinanced or otherwise changed liability; (2) the Partner contributes capital to the Partnership that is used to repay the Nonrecourse Liability and the Partner's share of the net decrease in Partnership Minimum Gain results from the repayment; or (3) the Minimum Gain Chargeback Requirement would cause a distortion and the Commissioner of the Internal Revenue Service waives such requirement.

A Partner's share of Partnership Minimum Gain shall be computed in accordance with Treasury Regulation § 1.704-2(g) and as of the end of any Partnership taxable year shall equal: (1) the sum of the nonrecourse deductions allocated to that Partner up to that time and the distributions made to that Partner up to that time of proceeds of a Nonrecourse Liability allocable to an increase of Partnership Minimum Gain, minus (2) the sum of that Partner's aggregate share of net decrease in Partnership Minimum Gain plus such Partner's aggregate share of decreases resulting from revaluations of Partnership Property subject to Partnership Nonrecourse Liabilities. In addition, a Partner's share of Partnership Minimum Gain shall be adjusted for the conversion of recourse and Partner Nonrecourse Liabilities into Nonrecourse Liabilities in accordance with Treasury Regulation § 1.704-2(g)(3). In computing the above, amounts allocated or distributed to the Partner's predecessor in interest shall be taken into account.

(c) <u>Partner Minimum Gain Chargeback</u>. Notwithstanding any other provision of this <u>Article VI</u>, if there is a net decrease in Partner Minimum Gain during a Partnership taxable year, any Partner with a share of that Partner Minimum Gain (determined under Treasury Regulation § 1.704-2(i)(5)) as of the beginning of the year shall be allocated items of income and gain for such year (and, if necessary, for subsequent years) equal to that Partner's share of the net decrease in Partner Minimum Gain. In accordance with Treasury Regulation § 1.704-2(i)(4), a Partner is not subject to the Partner Minimum Gain Chargeback requirement to the extent the net decrease in Partner Minimum Gain arises because the liability ceases to be Partner Nonrecourse Debt due to a conversion, refinancing or other change in the debt instrument that causes it to be partially or wholly a Nonrecourse Debt. The amount that would otherwise be subject to the Partner Minimum Gain Chargeback requirement is added to the Partner's share of Partnership Minimum Gain.

(d) <u>Qualified Income Offset</u> In the event any Partner unexpectedly receives an adjustment, allocation or distribution described in Treasury Regulation § 1.704-1(b)(2)(ii)(d)(4), (5) or (6), which causes or increases such Partner's Adjusted Capital Account Deficit, items of Partnership income and gain shall be specially allocated to such Partner in an amount and manner sufficient to eliminate such Adjusted Capital Account Deficit as quickly as possible, provided that an allocation under this <u>Section 6.5(d)</u> shall be made if and only to the extent such Partner would have an Adjusted Capital Account Deficit after all other allocations under this <u>Article VI</u> have been made.

(e) <u>Nonrecourse Deductions</u>. Nonrecourse Deductions for any fiscal year or other period shall be allocated to the Partners in proportion to their Percentage Interests.

(f) <u>Partner Nonrecourse Deductions</u>. Any Partner Nonrecourse Deduction shall be allocated to the Partner who bears the risk of loss with respect to the loan to which such Partner Nonrecourse Deductions are attributable in accordance with Treasury Regulation § 1.704-2(i).

(g) <u>Curative Allocations</u>. Any special allocations of items of income, gain, deduction or loss pursuant to <u>Section 6.5 (b), (c), (d), (e), (f) and (h)</u> shall be taken into account in computing subsequent allocations of income and gain pursuant to this <u>Article VI</u>, so that the net amount of any items so allocated and all other items allocated to each Partner pursuant to this <u>Article VI</u> shall, to the extent possible, be equal to the net amount that would have been allocated to each such Partner pursuant to the provisions of this <u>Article VI</u> if such adjustments, allocations or distributions had not occurred.

(h) <u>Loss Allocation Limitation</u>. Notwithstanding the other provisions of this <u>Article VI</u>, no Limited Partner shall be allocated Loss in any taxable year which would cause or increase an Adjusted Capital Account Deficit as of the end of such taxable year. All Loss in excess of the limitation in this <u>Section 6.5(h)</u> shall be allocated to the General Partner.

(i) <u>Share of Nonrecourse Liabilities</u>. Solely for purposes of determining a Partner's proportionate share of the "excess nonrecourse liabilities" of the Partnership within the meaning of Treasury Regulation § 1.752-3(a)(3), each Partner's interest in Partnership profits is equal to such Partner's respective Percentage Interest.

(j) <u>Compliance with Regulations</u>. The provisions of Section 4.4 and this Section 6.5 are intended to comply with Treasury Regulations § 1.704-1(b), 1.704-2 and 1.752-1 through 1.752-5, and shall be interpreted and applied in a manner consistent with such regulations. In the event the General Partner reasonably determines that it is prudent or advisable to modify the manner in which the Capital Accounts, or any increases or decreases thereto, are computed and the allocations pursuant to this <u>Article VI</u> in order to comply with such Treasury Regulations, the General Partner may cause such modification to be made without the consent of the Limited Partners, provided that such modification is not likely to have a material effect on the amounts distributable to any Partner upon the dissolution of the Partnership.

(k) <u>General Allocation Provisions</u>. Except as otherwise provided in this Agreement, all items that are components of net income or net loss shall be divided among the Partners in the same proportions as they share such net income or net loss, as the case may be, for the year. For purposes of determining the Income, Loss or any other items for any period, Income, Loss or any such other items shall be determined on a daily, monthly or other basis, as reasonably determined by the General Partner using any permissible method under Code Section 706 and the Treasury Regulations thereunder.

(1) <u>Negative Capital Make-up Requirement</u>. If on the liquidation and dissolution of the Partnership the General Partner has a negative Capital Account, after the allocation of all income, gain, loss and deductions pursuant to this <u>Article VI</u>, the General Partner shall contribute cash in an amount equal to such negative Capital Account.

6.6 <u>No Priority</u>. Except as may be otherwise expressly provided herein, no Partner shall have priority over any other Partner as to Partnership income, gain, loss, credits and deductions or distributions.

6.7 <u>Tax Withholding</u>. Notwithstanding any other provision of this Agreement, the General Partner is authorized to take any action that it determines to be necessary or appropriate to cause the Partnership to comply with any withholding requirements established under any federal, state or local tax law, including, without limitation, withholding on any distribution to any Partner. For all purposes of this <u>Article VI</u>, any amount withheld on any distribution and paid over to the appropriate governmental body shall be treated as if such amount had in fact been distributed to the Partner.

ARTICLE VII

MANAGEMENT AND CONTROL

7.1 <u>Management by the General Partner</u>. Subject to the provisions of this Agreement, the General Partner, acting through its Board of Directors, shall have the exclusive right to manage, control and conduct, and shall be solely responsible for the management, control and conduct of, the Partnership's business and affairs, and shall have all rights and powers generally conferred by law or necessary, advisable or incidental in connection therewith. The Limited Partners hereby consent to the exercise by the General Partner of the powers conferred on it by this Agreement. No Limited Partner (except one who may also be a General Partner, an officer, employee, agent, or contractor of the Partnership, or a stockholder, director, officer, employee, agent or contractor of the General Partner, and then only in such capacity) shall participate or take part in the management or control of the business and affairs of the Partnership or have any right or authority to act for or bind the Partnership.

7.2 Rights, Powers and Authority of General Partner. Except as otherwise provided herein, the General Partner shall have and possess all the rights, powers and authority of any general partner of a partnership without limited partners formed under the laws of the State of Delaware. In addition to any other rights, powers and authority it may possess under law or this Agreement, the General Partner shall have, subject to the provisions of Section 7.3 and any other limiting provision of this Agreement, all specific rights, powers and authority required for or appropriate to the management of the Partnership's business and affairs, which, by way of illustration but not by way of limitation, shall include the following rights, powers and authority to act for, in the name of, and on behalf of the Partnership to:

(a) perform or cause to be performed all of the Partnership's obligations under this Agreement, and to exercise and enforce all rights and remedies of the Partnership hereunder;

(b) perform or cause to be performed all of the Partnership's obligations under the LGS Transaction Documents, and to exercise and enforce all rights and remedies of the Partnership thereunder;

(c) purchase, acquire, own, hold, equip, operate, manage, maintain, develop, sell, convey, exchange, or otherwise dispose of or deal with the Business or the Partnership's

interest therein, the Partnership Properties, or any part thereof, upon such terms and conditions and for such consideration as it may determine;

(d) acquire and enter into any contract of insurance that the General Partner reasonably deems necessary or proper for the protection of the Partnership, for the conservation of the Partnership Properties or for any purpose beneficial to the Partnership;

(e) bring or defend, pay, collect, compromise, arbitrate, or otherwise adjust, claims or demands of or against the Partnership;

(f) make such elections under the tax laws of the United States, the States of Delaware and Wyoming, and other relevant jurisdictions as to the treatment of items of Partnership income, gain, loss, deduction, credit and other relevant matters, as it believes necessary or desirable and to file any and all necessary reports, tax returns and records with appropriate government and regulatory agencies;

(g) engage, retain or otherwise deal with any Persons as employees, agents, contractors, consultants, accountants, attorneys, or advisors of the Partnership, or in any other capacity as the General Partner deems necessary or desirable;

(h) designate and change from time to time the address of the registered office and the name of the registered agent for service of process on the Partnership in any state in which such designations are required;

(i) temporarily invest any funds of the Partnership in short-term, liquid investments where there is appropriate safety of principal;

(j) enter into any and all agreements that it may deem appropriate in carrying out the purposes of the Partnership and to perform or cause to be performed all of the Partnership's obligations under any such agreement, and to exercise any right of the Partnership under any such agreement; and

(k) take any and all other actions and execute, acknowledge, file, record, publish and deliver any and all instruments which it may deem necessary or appropriate to conduct the Business and exercise the powers of the Partnership.

7.3 Actions Requiring a Class B Limited Partner Majority. For so long as Ross Avenue Investments and its Affiliates own at least 100 Units representing Class B Limited Partnership Interests, the General Partner shall not take any of the following actions for itself or on behalf of the Partnership (as the case may be), and shall not otherwise permit the Partnership to take any of the following actions, without the prior written approval of a Class B Limited Partner Majority:

(a) issue (i) any Class B Limited Partnership Interests after the Effective Date or (ii) any Partnership Interests with rights, preferences or priorities senior to or otherwise more favorable than the rights and preferences of any then outstanding Limited Partnership Interests;

(b) grant the Partnership's consent to any of the following matters under the LGS Lease (or any other LGS Transaction Document): (i) any extension of the term of the LGS

Lease; (ii) any change to the methodology for determining any rent or purchase price for the Leased Property under the LGS Lease; (iii) the making of any material addition or improvement to the Leased Property; (iv) any proposed modification of terms or definitions incorporated by reference into this Agreement or the LGS Equity Investors' Agreement or that are otherwise applicable to a Partner hereunder or in another LGS Transaction Document; and (v) any other matter that could reasonably be expected to either reduce the present value of rent payments to be received by the Partnership under the LGS Lease or meaningfully increase the legal or financial risk borne by the Partnership under the LGS Lease;

(c) merge or consolidate with any other Person, acquire any properties or assets unrelated to or inconsistent with the conduct of the Business, or undertake any restructuring transaction that could reasonably be expected to adversely affect the Limited Partners;

(d) change or modify the purpose of the Partnership, as set forth in<u>Section 3.1</u>;

(e) incur, refinance, recapitalize, extend the term of, or otherwise amend the terms of any Debt of the Partnership if, as a result of such action, the Partnership would have aggregate Debt in an amount in excess of (i) while the LGS Lease (or any subsequent agreement under which the Partnership, as lessor, leases the Leased Property) is in effect, 4.0 times the annual rent payable for the current year under such lease, and (ii) if no such lease is in effect, 4.0 times the projected annual net revenue (i.e., gross revenue minus the Management Fee and all other expenses of the Partnership, including expenses of the General Partner reimbursed by the Partnership, all determined on a cash basis) of the Partnership for the current year, as reasonably determined by the General Partner;

(f) except for the issuance of New Units to any Partner or any Affiliate of any Partner in accordance with <u>Section 4.6</u>, enter into or amend any agreement with any Partner, any Affiliate of any Partner, or any director, manager, officer, or employee of any Partner or any Affiliate of any Partner;

(g) (i) make any loan to any Person other than a subsidiary Affiliate of the Partnership or (ii) allow the General Partner to incur debt that is not allowed under the LGS Lease;

(h) make (i) any material change to any tax or accounting policy of the Partnership or (ii) any change to any tax or accounting policy of the Partnership that would reasonably be expected to result in an adverse consequence to the holders of the Class B Limited Partnership Interests;

(i) redeem any Partnership Interest held by the General Partner, any of its Affiliates or any holder of a Limited Partnership Interest that is not a Class B Limited Partnership Interest;

(j) liquidate or dissolve, or take or acquiesce in the taking of any Bankruptcy Action;

(k) enter into any agreement or other binding commitment to take any of actions described in this Section 7.3; or

(1) take any action which by any other provision of this Agreement requires the prior written approval of a Class B Limited Partner Majority.

7.4 Advisory Board; Observer Rights.

(a) Advisory Board. The General Partner may, in its sole discretion, form an advisory board for the Partnership (the "Advisory Board"), which shall meet from time to time to discuss the operation of the Leased Property, the performance by the Partnership and Ultra Wyoming and Ultra Newco under the LGS Transaction Documents, and any other matters relating to the Business. If the General Partner forms the Advisory Board, then a Class B Limited Partner Majority, as well as any Limited Partner that, without regard to its holdings of Class B Limited Partnership Interests, holds a Percentage Interest of greater than fifteen percent (15%), shall each be entitled to appoint one member to the Advisory Board and to have additional observer rights at any meeting of the Advisory Board (subject to reasonable limitations established by the General Partner). Notwithstanding the foregoing, no member of the Advisory Board may be, or may be appointed by, a Disqualified Person. The Advisory Board will have no authority to take any action on behalf of, or to approve any action by, the Partnership. The Partnership will reimburse the reasonable out-of-pocket costs of Advisory Board members incurred in connection with attending any meeting of the Advisory Board.

(b) Observer Rights. A Class B Limited Partner Majority shall be entitled to appoint a representative (who shall be an employee of a Limited Partner) of the Limited Partners holding Class B Limited Partnership Interests (the "Observer") to (i) attend in a non-voting capacity any meeting of the board of directors of the General Partner, or of any committee of the board of directors of the General Partner, whether in person, telephonically, or by such other means whereby all attendees of the meeting can hear each other, (ii) receive notices of any such meeting, and any other information distributed in connection with any such meeting, concurrently with the delivery of such notices and other information to the directors or committee members, as the case may be, (iii) receive a information (and any supporting materials) in respect thereof) proposed to be taken in lieu of any such meeting, concurrently with the delivery of such consent action (and supporting materials) to the directors or committee members, as the case may be, and (iv) receive a copy of such consent action (and supporting materials) to the directors or committee members, as the case may be, and (iv) receive a copy of such consent action (and supporting materials) to the directors or committee members, as the case may be, and (iv) receive a copy of any signed unanimous consent action (and supporting materials) to the directors or committee members, as the case may be, and (v) receive a copy of any signed unanimous consent action taken in lieu of any such meeting, onclurently with the delivery of such consent. However, the General Partner shall have the right to withhold from the Observer any materials or other information, including portions of the minutes of meetings, and to exclude the Observer from any portion of any such meeting, could (x) jeopardize an attorney-client privilege, (y) result in a breach by the Partnership of its obligations under any agreement (including any LGS Transaction Document) or violate any applicable law, or (z) interfere

materials or other information relating to such transaction or matter and exclude the Observer from any portion of a meeting in which such transaction or matter is discussed. The Partnership will reimburse the Observer's reasonable out-of-pocket costs incurred in connection with attending any meeting of the board of directors of the General Partner, or any committee of the board of directors of the General Partner.

(c) <u>Directors and Officers Liability Insurance</u>. To the extent such coverage is available on commercially reasonable terms, the General Partner agrees to cause the Observer and the members of the Advisory Board to be covered by directors and officers liability insurance to the same extent that the directors of the General Partner are covered by such insurance, at the cost of the Partnership.

7.5 Other Businesses of Partners Any Partner and any of its Affiliates may engage independently or with others in other business ventures of every nature and description including, without limitation, any business competing with or supplemental to the Business, except that each Partner shall present to the Partnership any investment opportunity of such Partner or any of its Affiliates that relates to the Leased Property. No Partner shall be obligated to present any particular investment opportunity to the Partnership that is not related to the Leased Property even if such opportunity is of a character which, if presented to the Partnership, could be taken by the Partnership, and each Partner shall have the right to take for its own account or with others or to recommend to others any such particular investment opportunity. Subject to the provisions of Section 7.3, nothing in this Agreement shall be deemed to prohibit any Partner or any of its Affiliates from dealing with or otherwise engaging in business with Persons who have transacted business with, are transacting business with, or may transact business with, the Partnership. Neither the Partnership nor any Partner shall have any right by virtue of this Agreement or the partnership relationship created hereby in or to such other ventures or activities of another Partner, or to the income or proceeds derived therefrom, and the pursuit of such ventures, even if competitive with or supplemental to the Business, shall not be deemed wrongful or improper.

7.6 Exculpation and Indemnification.

(a) <u>Exculpation</u>. To the extent permitted by law, (i) the General Partner and its directors, officers, employees and agents (collectively, the "General Partner Parties") shall not be liable, responsible or accountable in damages or otherwise to the Partnership or any Limited Partner for any act, omission or error in judgment performed, omitted or made by any of the General Partner Parties in good faith and in a manner believed to be within the scope of authority granted to the General Partner Parties by this Agreement and in the best interests of the Partnership, provided that such act, omission or error in judgment does not constitute fraud, gross negligence, willful misconduct or breach of fiduciary duty and (ii) each Limited Partner and its directors, officers, employees and agents, as well as the Observer and all members of the Advisory Board appointed by a Class B Limited Partner Majority or any Limited Partner (collectively, the "Limited Partner Parties"), shall not be liable, responsible or accountable in damages or otherwise to the Partnership or the General Partner for any act, omission or error in judgment performed, omitted or made by any of the Limited Partner Parties in good faith and in a manner believed to be within the scope of authority granted to the Limited Partner Parties and in the best interests of the Partnership or the General Partner for any act, omission or error in judgment performed, omitted or made by any of the Limited Partner Parties in good faith and in a manner believed to be within the scope of authority granted to the Limited Partner Parties by this Agreement and in the best interests of the Partnership, provided that such act, omission or error in judgment does not constitute fraud, gross negligence, willful misconduct or breach of fiduciary duty.

(b) Indemnification. To the extent permitted by law, the Partnership shall indemnify and hold harmless the General Partner Parties and the Limited Partner Parties, in such capacities, from and against any loss, expense, damage or injury suffered, sustained or incurred by any of them arising out of any actual or threatened action, suit, proceeding or claim relating to their activities on behalf of the Partnership, or in furtherance of the interests of the Partnership, including, but not limited to, any judgment, award, settlement, reasonable attorney's fees and other costs or expenses incurred in connection with the defense or settlement of any actual or threatened action, suit, proceeding or claim; provided, however, that the acts, omissions or alleged acts or omissions upon which such actual or threatened action, suit, proceeding or claim is based were taken or omitted in good faith in a manner believed to be within the scope of the authorization granted by this Agreement and in the best interests of the Partnership and did not constitute fraud, gross negligence, willful misconduct or breach of fiduciary duty on the part of such indemnified party. The satisfaction of any foregoing indemnification obligation shall be from, and limited to, Partnership Property, and no Limited Partner shall have any personal liability with respect thereto.

(c) <u>Savings Clause</u>. If all or any portion of this <u>Section 7.6</u> is invalidated on any ground by any court of competent jurisdiction, then the Partnership will nevertheless indemnify and hold harmless the General Partner Parties and the Limited Partner Parties, in such capacities, from and against any loss, expense, damage or injury suffered, sustained or incurred by any of them arising out of any actual or threatened action, suit, proceeding or claim relating to their activities on behalf of the Partnership to the full extent permitted by any applicable portion of this <u>Section 7.6</u> that is not invalidated and to the fullest extent permitted by applicable law.

(d) <u>Scope Limitation</u>. The exculpation and indemnity provisions of this <u>Section 7.6</u> shall in no way limit, diminish, or reduce, and shall not apply to limit, diminish, or reduce, (i) the indemnity or reimbursement obligations of the Partners pursuant to <u>Section 2.8</u> or the LGS Equity Investors' Agreement or (ii) the liability of any Partner for a breach of this Agreement.

7.7 <u>Partnership Assets to be Used First</u>. All Partnership obligations and liabilities shall be satisfied first from Partnership Properties before the General Partner is responsible therefor, except as the General Partner may otherwise agree.

7.8 <u>No Preferential Transactions with Partners</u>. Except as and to the extent otherwise specifically provided or contemplated herein, the Business of the Partnership shall be operated in all respects as a business separate and distinct from the businesses of its Partners, and no Partner or its Affiliates shall have any right to preferential treatment from the Partnership over other Partners or third parties. The Partnership shall have no obligation to buy raw materials or other products or services from any Partner or such Partner's Affiliates or to sell any production output or other products or services to any Partner or such Partner's Affiliates.

ARTICLE VIII

ACCOUNTING AND BANK ACCOUNTS

8.1 <u>Books and Records</u>. At all times during the existence of the Partnership, the General Partner shall maintain, or cause to be maintained, full and true books of account reflecting

all Partnership transactions, and which shall be appropriate and adequate for the Business and in compliance with the requirements of the LGS Lease and the financing documents to which the Partnership is a party. Such books of account, together with a copy of this Agreement and a copy of the Certificate, a current list of the names and last known mailing address of each Partner and any other records and reports required to be maintained under the Act, shall at all times be maintained at the principal place of business of the Partnership. The Partnership shall not be required to deliver to any Limited Partner a copy of the Certificate or any amendment thereto after the same has been duly filed, but the Partnership shall deliver such a copy to any Limited Partner who requests a copy in writing. Each Limited Partner and its designated representatives (including, without limitation, attorneys and accountants) shall, subject to limitations on access for Disqualified Persons and confidentiality obligations set forth in the LGS Equity Investors' Agreement, have the right at any time and at such Limited Partner's expense (unless the Partnership), to inspect such books of account and other records of the Partnership in accordance with reasonable limitations and procedures established by the General Partner.

8.2 <u>Taxable and Fiscal Year</u>. The Partnership will have an initial tax year ending December 31, 2012, and will change to any different tax year required pursuant to Code Section 706. The Partnership's fiscal year shall end on December 31 of each year.

8.3 <u>Reporting</u>.

(a) <u>Financial Statements</u>. Within one hundred (100) days after the end of each fiscal year of the Partnership, the General Partner shall deliver to each Limited Partner (i) an audited balance sheet as of the end of such year and audited statements of income and cash flows for such year and (ii) all information with respect to the Partnership necessary for the Limited Partner's Federal and state income tax returns. Within sixty (60) days after the end of the first, second, and third quarters of each fiscal year of the Partnership, the General Partner shall deliver to each Limited Partner a balance sheet as of the end of such quarter and an income statement and statement of cash flows for such quarter.

(b) <u>Partnership Notices</u>. Subject to any limitations on the disclosure of information to Disqualified Persons and confidentiality obligations set forth in the LGS Equity Investors' Agreement, the General Partner shall promptly deliver to each Limited Partner (i) all notices received by the Partnership under the LGS Transaction Documents or any financing document to which the Partnership is a party, (ii) all material notices received by the Partnership under any other agreement to which the Partnership is a party, (iii) notice of any material event affecting the Partnership or the Leased Property, and (iv) copies of operating reports with respect to the Leased Property.

8.4 <u>Tax Returns and Elections; Tax Matters Partner</u>. The General Partner shall cause to be prepared and timely filed all Federal, state and local income tax returns and other returns or statements required of the Partnership by applicable law. The Partnership shall claim all deductions and make such elections for Federal or state income tax purposes which the General Partner reasonably believes will produce the most favorable tax results for the Partners. The General Partner is hereby designated as, and the General Partner hereby accepts is designation as, the Partnership's "Tax Matters Partner," as defined in the Code, and in such capacity the General Partner is hereby authorized and empowered to act for and represent the Partnership and each of the Limited Partners before the Internal Revenue Service in any audit or examination of any Partnership tax return and

before any court selected by the General Partner for judicial review of any adjustment assessed by the Service. Each of the Limited Partners does by execution of this Agreement consent to and agree to become bound by all actions of the General Partner as "Tax Matters Partner," including any contest, settlement or other action or position which the General Partner may deem proper under the circumstances. The General Partner and the other Partners specifically acknowledge, without limiting the general applicability of this <u>Section 8.4</u>, that the General Partner shall not be liable, responsible or accountable in damages or otherwise to the Partnership or any Partnerwith respect to any action taken by it in its capacity as the Tax Matters Partner except for gross negligence or willful misconduct. All out-of-pocket expenses incurred by the General Partner in its capacity as the Tax Matters Partner shall be considered expenses of the Partnership for which the General Partner shall be entitled to full reimbursement in the absence of gross negligence or willful misconduct.

8.5 Section 754 Election. In the event a distribution of Partnership Properties occurs which satisfies the provisions of Section 734 of the Code or in the event a transfer of a Partnership Interest occurs which satisfies the provisions of Section 743 of the Code, the Partnership shall, if requested to do so by the distributee or transferee, elect, pursuant to Section 754 of the Code, to adjust the basis of the Partnership Properties to the extent allowed by such Section 734 or 743 and shall cause such adjustments to be made and maintained. Any additional accounting expenses incurred by the Partnership in connection with making or maintaining such basis adjustment shall be reimbursed to the Partnership by the distributee of such assets or the transferee of such Partnership Interest who benefits from the making and maintenance of such basis adjustment.

8.6 <u>Bank Accounts</u>. All funds of the Partnership shall be deposited in the Partnership's name in such bank, money market or similar account(s) as may be designated by the General Partner and where there is appropriate safety of principal. Withdrawals therefrom shall be made only by persons authorized to do so by the General Partner.

ARTICLE IX

LIMITED PARTNERS

9.1 <u>Rights of Limited Partners</u>. The Limited Partners shall have the powers and shall be entitled to exercise the rights afforded to them by the terms of this Agreement and the Act, and the exercise of such rights shall not be deemed to constitute participation or taking part in the management or control of the Partnership or the Business.

9.2 <u>Limitation on Limited Partners' Liability</u>. No Limited Partner shall be personally liable for the expenses, liabilities or obligations of the Partnership or the General Partner. The liability of each Limited Partner shall be limited solely to the amount of such Limited Partner's Capital Contributions. No provision of this Agreement shall be construed to create an obligation of a Limited Partner to contribute additional capital to the Partnership for the benefit of any third party.

9.3 <u>Power of Attorney</u>. Subject to <u>Section 7.3</u>, each Limited Partner hereby irrevocably constitutes and appoints the General Partner as such Limited Partner's true and lawful attorney-in-fact with full power and authority to act in such Limited Partner's name, place and stead to make, execute, acknowledge, deliver, swear to, file, record and publish (a) any amendments to or cancellations of the Certificate that may require the signature of such Limited Partner, (b) all fictitious name registration forms, (c) any amendments to this Agreement adopted in accordance with this Agreement, (d) all certificates, forms and other instruments necessary to qualify, continue or terminate the qualification

of the Partnership in any state where it may be doing business, and (e) any other instrument which the General Partner deems to be in the best interests of the Partnership to file and which is not inconsistent with this Agreement. The Power of Attorney hereby granted by each Limited Partner is coupled with an interest, is irrevocable, and shall survive the death, incapacity, dissolution, termination of or Bankruptcy Action with respect to the Limited Partner or the transfer by the Limited Partner of all or part of such Limited Partner's Partnership Interest. The Power of Attorney hereby granted is a durable power of attorney. Each Limited Partner shall deliver to the General Partner, within five days of receipt of a request therefor from the General Partner, such other and further powers of attorney or instruments as the General Partner deems necessary or appropriate to carry out the intent and purposes of this Section 9.3 and the remainder of this Agreement.

9.4 <u>Changes in Addresses</u>. Each Limited Partner shall promptly report to the Partnership any changes in the address of the Limited Partner as previously reported to the Partnership. The General Partner shall cause this Agreement (specifically, <u>Exhibit A</u>) to be duly amended to reflect the address changes of the Limited Partners, if any.

9.5 <u>General Partner as Limited Partner</u>. If the General Partner acquires or becomes a transferee of all or a part of the Partnership Interest of a Limited Partner, the General Partner, subject to the provisions of <u>Article X</u>, shall be treated also as a Limited Partner to the extent of such Partnership Interest. If any General Partnership Interests are converted to Limited Partnership Interests pursuant to <u>Section 4.8</u>, the General Partner shall be treated also as a Limited Partner to the extent of such converted Partnership Interests.

ARTICLE X

TRANSFERS OF PARTNERSHIP INTERESTS BY LIMITED PARTNERS

10.1 General Restrictions. No Limited Partner shall Transfer all or any part of its Limited Partnership Interest except as provided in this Agreement. Any purported Transfer of a Limited Partnership Interest by a Limited Partner in violation of the terms of this Agreement shall be void *ab initio* to the extent permissible under applicable law. To the extent that it is not permissible under applicable law for a Transfer by a Limited Partner in violation of the terms of the terms of this Agreement to be void *ab initio*, the transferee shall under no circumstances be admitted as a Substitute Limited Partner or be entitled to exercise any voting or approval rights under <u>Section 7.3(b)</u> or otherwise, and instead shall only be entitled to receive, to the extent of the Partnership Interest transferred, the Distributions to which the transferor would be entitled (and provided for clarity that a transferee by means of an unenforced charge, pledge, encumbrance, mortgage or other like Transfer shall not be entitled to any such Distributions). Furthermore, any purported Transfer of a Limited Partnership Interest hat contravenes Section 2.3 of the LGS Equity Investors' Agreement (so long as such agreement remains in effect) shall be, and shall be treated by the Partnership and the other Partners as, void *ab initio*, and the purported transfere shall (a) under no circumstances be admitted as a Substitute Limited Partner, (b) not be entitled to any rights hereunder whatsoever, including any voting, veto, or information rights, and (c) not be entitled to any Distributions whatsoever under any circumstances. A permitted Transfer shall be effective as of the date specified in the instruments relating thereto that are delivered to the General Partner. No Limited Partner shall request the General Partner or the Partnership to grant all or any portion of any rights in any Limited Partnership Interest to any Transfer that is not permitted pursuant to this Agreement or under the LGS Equity Investors' Agreement (so long as such ag

10.2 Permitted Transfers. A Limited Partner shall have the right to Transfer (but not to substitute the transferee as a Substitute Limited Partner in such Limited Partner's place except in accordance with <u>Section 10.3</u>) all or any part of such Limited Partner's Limited Partnership Interest provided that (a) the Transfer would not result in the "termination" of the Partnership pursuant to Section 708 of the Code, (b) except in the case of a Transfer to an Affiliate of the transferor, the Transfer occurs no earlier than six (6) months after the Effective Date, (c) the transferor has complied with the provisions of <u>Section 10.7</u> and <u>Section 10.8</u>, if applicable, and the Transfer otherwise complies with the requirements of this Agreement, (d) the transferor has delivered to the General Partner and the other Limited Partners that neither the Transfer nor any offering in connection therewith is required to be registered under either the Securities Act of 1933, as amended, or any applicable state securities laws, and (e) to the extent applicable, the Transfer complies with the requirements of the LGS Equity Investors' Agreement. Any transfere desiring to make a further Transfer of all or any part of a Limited Partners shall be subject to all of the provisions of this <u>Article X</u> to the same extent and in the same manner as any Limited Partner desiring to make any such Transfer.

10.3 Substitute Limited Partners. No transferee of all or any part of any Limited Partnership Interest shall become a Substitute Limited Partner unless and until:

- (a) the Transfer is permitted under <u>Section 10.2</u> and is not prohibited under <u>Section 10.1</u>;
- (b) the transferor has stated such intention in the instrument of Transfer;

(c) the transferee has executed an instrument accepting and adopting the terms and provisions of this Agreement and has provided such other instruments and such assurances as to the enforceability of all such instruments as the General Partner may reasonably require;

(d) the transfereor or transferee has paid all reasonable expenses of the Partnership in connection with the admission of the transferee as a Substitute Limited Partner; and

(e) to the extent applicable, the transferee satisfies the requirements of the LGS Equity Investors' Agreement, including (if applicable) by agreeing in writing to be bound by the LGS Equity Investors' Agreement.

The Partners hereby consent to any substitution made in accordance with the provisions of this<u>Section 10.3</u>. Upon satisfaction of all of the foregoing conditions with respect to a particular transferee, the General Partner shall cause this Agreement (specifically, <u>Exhibit A</u>) to be duly amended to reflect the admission of the transferee as a Substitute Limited Partner.

10.4 Effect of Bankruptcy, Death, Incompetency or Termination of a Limited Partner. The death, adjudication of incompetency, dissolution, or termination of the existence of, or the occurrence of a Bankruptcy Action with respect to, a Limited Partner shall not cause the termination or dissolution of the Partnership. Upon any such occurrence, the trustee, executor, administrator, guardian, conservator or similar legal representative of the Limited Partner, in such capacity, shall be liable for all of the obligations of the Limited Partner and shall have all of the

rights of the Limited Partner for the purpose of settling or managing the Limited Partner's estate or property.

10.5 Effect of Admission as a Substitute Limited Partner. Unless and until admitted as a Substitute Limited Partner pursuant to Section 10.3, a transferee of a Limited Partnership Interest shall not be entitled to exercise any rights of a limited partner in the Partnership, including the right to vote, grant approvals, or give consents with respect to such Partnership Interest, the right to require any information or accounting of the Partnership's business and affairs, or the right to inspect the Partnership's books and records. Rather, such transferee shall only be entitled to receive, to the extent of the Partnership Interest transferred, the Distributions to which the transferor would be entitled (and provided for clarity that a transferee by means of an unenforced charge, pledge, encumbrance, mortgage or other like Transfer shall not be entitled to any such Distributions). A transferee who has become a Substitute Limited Partner has, to the extent of the Partnership Interest transferred to it, all the rights and powers of the Limited Partner for whom the transferee is substituted and is subject to the restrictions and liabilities of a Limited Partner under this Agreement and the Act. Upon admission of a transferee as a Substitute Limited Partnership Interest to acquired by the Substitute Limited Partner shall cease to be a limited partner of the Partnership to the extent of such Partnership Interest. A Person shall not cease to be a Limited Partner upon assignment of all of such Limited Partner's Limited Partnership Interest unless and until the transferee(s) become(s) a Substitute Limited Partner.

10.6 Withdrawal of a Limited Partner. No Limited Partner shall have the right or power to withdraw, resign or retire from the Partnership.

10.7 <u>Right of First Bid</u>.

(a) Any Limited Partner proposing to Transfer (other than by charge, pledge, encumbrance, mortgage or other like Transfer) a Limited Partnership Interest to any Person other than a Wholly Owned Affiliate of such Limited Partner shall provide written notice of its intention to make such Transfer (a "*Transfer Notice*") to the other Partners.

(b) Upon receipt of the Transfer Notice, each Partner shall have the right (individually or collectively), for a period of thirty (30) days following the date of the Transfer Notice, to submit to the transferring Limited Partner (either alone or in combination with one or more other Partners) an irrevocable and unconditional offer to purchase, at the per Unit price (the "*Bid Price*") and on the terms (the "*Bid Terms*") set forth in such offer (a "*Bid*"), all, but not less than all, of the Units subject to the Transfer Notice.

(c) Upon receipt of a proper Bid, the transferring Limited Partner may, in its sole discretion, accept a Bid by delivering written notice (an "*Acceptance*") to the Partner or Partners submitting such Bid within thirty (30) days of receipt of such Bid. If separate and differing Bids are submitted by Partners entitled to receive the Transfer Notice, the transferring Limited Partner may, in its sole discretion, determine which Bid is superior (based on Bid Price and Bid Terms). The transferring Limited Partner must inform all of the offering Partners of the Bid it determines is superior by delivering an Acceptance Notice to each offering Partner, which specifies the Bid Price and the Bid Terms of the superior Bid.

(d) The closing of the Transfer of the Units covered by any Bid shall occur within sixty (60) days following the date of the Acceptance Notice (or, if later, within five (5)

business days after the receipt of all necessary approvals from applicable Governmental Authorities). To consummate such purchase, each Partner who submitted the Bid accepted by the transferring Limited Partner (each a "*Purchasing Partner*" and collectively, the "*Purchasing Partners*") shall (i) pay (by wire transfer of immediately available funds (in U.S. dollars) to such U.S. bank accounts as the transferring Limited Partner may designate in writing to each Purchasing Partner no later than three (3) business days prior to the closing date for the Transfer) an amount equal to the product of (A) the Bid Price, multiplied by (B) the number of Units being purchased by each Purchasing Partner pursuant to such Bid, and (ii) take all such further actions and execute, acknowledge and deliver all such further documents that are necessary to effectuate the Transfer of the Units contemplated by this <u>Section 10.7</u>. Notwithstanding the preceding sentence, the Partners acknowledge and agree that the Bid Price may be paid in whole or in part with equity of any Purchasing Partner or any of its Affiliates, as the transacting Partners may agree.

(e) Upon the closing of such Transfer, (i) all of such transferring Limited Partner's obligations and liabilities associated with the Units which are the subject of such Transfer will terminate except those obligations and liabilities accrued through the date of such closing, (ii) such transferring Limited Partner shall have no further rights as a Limited Partner in respect of the Units which are the subject of such Transfer, and (iii) all the rights, obligations and liabilities associated with the Units which are the subject of such Transfer (other than the obligations and liabilities of the transferring Limited Partner accrued through the date of such closing, unless otherwise agreed in writing) shall become the rights, obligations and liabilities of the Purchasing Partner acquiring such Units.

(f) If no Bid is submitted within the 30-day period following the date of the Transfer Notice, or if Bids are submitted but (i) no Bid is accepted by the transferring Limited Partner, or (ii) if a Bid is accepted by the transferring Limited Partner but the closing of the Transfer has not occurred by the later of (A) the date sixty (60) days following the date of the Acceptance Notice and (B) the date five (5) days after the receipt of all necessary approvals from applicable Governmental Authorities, then the transferring Limited Partner shall have the right for a period of one hundred eighty (180) days (two hundred forty (240) days in the circumstance where the closing has not occurred as a result of a failure of the Purchasing Partner to abide by its obligations in connection with the closing of the Transfer) thereafter to Transfer such Units to any Person at any price and on any terms so long as such price is no less favorable to such transferring Limited Partner, than the price and terms contained in the best of any Bids (based on Bid Price and Bid Terms) received hereunder within such 30-day period following the date of the Transfer Notice or selected as superior pursuant to <u>Section 10.7(c)</u>. If such Transfer is not consummated within such 180-day or 240-day period, as the case may be (provided that, if a binding purchase agreement has been entered into within such period but the consummation of such Transfer has not occurred by the end of such period, then such Transfer may be consummated within 90 days following the end of such period, then such Transfer may be consummated within 90 days following the end of such period, then such Transfer to first bid as set forth in this <u>Section 10.7</u>.

10.8 Tag-Along Right in Favor of Holders of Class B Limited Partnership Interests Any Partner (including the holder of a Class B Limited Partnership Interest) proposing to Transfer (other than by charge, pledge, encumbrance, mortgage or other like Transfer) a Limited Partnership

Interest or a General Partnership Interest that, pursuant to Section 4.8 may be converted into a Limited Partnership Interest, to a non-Affiliate shall provide to each holder of a Class B Limited Partnership Interest that is not an Affiliate of the General Partner (a "Tag-Along Right Holder"), written notice of the transferring Partner's intention to make such Transfer (a "Tag-Along Transfer Notice"). The Tag-Along Transfer Notice shall set forth the price and terms of the proposed Transfer (the 'Transfer Terms'), as well as the portion (by percentage) of the transferring Partner's Partnership Interest (including, where applicable, a calculation of the Limited Partnership Units into which any General Partner Units would convert into under Section 4.8) proposed to be Transferred to the proposed transferee (the "Subject Interest"), provided that the representations and warranties of the Tag-Along Right Holder shall be limited to representations and warranties as to its ownership of the Tag-Along Interest (as hereinafter defined) and the absence of adverse claims with respect to the Tag-Along Interest, other than the security interest in favor of the lender or lenders to the Partnership and the restrictions on transfer contained in the LGS Equity Investors' Agreement. Upon receipt of the Tag-Along Transfer Notice, each Tag-Along Right Holder shall have the right, for a period of thirty (30) days following the date of the Tag-Along Transfer Notice, to submit to the transferring Partner an irrevocable election to participate in the proposed Transfer by the transferring Partner on the Transfer Terms (the "Tag-Along Election Notice"). Such election shall apply, as the Tag-Along Election Notice provides, to all or any part of the same portion (by percentage) of the Tag-Along Right Holder's Limited Partnership Interest as the portion (by percentage) of the transferring Partner's Partnership Interest represented by the Subject Interest (the "Tag-Along Interest") (i.e., if the Subject Interest represents thirty percent (30%) of the transferring Partner's Partnership Interest, the Tag-Along Interest may represent up to thirty percent (30%) of the Partnership Interest of the Tag-Along Right Holder, as the Tag-Along Right Holder so elects). In the event any Tag-Along Right Holder exercises its tag-along right under this Section 10.8, the transferring Partner will, as a condition to its right to Transfer the Subject Interest to the proposed transferee, cause the proposed transfere to purchase all Tag-Along Interests as well as the Subject Interest on the Transfer Terms and at the same time. Such purchase shall occur no later than the later of (a) the date fifteen (15) business days following the expiration of the 30-day period for delivering Tag-Along Election Notices and (b) the date five (5) business days after the receipt of all necessary approvals from applicable Governmental Authorities for the purchase of the Subject Interest and all Tag-Along Interests. In the event that no Tag-Along Right Holder timely delivers a Tag-Along Election Notice, then the transferring Partner shall be permitted to Transfer the Subject Interest to the proposed transfere on the Transfer Terms within sixty (60) days or, if later, the date five (5) business days after the receipt of all necessary approvals from applicable Governmental Authorities.

ARTICLE XI

CHANGES IN THE GENERAL PARTNER AND TRANSFERS OF GENERAL PARTNERSHIP INTERESTS

11.1 <u>General Restrictions</u>. No General Partner shall Transfer all or any part of its General Partnership Interest except as provided in this Agreement. Any purported Transfer of a General Partnership Interest by a General Partner in violation of the terms of this Agreement shall void *ab initio* to the extent permissible under applicable law. To the extent that it is not permissible under applicable law for a Transfer by a General Partner in violation of the terms of this Agreement to be void *ab initio*, the transferee shall under no circumstances be admitted as an additional or successor General Partner, and instead shall only be entitled to receive, to the extent of the Partnership Interest transferred, the Distributions to which the transferor would be entitled (and provided for clarity that a transferee by means of an unenforced charge, pledge, encumbrance,

mortgage or other like Transfer shall not be entitled to any such Distributions). Furthermore, any purported Transfer of a General Partnership Interest that violates the LGS Equity Investors' Agreement (so long as such agreement remains in effect) shall be, and shall be treated by the Partnership and the other Partners as, void *ab initio*, and the purported transfere shall (a) under no circumstances be admitted as a substitute or replacement General Partner (or as a Limited Partner or Substitute Limited Partner), (b) not be entitled to any rights hereunder whatsoever, including any management, voting, veto, or information rights, and (c) not be entitled to any Distributions whatsoever under any circumstances. No General Partner shall request the Partnership to grant all or any portion of any rights in any General Partnership Interest to any Person as a result of any Transfer that is not permitted pursuant to this Agreement or under the LGS Equity Investors' Agreement (so long as such agreement remains in effect).

Permitted Transfers. A General Partner shall have the right to Transfer (but not to substitute the transferee as a General Partner in such General Partner's stead) all or any part of such General Partnership Interest provided that (a) the Transfer would not result in the "termination" of the Partnership pursuant to Section 708 of the Code, (b) except in the case of a Transfer to an Affiliate of the transferor, the Transfer occurs no earlier than six (6) months after the Effective Date, (c) except in the case of a Transfer to an Affiliate of such General Partner, the remaining General Partners (if any) and a Class B Limited Partner Majority have consented in writing to such Transfer and the proposed transferee, (d) the Transfer complies with the requirements of this Agreement, (e) the transferor has delivered to the remaining General Partners (if any) and the Limited Partners an opinion of counsel reasonably satisfactory to the remaining General Partners and the Limited Partners that neither the Transfer nor any offering in connection therewith is required to be registered under either the Securities Act of 1933, as amended, or any applicable state securities laws and that such Transfer does not adversely affect any exemption from registration that was available to the Partnership, (f) to the extent applicable, the Transfer (and the organizational documents of the Transferee) complies with the requirements of the LGS Lease and the LGS Equity Investors' Agreement, including applicable requirements to maintain the Partnership's status as a Special Purpose Bankruptcy Remote Entity, and the Transferee has agreed to be Party to the LGS Equity Investors' Agreement. Any transferee desiring to make a further Transfer of all or any part of a General Partnership Interest shall be subject to all of the provisions of this Article XI to the same extent and in the same manner as any General Partner desiring to make any such Transfer. The Limited Partners hereby agree to not unreasonably withhold consent to any Transfer made in accordance with the provisions of this Section 11.2 and not prohibited by Section 11.1. A Person shall not cease to be a General Partner upon the Transfer of all of such General Partner's General Partnership Interest unless and until the transferee(s) thereof has (have) been admitted to the Partnership as a successor general partner pursuant to Section 11.5(a). Upon such admission, the successor general partner automatically shall be deemed to have exercised a right to carry on the Partnership, and the Partnership shall not be deemed to have dissolved and be required to wind up its affairs.

11.3 <u>Withdrawal or Removal of a General Partner</u>. Except with the prior written approval of the other General Partners (if any) and a Class B Limited Partner Majority, no General Partner shall have the right to withdraw, resign or retire from the Partnership. A General Partner may not be removed by the other Partners.

11.4 Effect of Withdrawal or Removal of a General Partner. In the event that a General Partner withdraws, resigns, or retires with the prior written approval of the other General Partners (if any) and a Class B Limited Partner Majority in accordance with Section 11.3, or in the

event that a General Partner is the subject of a Bankruptcy Action or ceases to be a general partner of the Partnership pursuant to the Act, the following shall apply:

(a) Such General Partner shall immediately cease to be a general partner of the Partnership (thereby terminating all management powers, duties and responsibilities of such General Partner) and such General Partner's Partnership Interest shall be deemed to have been surrendered (except to the extent it is to be purchased pursuant to subsection (d) below).

(b) The remaining General Partners, if any, shall have the right and power to continue the business of the Partnership rather than allow the Partnership to be dissolved and its affairs wound up. Any exercise of such right by the remaining General Partners shall be made in writing, within sixty (60) days following such event, and a copy thereof shall be given to each Limited Partner. The remaining General Partners shall cause this Agreement and the Certificate to be duly amended and filed to reflect any such event or continuation as and when required by the Act.

(c) If the Partnership is dissolved as a result of such event and the Partnership is wound up and terminated pursuant to the provisions of <u>Article XII</u>, then such General Partner (the "*Former General Partner*") shall remain entitled to the Distributions and allocations that would otherwise be made to it under the provisions of <u>Articles VI and XII</u>. If the business of the Partnership is continued by the remaining General Partners, if any, the Former General Partner shall be entitled to receive from such remaining General Partners (and not from the Partnership) the amount, in the manner and on the terms and conditions set forth in subsection (d) below. If the business of the Partnership is continued by the Limited Partners pursuant to the provisions of <u>Section 11.5(b)</u>, the Former General Partner shall be entitled to receive from the Partnership or the successor General Partner(s) (and not from the Limited Partners) the amount, in the manner and on the terms and conditions set forth in subsection (d) below. If the business of the Partnership or the successor General Partner(s) (and not from the Limited Partners) the amount, in the manner and on the terms and conditions set forth in subsection (d) below.

(d) Unless otherwise agreed upon by the Former General Partner and the Person(s) specified in the last two sentences subsection (c) above (the "*Purchaser*"), the amount payable to the Former General Partner by the Purchaser shall be determined by the Accountant and shall be equal to the amount, if any, that the Former General Partner would receive if the non-liquid assets of the Partnership were sold for their Fair Value as of the Valuation Date (as defined below), the Partnership was dissolved and the proceeds from the deemed sale and all liquid assets of the Partnership as of the Valuation Date were distributed to the Partners in accordance with the provisions of <u>Section 6.2(c)</u> after compliance with <u>Sections 6.2(a) and (b)</u> as of such date. The "*Valuation Date*" shall mean the date of the withdrawal, resignation, or retirement of the Former General Partner, or the date that the Former General Partner ceased to be a general partner of the Partnership pursuant to the Act. The Fair Value of any non-liquid assets shall be determined by a qualified appraiser(s) selected by the remaining General Partner, if any, or if none, by a Limited Partner Majority. The appraisal shall be completed and delivered to the Accountant as soon as practicable following the Valuation Date.

Within fifteen (15) days after the Fair Value of the non-liquid assets is determined, the Accountant shall determine the amount payable to the Former General Partner and shall give written notice thereof to the Former General Partner, the remaining General Partners, if any, the successor General Partner, if any, and the Limited Partners. Payment shall occur on

such date (the "*Purchase Date*") as may be selected by the Purchaser, which date shall be not more than thirty (30) days following the date the notice is given by the Accountant. At the closing, the Former General Partner shall execute and deliver to the Purchaser such deeds, bills of sale, assignments and other instruments as shall reasonably be requested by the Purchaser to effect the transfer to the Purchaser, as of the Valuation Date, of all of the Former General Partner's right, title and interest in its General Partnership Interest.

The Partnership shall pay the fees and expenses of the Accountant and appraiser in connection with the operation of the provisions of this Section but the cost thereby shall be taken into account by the Accountant in determining the amount to be paid to the Former General Partner.

(e) In the event the Partnership suffers damages as a result of or in connection with any withdrawal, resignation, or retirement of a Former General Partner, the Partnership shall be entitled to receive from the Purchaser, out of the amount otherwise due from the Purchaser to the Former General Partner, such portion of such amount as shall be equal to such damages or all of such amount if less than such damages. The rights of the Partnership under this subsection shall be in addition to any other rights and remedies it may have by law against the Former General Partner.

(f) On the Purchase Date, the Purchaser shall agree to indemnify and hold the Former General Partner harmless from and against any and all loss, damage, liability or expense that the Former General Partner may incur under any liability, debt or obligation of the Partnership that was included by the Accountant as a Partnership liability, debt or obligation in determining the amount payable to the Former General Partner by the Purchaser.

11.5 Additional or Successor General Partners.

(a) The General Partner with the written approval of a Class B Limited Partner Majority, shall have the right and power to admit to the Partnership from time to time one or more Persons as general partners, whether as an additional general partner or as a successor general partner; provided, however, that no additional or successor general partner may be admitted unless such general partner becomes a party to the LGS Equity Investors' Agreement and the other requirements of Section 2.3 of the LGS Equity Investors' Agreement have been satisfied. If such admission is to result in a reduction or dilution of the Percentage Interests of the Limited Partners, the reduction or dilution must be proportionate among all Partners.

(b) In the event the General Partner withdraws, resigns, retires, or is the subject of a Bankruptcy Action or the Partnership otherwise ceases to have any general partners for any reason (other than upon the winding up and termination of the Partnership), the Limited Partners shall have the right and power to continue the business of the Partnership if, within ninety (90) days after (but effective as of) the date on which there ceases to be a general partner, all of the Limited Partners determine in writing to so continue the Partnership and select a successor general partner; provided, however, that no successor general partner may be admitted unless such general partner becomes a party to the LGS Equity Investors' Agreement and the other requirements of Section 2.3 of the LGS Equity Investors' Agreement have been satisfied.

(c) The Limited Partners hereby consent to any admission made in accordance with the provisions of this<u>Section 11.5</u>. No Person shall be deemed to be admitted as a General Partner unless and until such Person shall have executed an instrument accepting and agreeing to be bound by all of the terms of this Agreement and executed an amendment to the Certificate. The General Partners shall cause this Agreement and the Certificate to be duly amended and filed to reflect any such admission.

11.6 <u>Changes in General Partners.</u> The General Partners shall promptly cause this Agreement (specifically, <u>Exhibit A</u>) and the Certificate to be duly amended and filed to reflect any changes in the composition or addresses of the General Partners.

ARTICLE XII

DISSOLUTION AND TERMINATION

12.1 Events Causing Dissolution. Subject to Section 7.3, the Partnership shall be dissolved upon the first to occur of the following events:

(a) any sale or other disposition by the Partnership of all or substantially all of its interests in non-liquid Partnership Properties, unless the Partnership in connection with such disposition acquires a promissory note or other evidence of indebtedness, in which case the Partnership shall be dissolved following the payment of such note or other evidence of indebtedness;

(b) a Bankruptcy Action occurs with respect to the Partnership;

(c) the expiration of ninety (90) days after the withdrawal of the last remaining General Partner or an event of withdrawal with respect to the last remaining General Partner occurs, unless an election and appointment has been made pursuant to <u>Section 11.5(b)</u>;

- (d) the determination of the General Partner and a Limited Partner Majority; or
- (e) except as otherwise provided herein, any other event causing a dissolution of the Partnership under the provisions of the Act.

12.2 Effect of Dissolution. Except as otherwise provided in this Agreement, upon the dissolution of the Partnership, the General Partner (or if there is no remaining General Partner, a liquidating trustee appointed by a Limited Partner Majority) shall proceed to wind up, liquidate and terminate the business and affairs of the Partnership. In connection therewith, the General Partner or liquidating trustee shall have the specific power and authority for, in the name of, and on behalf of the Partnership to liquidate and reduce to cash (to the extent necessary or advisable) the Partnership Properties as promptly as is consistent with obtaining a reasonable value therefor, to apply and distribute the proceeds of such liquidation and any remaining assets in accordance with the provisions of <u>Section 12.3</u>, and to do any and all acts and things authorized by, and in accordance with, the Act and other applicable laws for the purpose of winding up and liquidation. Notwithstanding the foregoing, no assets shall be distributed in kind to any Limited Partner without the consent of such Limited Partner.

12.3 Application of Proceeds. Upon the dissolution and liquidation of the Partnership, the assets of the Partnership shall be applied and distributed in the order of priority established in Section 6.2.

ARTICLE XIII

DISPUTE RESOLUTION

13.1 Disputes. This Article XIII shall apply to any (a) dispute, claim, or controversy arising out of or relating to this Agreement (whether arising in contract, tort or otherwise, and whether arising at law or in equity), including the performance, breach, validity, construction, interpretation, application, enforceability or termination of this Agreement or any of its provisions, and including with respect to whether a Partner is in compliance with, or breach of, any provision of this Agreement, and (b) the applicability of this Article XIII to a particular dispute, claim, or controversy (collectively, a "Dispute").

13.2 <u>Negotiations to Resolve Disputes</u>. The disputing Partners will, prior to the initiation of any claim or cause of action, each promptly appoint an officer or representative that has settlement authority to meet (in person or by teleconference) as soon as reasonably practicable in a good faith effort to settle or otherwise resolve the Dispute equitably and reasonably promptly; <u>provided</u>, <u>however</u>, that no settlement will be binding on the disputing Partners until reduced to writing and signed by the disputing Partners. If the Dispute is not settled or otherwise resolved by the earlier of the date forty-five (45) days following the first meeting of the representatives, or the date on which the representatives unanimously agree that a settlement or other resolution of the Dispute is not possible, then the disputing Partners may proceed as set forth in <u>Section 13.3</u>.

13.3 Arbitration.

(a) Any Dispute not resolved pursuant to <u>Section 13.2</u> shall be finally resolved by binding arbitration initiated upon the written notice (an "*Arbitration Notice*") of any disputing Partner. The arbitration shall be administered by the American Arbitration Association (the "*AAA*") and shall be conducted in accordance with this Agreement and the then current Commercial Arbitration Rules of the AAA (the "*AAA Rules*"), and judgment on the award may be entered in any court having jurisdiction thereof.

(b) The seat of the arbitration shall be Dallas, Texas, or such other place as the disputing Partners may agree. The arbitrator(s) shall determine the matters at issue in the Dispute in accordance with the substantive law of the State of Delaware, excluding the conflicts provisions of such law.

(c) In the event that any Partner's claim or counterclaim equals or exceeds \$1,000,000, exclusive of interest or attorneys' fees, the Dispute shall be heard and determined by three (3) arbitrators. Otherwise, the Dispute shall be heard and determined by one (1) arbitrator.

(d) In the event that one arbitrator shall hear the Dispute, the disputing Partners shall attempt to agree upon a qualified individual to serve as arbitrator. If the disputing Partners are unable to so agree within thirty (30) days of the Arbitration Notice, then the arbitrator shall be selected and appointed in accordance with the AAA Rules.

(e) In the event that three arbitrators shall hear the Dispute, each disputing Partner shall, within thirty (30) days of the Arbitration Notice, select one person to act as an arbitrator. The two arbitrators so selected shall, within fifteen (15) days of their appointment, select a third arbitrator who shall serve as the chair of the arbitral panel. The two Partner-selected selected arbitrators will serve in a non-neutral capacity. The arbitrators selected shall be qualified by education, training, and experience to hear and determine matters in the nature of the Dispute.

(f) If a Partner fails to appoint an arbitrator as provided herein, or if the arbitrators selected by the Partners are unable or fail to agree upon a third arbitrator within fifteen (15) days of their appointment, then that arbitrator shall be selected and appointed in accordance with the AAA Rules.

(g) Should an arbitrator die, resign, refuse to act, or become incapable of performing his or her functions as an arbitrator, the AAA may declare a vacancy on the panel. The vacancy shall be filled by the method by which that arbitrator was originally appointed.

(h) The arbitrator(s) shall be bound by and shall follow the then current American Bar Association /AAA Rules of Ethics for Arbitrators in Commercial Disputes.

(i) The arbitrator(s) shall apply this Agreement as written and according to its plain language in all respects, and shall in no circumstances have authority to add, delete, modify, or deviate from any of the terms of this Agreement as written, nor shall it/they have any authority to cancel or void this Agreement, in whole or in part, or to extend the term of this Agreement, other than as may be expressly provided herein.

(j) All awards shall be in writing and shall state the reasoning upon which the award rests including a statement of facts and conclusions of law. Any award shall be made and signed by at least a majority of the arbitrator(s).

(k) Unless the Partners agree otherwise, the Partners, the arbitrator(s), and the AAA shall treat the dispute resolution proceedings provided for herein, any related disclosures, and the decisions of the arbitrator(s), as confidential, except in connection with judicial proceedings ancillary to the dispute resolution proceedings and as otherwise required by law to protect the legal rights of a Partner.

13.4 <u>Survival</u>. The terms of this <u>Article XIII</u> shall survive the termination or expiration of this Agreement.

ARTICLE XIV

MISCELLANEOUS

14.1 <u>Meetings</u>. Meetings of the Partners may be called from time to time by the General Partner and shall be called by the General Partner upon the written request from time to time of a Limited Partner Majority. Unless all Limited Partners either waive in writing notice of a meeting or are represented at a meeting in person or by proxy, written notice of each meeting, stating its place, day and hour of the meeting and the purpose(s) thereof, shall be given to each Limited Partner not

less than five (5) nor more than thirty (30) days before the date of such meeting. No meetings of the Limited Partners or the General Partner shall be required for a group of Limited Partners or the General Partner to take any action or make any decision, and any written document executed by the requisite number of Limited Partners or the General Partner shall constitute the act or decision of the Limited Partners or the General Partner. The General Partner shall have the right to establish reasonable rules and procedures regarding the conduct of, and voting of the Limited Partners at, meetings of the Limited Partners.

14.2 <u>Title to Assets</u>. Title to the Partnership Properties shall be held in the name of the Partnership. No Partner shall individually have any ownership interest or rights in the Partnership Properties, except indirectly by virtue of such Partner's ownership of an interest in the Partnership. No Partner shall have any right to seek or obtain a partition of the Partnership Properties, nor shall any Partner have the right to demand and receive any specific assets of the Partnership upon the liquidation of or any distribution from the Partnership. The General Partner shall execute and file such documents as may be necessary to reflect the Partnership's ownership of the Partnership Properties in such public offices as may be required.

14.3 Nature of Interest in the Partnership. A Partner's interest in the Partnership shall be personal property for all purposes.

14.4 Notices. Any formal notice, demand, request or other communication required or permitted to be given by this Agreement shall be sufficient if in writing and delivered by hand or sent, postage prepaid, by registered, certified, or express mail, or by reputable overnight courier service, and shall be deemed given when so delivered by hand, or if mailed, three (3) days after mailing (or one (1) business day in the case of express mail or courier service for overnight delivery).

14.5 Third Party Beneficiaries. The provisions of this Agreement are for the exclusive benefit of the Partners, and solely with respect to Section 7.6 and Section 11.4(f), the indemnified Persons described therein. Except for the foregoing, this Agreement is not intended to, and does not, benefit or create any rights in any other Person.

14.6 <u>Entire Agreement</u>. This Agreement contains the entire agreement among the Partners, in such capacity, relative to the formation, operation and continuation of the Partnership.

14.7 <u>Amendments.</u> Except as otherwise provided herein, any amendment to this Agreement shall require the written approval of the General Partner and each Limited Partner; provided, however, that this Agreement (including <u>Schedule A</u>) (a) may be amended by the General Partner, without any execution of such amendment by any Limited Partner, if the amendment could not reasonably be expected to adversely affect any Limited Partner, or to reflect any action that the General Partner is expressly permitted to take, or any event that otherwise properly occurs, without the approval of any Limited Partner, and (b) may be amended by the General Partner, with the prior written approval of a Class B Limited Partner Majority, to reflect any action that the General Partner is expressly permitted to take with the approval of a Class B Limited Partner Majority, to reflect any action that the General Partner is expressly permitted to take with the approval of a Class B Limited Partner Majority, and <u>provided further</u> that this Agreement may not be amended in contravention of any then applicable provisions of the LGS Lease or the LGS Equity Investors' Agreement.

14.8 Governing Law. This Agreement and the rights and obligations of the Partners hereunder shall be subject to, governed by and interpreted in accordance with the Act and other

applicable laws of the State of Delaware, without regard to any "conflict of laws" principles or rules that would cause the laws of any other jurisdiction to apply.

14.9 Severability. The provisions of this Agreement will be deemed severable, and the invalidity or unenforceability of any provision in any jurisdiction will, as to that jurisdiction, be ineffective only to the extent of such invalidity or unenforceability, and will not affect the validity or enforceability of that or any other provision in any other jurisdiction. If any provision of this Agreement, or the application thereof to any Person or any circumstance, is invalid or unenforceable in any jurisdiction, (a) a suitable and equitable provision will be substituted for the invalid or unenforceable provision in order to carry out in that jurisdiction, so far as may be valid and enforceable, the intent and purpose of the invalid or unenforceable provision, and (b) the remainder of this Agreement, and the application of that provision to other persons or circumstances or in other jurisdictions, will not be affected.

14.10 Binding Agreement. Subject to the restrictions on the Transfer of Partnership Interests herein contained, the provisions of this Agreement shall be binding upon, and shall inure to the benefit of, the parties hereto and their respective heirs, personal and legal representatives, and permitted successors and assigns. This Agreement may be assigned by a Partner only as part of a transfer of its Partnership Interest as permitted by the terms of this Agreement.

14.11 Headings. The article, section, subsection, and paragraph headings contained herein are for convenience of reference only and are not intended to define, limit, or describe the scope or intent of any provision of this Agreement.

14.12 <u>Counterparts; Electronic Delivery</u>. This Agreement and any amendments hereto may be executed in any number of counterparts, each of which shall be deemed to be an original and all of which shall constitute one agreement that is binding upon all the parties hereto, notwithstanding that all parties are not signatories to the same counterpart. This Agreement may be delivered by the facsimile or other electronic transmission of signed signature pages.

[signature page follows]

2	0
2	9

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first written above.

PARTNERSHIP:

Pinedale Corridor, LP

By: Pinedale GP, Inc., its General Partner

/s/ David J. Schulte

Name: David J. Schulte Title: CEO

GENERAL PARTNER:

Pinedale GP, Inc.

/s/ David J. Schulte Name:David J. Schulte Title: CEO

LIMITED PARTNER:

Ross Avenue Investments, LLC

/s/ Ric E. Abel Name:Ric E. Abel Title: Executive Vice President

[Signature Page to Pinedale Corridor, LP Amended and Restated Limited Partnership Agreement]



CorEnergy Completes Acquisition of LGS from Ultra Petroleum

Positioning CorEnergy to Qualify as a REIT for 2013

FOR IMMEDIATE RELEASE

LEAWOOD, Kan. – December 20, 2012 – CorEnergy Infrastructure Trust, Inc. ("CorEnergy") (NYSE: CORR) today announced that it has closed its previously announced acquisition of a Liquids Gathering System ("LGS") from Ultra Petroleum Corp. (NYSE: UPL). The LGS will continue to be operated by UPL under a long-term triple net lease. Located in the Pinedale field in Wyoming, the LGS is a vital component of natural gas production in one of the top five natural gas fields in the U.S.

"We are pleased to have completed our largest acquisition of REIT-qualifying assets to-date," said Chief Executive Officer, David Schulte. "This is a mission-critical asset, with a high-quality tenant, and it represents a cornerstone for our energy infrastructure REIT strategy. The acquisition is accretive to our distribution and our Board of Directors has indicated that it intends to approve an increase in our quarterly distribution from \$0.11 to \$0.125 for the first full quarter following the acquisition. The REIT structure provides our investors direct access to U.S. energy infrastructure in an attractive vehicle with transparent cash flow."

Transaction Financing

CorEnergy paid \$205 million in cash and approximately \$24 million in certain other equity securities for the purchase of the LGS from UPL. The cash portion of the acquisition was funded with the net proceeds of CorEnergy's \$78 million common stock offering, approximately \$26 million from the sale of CorEnergy's publicly-traded master limited partnership equity securities, approximately \$5 million of cash from CorEnergy's balance sheet, a \$30 million concurrent co-investment from Prudential Capital Group and \$70 million in debt financing.

As a result of the transaction, CorEnergy's ratio of total debt to total assets is approximately 25 percent. CorEnergy expects to maintain a debt to asset ratio of between 25 and 50 percent.

REIT Status

The LGS now accounts for approximately 81 percent of CorEnergy's total assets on a pro forma basis as of August 31, 2012 and the LGS lease payments account for approximately 66 percent of total revenue on a pro forma basis for the nine months ended August 31, 2012.

CorEnergy anticipates that the LGS acquisition will allow the Company to meet the income and asset tests necessary to qualify and elect to be taxed as a REIT for 2013. Based on the value of CorEnergy's existing assets as of August 31, 2012, the Company expects that pro forma income for the nine month period ended August 31, 2012 would satisfy the REIT income tests and that at least 75 percent of our pro forma assets as of August 31, 2012 will qualify under the REIT requirements. Because certain of CorEnergy's assets may not produce REIT-qualifying income or be treated as interests in real property, the Company intends to contribute those assets into taxable REIT subsidiaries prior to 2013, in order to qualify as a REIT for 2013.

Distributions

The character of distributions made during the year may differ from their ultimate characterization for federal income tax purposes. As of November 30, 2012, the Board of Directors had declared total distributions of \$0.44 per share (\$0.11 per quarter) in the current year. CorEnergy expects to announce an anticipated annual distribution increase of \$0.06 per share from \$0.44 to \$0.50 per share, as the Company's current investments and LGS asset acquisition are expected to allow for such an annual distribution rate. If CorEnergy changes its fiscal year to a calendar year as anticipated, the Company's next distribution will be for the period beginning on

4200 W. 115th Street Suite 210 Leawood, KS 66211 | Main: 913-387-2790 | Fax: 913-387-2791 | corridortrust.com

December 1, 2012 and ending on March 31, 2013, with the anticipated \$0.125 per share quarterly distribution amount applicable to the period beginning January 1, 2013.

Going forward, CorEnergy intends to report standard performance measures utilized by REITs, including Funds from Operations ("FFO"), Adjusted Funds from Operations ("AFFO") and Cash Available for Distributions ("CAD"). A REIT is generally required to distribute during the taxable year an amount equal to at least 90 percent of the REIT taxable income (determined under IRC section 857(b)(2), without regard to the deduction for dividends paid). CorEnergy intends to adhere to this requirement in order to qualify as a REIT.

Strategy Update

CorEnergy intends to continue executing its strategy to be a diversified, energy infrastructure REIT. The Company expects to acquire assets that are diversified by asset size, geography and management team.

CorEnergy does not currently have any signed agreements or binding letters of intent for additional acquisitions. Certain opportunities are in preliminary stages of review, and consummation of any of these opportunities depends on a number of factors beyond CorEnergy's control. There can be no assurance that any of these acquisition opportunities will result in consummated transactions. With potential transactions ranging in value between \$50 million and \$200 million, CorEnergy expects to grow over time in order to attain the desired diversification.

In order to complete possible future transactions, CorEnergy has available to it the following funding mechanisms: issuance of common stock or other equity securities such as convertible or preferred stock, debt issuance, and equity partnerships, like that with Prudential Capital Group. CorEnergy's external manager is committed to structuring acquisitions that are accretive to CorEnergy's distributions to shareholders.

Asset Description

The Ultra Petroleum LGS was completed in 2010 and consists of more than 150 miles of underground gathering pipelines with 107 receipt points and four above-ground central gathering facilities that are utilized by UPL as a method of separating water, condensate and associated natural gas from a unified stream and subsequently selling or treating and disposing of the separated products. UPL's non-operating working interest partners in the Pinedale field where the LGS is located pay UPL a fee for the use of UPL's LGS. To date, no major operational issues have been reported with respect to the LGS.

The LGS has a current capacity of approximately 45,000 barrels per day and average throughput of approximately 36,000 barrels per day during the four quarters ended September 30, 2012. The underground pipes that make up the majority of the LGS and certain other components, such as the separators, have useful lives that extend beyond the initial term of the lease. CorEnergy believes that the LGS can be expanded at a relatively low incremental cost by, for example, adding additional separating equipment.

Most of UPL's exploration and development in the Pinedale field takes place on land under the jurisdiction of the Bureau of Land Management (BLM). The BLM has the authority to approve or deny oil and gas leases or to impose environmental restrictions on leases where appropriate. The BLM issued the Pinedale Record of Decision ("ROD") in September 2008. Under the ROD, Ultra Petroleum gained year-round access to the Pinedale field for drilling and completion activities in development areas, provided that Ultra Petroleum conducts an environmental mitigation effort, which includes the use of a liquids gathering system. This additional access resulted in increased drilling efficiencies and allowed for accelerated development of the field.

Lease Description

The 15-year triple net lease provides a minimum annual base rent of \$20 million, subject to an inflation adjustment. The lease also includes an additional participating rent based on volume growth of liquids in the LGS. The total rent (minimum plus participating) is capped at \$27.5 million annually for the initial 15-year

term. At the conclusion of the initial term, UPL's subsidiary may renew the lease for additional, successive 5-year terms. UPL's subsidiary will operate the LGS and will retain responsibility for maintenance and other capital expenditures required for its operation. UPL and one of its wholly-owned subsidiaries guarantee the lessee's obligations.

Guarantor Description

UPL was incorporated in 1979 and is an independent oil and gas company engaged in the development, production, operation, exploration and acquisition of oil and natural gas properties. UPL leases approximately 93,000 gross (53,000 net) acres in and around the Pinedale and Jonah natural gas fields of the Greater Green River Basin in southwest Wyoming. The most recently available EIA data, dated 2009, indicated that the Pinedale field was among the top five U.S. natural gas plays based on proved reserves. As of December 31, 2011, UPL held an approximately 50 percent working interest in approximately 1,700 producing wells in these fields. The Pinedale and Jonah fields have estimated natural gas reserves of over 48 Tcfe as of December 31, 2011.

As of December 31, 2011, UPL had an estimated 4.3 Tcfe of proved reserves and 10.2 Tcfe of proved, probable and possible (3P) reserves in the Pinedale and Jonah fields. UPL's third-party reservoir engineering firm, Netherland, Sewell & Associates, Inc., has identified an inventory of over 5,000 economic, future drilling locations.

UPL derives its revenues principally from the sale of its natural gas and associated condensate production from wells operated by UPL and others in the Greater Green River Basin. UPL is recognized as a low-cost operator in the industry in terms of both adding and producing oil and natural gas reserves. UPL's all-in cash costs, defined as all-in costs excluding DD&A expenses, have consistently been lower than natural gas prices and for the twelve month period ended September 30, 2012 were \$1.43 per Mcfe.

CorEnergy Advisors

KeyBanc Capital Markets acted as exclusive financial advisor to CorEnergy in connection with the acquisition.

BofA Merrill Lynch is acting as exclusive structuring advisor in connection with CorEnergy's energy infrastructure real asset strategy.

About CorEnergy Infrastructure Trust, Inc.

CorEnergy Infrastructure Trust, Inc. (NYSE: CORR), acquires midstream and downstream U.S. energy infrastructure assets and concurrently enters into long-term triple net leases with energy companies. CorEnergy intends to acquire infrastructure assets that qualify as real property for REIT purposes. The Company's principal objective is to provide stockholders with an attractive risk-adjusted total return, with an emphasis on distributions and distribution growth. Formerly Tortoise Capital Resources Corp., CorEnergy previously traded under the ticker TTO. CorEnergy is managed by Corridor InfraTrust Management, LLC. Corridor is an affiliate of Tortoise Capital Advisors, L.L.C., a registered investment adviser with over \$9.4 billion of assets under management in the U.S. energy infrastructure sector as of November 30, 2012. For more information, please visit <u>www.corridortrust.com</u>.

About Prudential Capital Group

Prudential Capital Group has been a leading provider of private debt, mezzanine and equity securities to companies worldwide for more than 70 years. Managing a portfolio of nearly \$65 billion as of September 30, 2012, Prudential Capital offers senior debt and mezzanine capital, leverage leases, credit tenant leases, and equipment finance to companies worldwide. The global regional office network has locations in Atlanta, Chicago, Dallas, Frankfurt, London, Los Angeles, Minneapolis, Newark, N.J., New York, Paris and San Francisco. For more information, please visit <u>www.prudentialcapitalgroup.com</u>.

About Ultra Petroleum Corp.

Ultra Petroleum Corp. is an independent exploration and production company focused on developing its long-life natural gas reserves in the Greater Green River Basin of Wyoming—the Pinedale and Jonah fields—and is in the ongoing exploration and early development stages in the Appalachian Basin of Pennsylvania. Ultra is listed on the New York Stock Exchange and trades under the ticker symbol "UPL".

Forward-Looking Statements

This press release contains certain statements that may include "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. All statements, other than statements of historical fact, included herein are "forward-looking statements." Although CorEnergy believes that the expectations reflected in these forward-looking statements are reasonable, they do involve assumptions, risks and uncertainties, and these expectations may prove to be incorrect. Actual results could differ materially from those anticipated in these forward-looking statements as a result of a variety of factors, including those discussed in CorEnergy's reports that are filed with the Securities and Exchange Commission. You should not place undue reliance on these forward-looking statements, which speak only as of the date of this press release. Other than as required by law, CorEnergy does not assume a duty to update any forward-looking statement. In particular, any distribution paid in the future to our stockholders will depend on the actual performance of CorEnergy, its costs of leverage and other operating expenses and will be subject to the approval of CorEnergy's Board of Directors and compliance with leverage covenants.

Contact Information:

Rachel Stroer, Investor Relations, 877-699-CORR (2677), info@corridortrust.com